

Delft University of Technology

Haptic feedback for flight envelope protection

van Baelen, D.

DOI 10.4233/uuid:1aa13612-f480-4862-9af1-7f3cc5570c62

Publication date 2020 **Document Version**

Final published version

Citation (APA) van Baelen, D. (2020). Haptic feedback for flight envelope protection. [Dissertation (TU Delft), Delft University of Technology]. https://doi.org/10.4233/uuid:1aa13612-f480-4862-9af1-7f3cc5570c62

Important note

To cite this publication, please use the final published version (if applicable). Please check the document version above.

Copyright Other than for strictly personal use, it is not permitted to download, forward or distribute the text or part of it, without the consent of the author(s) and/or copyright holder(s), unless the work is under an open content license such as Creative Commons.

Takedown policy

Please contact us and provide details if you believe this document breaches copyrights. We will remove access to the work immediately and investigate your claim.

This work is downloaded from Delft University of Technology. For technical reasons the number of authors shown on this cover page is limited to a maximum of 10.

Haptic Feedback

Flight Envelope Protection

Dirk Van Baelen

HAPTIC FEEDBACK FOR FLIGHT ENVELOPE PROTECTION

HAPTIC FEEDBACK FOR FLIGHT ENVELOPE PROTECTION

Proefschrift

ter verkrijging van de graad van doctor aan de Technische Universiteit Delft, op gezag van de Rector Magnificus Prof. dr. ir. T.H.J.J. van der Hagen, voorzitter van het College voor Promoties, in het openbaar te verdedigen op Woensdag 18 November 2020 om 12:30 uur

door

Dirk VAN BAELEN

Master of Science in Aerospace Engineering, Technische Universiteit Delft, Delft, Nederland, geboren te Herentals, België. Dit proefschrift is goedgekeurd door de promotoren:

Dr. ir. M. M. van Paassen Prof. dr. ir. D. A. Abbink

Samenstelling promotiecommissie:

Rector Magnificus,	voorzitter
Dr. ir. M. M. van Paassen,	Technische Universiteit Delft
Prof. dr. ir. D. A. Abbink,	Technische Universiteit Delft

Onafhankelijke leden:

Prof. dr. J. B. J. Smeets,	Vrije Universiteit Amsterdam
Prof. DrIng. G. Eitelberg	Technische Universiteit Delft
Prof. dr. A. M. L. Kappers	Technische Universiteit Eindhoven
Prof. DrIng. R. Luckner	Technische Universität Berlin
Prof. dr. F. C. T. van der Helm	Technische Universiteit Delft, reservelid

Prof. dr. ir. M. Mulder en dr. ir. J. Ellerbroek hebben in belangrijke mate aan de totstandkoming van het proefschrift bijgedragen.



Keywords: haptic feedback, flight envelope protection, manual control, situation awareness

- Printed by: Ipskamp
- Cover image: Stephan Timmers

Copyright © 2020 by D. Van Baelen

ISBN 978-94-6366-323-6

An electronic version of this dissertation is available at http://repository.tudelft.nl/.

Propositions

accompanying the dissertation

HAPTIC FEEDBACK FOR FLIGHT ENVELOPE PROTECTION

by

Dirk VAN BAELEN

- 1. Aircraft manufacturers should implement task-related haptic feedback on the flight deck. (this thesis)
- 2. Asymmetric vibrations can not only be used to alert pilots when unsafe aircraft states are imminent, but also to provide them with directional cues for control inputs towards safer states. (this thesis)
- 3. Asymmetric vibrations improve pilot learning rate to stay within flight envelope limits and combined with their relatively low-cost implementation should be used in flight training simulators. (this thesis)
- 4. Haptic guidance cues immediately increase a pilot's capability to stay within flight envelope limits, and should therefore be used in real flight, despite being more intrusive. (this thesis)
- 5. When applying Billings' design principles to haptic feedback, it follows that pilots should always be able to overrule any haptic feedback provided.
- 6. There is an inherent ambivalence in investigating pilot behaviour at the edges of the flight envelope: you want the pilot to approach the limit to see how (s)he behaves, and you want the pilot to stay away to prove the validity of your design.
- 7. Experiments with humans inherently have at least two confounding factors: the participant and the experimenter.
- 8. Time spent on activities not related to the doctoral study is a good indicator of progress in the doctoral study.
- 9. Trying to think of a proposition which proves the uselessness of propositions, actually proves the usefulness of propositions.
- 10. A nice cycling ride in Belgium gets you the fastest on top of the highest mountain hill; a nice ride in the Netherlands is neither the highest, nor fastest, it is one that does not end in a ditch ("gracht"). (home-work travel)

These propositions are regarded as opposable and defendable, and have been approved as such by the promotors dr. ir. M. M. van Paassen and prof. dr. ir. D. A. Abbink.



SUMMARY

Improving the safety level of aviation is vital to prevent serious accidents. One key area where improvements can be made is the prevention of loss of control occurrences, by preventing the aircraft state to pass beyond the limits from which no recovery is possible. Such improvements can focus on improved monitoring of the main flight parameters and active automation modes.

The limits of an aircraft are typically expressed in terms of a flight envelope which represents the allowable region of load factor versus velocity. Modern day aircraft can support pilots in monitoring the main flight parameters by employing a flight envelope protection system: the inputs of the pilots are routed to the flight control computers which can impose limits on those inputs. In doing so, the computers are protecting the aircraft state from leaving the flight envelope.

When the control device is linked to the control surfaces, for example using cables and pulleys, any limit imposed by the flight control computer can be felt by the pilot. With the advent of fly-by-wire control devices, the mechanical link is replaced by an electrical connection, resulting in the loss of this information using the sense of touch. This haptic information was initially not included as it requires active control devices which had issues regarding the size, power and stability requirements. The lack of such haptic information on the flight envelope protection system might have been a contributing factor in some accidents.

Nowadays active control devices do meet the requirements in terms of size, power and stability, and offer the possibility to re-introduce haptic feedback in fly-by-wire control systems. Therefore, this thesis looked at adding haptic feedback to the control device of a modern aircraft to increase pilot awareness of the flight envelope protection system.

First design iteration Based on literature, two main groups of haptic feedback were identified: vibro-tactile cues, for example vibrations on the side stick, and force feedback, such as changing stiffness for certain deflections. A first iteration, based on previous exploratory research, included both vibro-tactile cues and force feedback.

The design provided the pilots with a total of five cues: it (i) informs the pilot about an approaching limit using a single force pulse, called a 'tick-on-the-stick', (ii) indicates a non-desired control direction using the spring coefficient, (iii) warns the pilot of a dangerously low velocity using a stick shaker, (iv) shows a desired control input during an over-speed event by moving the control device, and (v) indicates the required control input at low velocities when a stick neutral position is not sufficient by moving the control device.

An evaluation using eleven professional Airbus pilots who flew a windshear and an icing scenario showed no significant changes in performance and safety margins for any of the conditions. It did show that the haptic feedback is not hindering pilots in performing their tasks and in the debriefing pilots indicated that they had an increased

situation awareness. The pilots expressed a clear potential benefit of implementing the haptic feedback system on a fly-by-wire flight deck.

This first experiment did show two possible points of improvement: the reason for the haptic feedback cues was not always clear, and the direction included in the tick-onthe-stick was ambiguous. To further investigate these challenges, they are isolated and analysed in smaller experiments.

Complementing visuals To make the triggering of the haptic feedback more transparent, a visual display was designed to supplement the haptic feedback. This new visual display, based on the existing primary flight display, showed all relevant variables for the haptic feedback. Using uniform colours, the triggering points were indicated, together with line thicknesses indicating the strength and direction of the haptic feedback.

The visual display was evaluated using sixteen professional Airbus pilots who flew a windshear, sidestep and go-around scenario. Results did not show significant differences in performance or safety margins when enabling or disabling the new display. The debriefing did show that pilot appreciation of the haptic feedback marginally increased and that they better understood the haptic feedback. Hence, we recommend to supplement haptic feedback on a flight deck with a visual support.

Vibrations design A haptic feedback system which involves only one group of haptic feedback, i.e., either vibro-tactile or force feedback, might be simpler to design and certify. Since the initial tick-on-the-stick was particularly well received by the pilots, a design with only vibro-tactile cues was considered next. Pilots did indicate that the direction of the cue was not clear, which required an investigation in a more effective vibration.

Using a just-noticeable-difference experiment, where the lowest perceivable threshold is determined, a sawtooth-shaped vibration was found to have best properties. Participants were able to indicate that a cue was provided and which direction it indicated. This type of vibration was therefore used in a next haptic feedback system to provide the pilot with a cue on the onset of the flight envelope protection, and an intermittent vibration for the duration of the protection activation.

This design was evaluated using 24 PPL/LAPL pilots who flew a challenging profile and encountered a windshear. Results showed again that the metrics did not change significantly. The results did show that the group of twelve participants who started with haptic feedback had a higher learning rate, compared to the other group of twelve participants who started without haptic feedback, and the former group did not have a change in metrics when the feedback was removed. This indicates a training benefit of the haptic feedback design using vibrations.

Guidance design Next, a haptic feedback system using the other group of feedback, force feedback, was designed. Literature showed that this system should be able to support the pilot from the first use, which was not the case for the vibration design. The force feedback design is actively moving the side stick to indicate the required input by the flight envelope protection system, and is increasing the stiffness for deflection which bring the aircraft state closer to the limits.

It was evaluated using another 24 PPL/LAPL pilots who flew the same scenarios as for the vibrations design: a challenging profile with a windshear encounter. Results showed that the twelve participants who were provided with the guidance design achieved improved safety margins from the first use, yet a deterioration was present when the feedback was no longer present. Participants who received no haptic feedback at all, confirmed the improved learning rate of the vibrations design, and the improved metrics at first use of the guidance design.

Conclusion Although the first design iteration had no conclusive results, the last two experiments showed that haptic feedback can be a useful addition to the flight deck. A vibration-based design can improve learning rate, which shows that pilots become more aware of the flight envelope, and is therefore recommended for use in training simulators. A guidance design can support pilots from the first use, again showing that pilot are more aware of the flight envelope, yet this support has to be always provided. In conclusion, haptic feedback can be used to improve pilot situation awareness of the flight envelope protection system.

CONTENTS

Nomenclature 1 Introduction 1 I Flight Envelope Protection					xiii
1 Introduction					
1.1Fight Envelope Flotection1.2Literature on haptics for vehicle control.1.3Haptic feedback design parameters.1.4Research goal.1.5Outline					1 5 7 10 11 12
I First design iteration					15
2 Design of a Haptic Feedback System for FEP2.1 Introduction2.2 Flight dynamics and control laws2.3 Haptic Display Design2.4 Operational test scenarios2.5 Conclusion3 Evaluation of a Haptic Feedback System for FEP3.1 Introduction3.2 Haptic Display3.3 Method3.4 Results3.5 Discussion3.6 Recommendations3.7 Conclusion	•	· · · · · · · · · · · · · · · · · · ·	· · · · · · · · · · · · ·	· · · · · · · · · · · · · · · · · · ·	 17 19 21 29 39 46 47 49 51 54 65 76 80 81
II Design improvements					83
4 Supplementing Haptics Through Visualization 4.1 Introduction 4.2 Background 4.3 Display design 4.4 Method 4.5 Results 4.6 Discussion	•				 85 87 88 90 95 104 116 110

5	Just Feeling the Force: JND for Asymmetric Vibrations	121			
	5.1 Introduction	123			
	5.2 Design of Asymmetric Vibrations	124			
	5.3 Method	127			
	5.4 Results	131			
	5.6 Influence of the asymmetric vibration	132			
	5.7 Conclusions	135			
6	Evaluating Asymmetric Vibrations for Feedback on FEP	137			
	6.1 Introduction	139			
	6.2 Haptic Display	139			
	6.3 Method	143			
	6.4 Results	152			
	6.5 Discussion	162			
	6.6 Conclusion	164			
7	Evaluating Guidance for Feedback on FEP	165			
	7.1 Introduction	167			
	7.2 Feedback Design	168			
	7.3 Method	174			
	7.4 Results	185			
	7.5 Discussion	199			
	7.6 Conclusion	203			
8	Discussion, Recommendations, Conclusion	205			
	8.1 Discussion	207			
	8.2 Recommendations	211			
	0.5 Conclusion	215			
Арр	endices	217			
A	Experiment questionnaires	219			
B	Approach Charts and Checklists	237			
C	Experiment Data	249			
R	References	261			
A	cknowledgements	273			
C	Curriculum Vitæ 27				
L	ist of Publications	279			

NOMENCLATURE

Air Data and Inertial Reference Unit ADIRU ATC Air Traffic Control Elevator & Aileron Computer ELAC FAC Flight Augmentation Computer FBW Fly-By-Wire FCC Flight Control Computer Flight Crew Operation Manual FCOM FCU Flight Control Unit FD Flight Director FEP **Flight Envelope Protection** Flight Management Guidance Computer FMGC HFL Haptic Feedback Law HMI Human Machine Interaction ILS Instrument Landing System Light Aircraft Pilot License LAPL Landing Gear Control Interface Unit LGCIU Navigation Display ND PFD Primary Flight Display Private Pilot License PPL **Radio Altimeter** RA RSME **Rating Scale Mental Effort** SEC Spoilers & Elevator Computer SFCC Slat Flap Control Computer SIMONA Research Facility for SImulation, MOtion and NAvigation Trimmable Horizontal Stabilizer THS Take Off/Go Around TOGA

Greek Symbols			
α	Angle of attack, rad		
β	Side slip angle, rad		
γ	Flight path angle, rad		
δ	Control device deflection, rad		
θ	Pitch angle, rad		
ρ	Density, kg/ m^3		
ϕ	Roll angle, rad		
Roman Sy	mbols		
a	Acceleration, m/ s ²		
b	Damping, Nm s/rad		
C_L	Lift coefficient, -		
D	Drag, N		
F	Force, N		
g	Gravitational acceleration, m/ s ²		
I	Amplitude of the discrete tick, Nm		
Κ	Gain, -		
k	Spring, N/ rad		
L	Lift, N		
M	Mach, velocity relative to the speed of sound, -		
т	Mass, kg		
n	Load factor, g		
q	Pitch rate, rad/ s		
S	Surface, m ²		
Т	Thrust, N		
t	Time, s		
V	Velocity, m/ s		
W	Weight, N		
x	Distance from starting position, m		
Subscripts			
a	Aerodynamic reference frame		
b	Body reference frame		
br	Breakout		
Ε	Vehicle-carried normal Earth reference frame		
haptic	Contribution of the haptic feedback system		
human	Human/pilot contribution		
lat	Lateral value		
lon	Longitudinal value		
max	Maximum value		
MO	Maximum operational value		
min	Minimum value		
nom	Nominal value		
пр	Neutral point		
prot	Protected region value		

INTRODUCTION

TRANSPORTATION safety is important for public trust and maintaining or increasing the safety level is essential, particularly in aviation where accidents are highly visible. [1] In order to make sure that the number of accidents does not grow, and even decreases, the current aviation safety level should be maintained and preferably improved. When looking where this improvement can be made, aviation safety boards, such as the European Union Aviation Safety Agency EASA, and airline associations, for example the International Air Transport Association IATA, indicate that one key area to improve is the number of occurrences of loss of control in flight. [2, 3] A loss of control event occurs when the aircraft reaches an unintended state, beyond intended operating limits, from which recovery to a state back into desired operating limits may be difficult or impossible before collision or disintegration. Looking into the details of the annual safety reports of the respective boards, aircraft limit excursions due to weather phenomena, and inadequate monitoring of main flight parameters or autopilot modes, are identified as some of the main contributors to loss of control. As such, improving the monitoring of main flight parameters and automation modes is a key step in ensuring future safe aviation.

Steps have been made in the past to ensure that pilots are informed about the limits, and even to prevent them to exceed the limits of the airplane. These allowable aircraft limits can be described in multiple ways, of which the most important for the current thesis is the flight envelope, which describes an allowable region of velocity and load factor combinations. In current day aircraft, these limits can be guarded by a Flight Envelope Protection (FEP) mechanism: the inputs from pilots are routed to the flight control computers which can act on approaching limits. One possibility is to impose hard limits on those input and, in doing so, the computers are effectively making sure that an input does not result in an aircraft state outside the allowable region, hence keeping the aircraft within an acceptable flight envelope.

When such a system is implemented on a set of reversible flight controls, i.e., there is a hard link from the control surfaces to the control device by for example a combination of cables and pulleys, any limit imposed by the FEP can be observed by the pilot by *feeling* what the controls are doing, i.e., by the haptic feedback present in the control device. A typical example with extreme effect: a stick-pusher as present in some high-tail configuration aircraft such as the Lockheed F-104 Starfighter, which provides a strong stick-forward force for high pitch rates and angles of attack, resulting in a nose-down input and protecting the aircraft from stalling. [4]

With the advent of fly-by-wire control devices, the mechanical link between the control device and control surfaces is lost, however, and sensors register the control device properties and use electrical wires to communicate pilot intent to the computer. The information of the FEP is therefore not observable anymore by the pilot using the sense of touch. Losing the direct mechanical connection from the pilot to the protection system might give rise to miscommunication between the pilot and the automation. [5, 6]

Air France 447 was an Airbus A330, a fly-by-wire aircraft with a passive control device, flying from Rio de Janeiro to Paris. Two hours in flight, the aircraft reverted to a less stringent FEP system due to a sensor failure. Surprised by the high altitude dynamics of the Airbus A330 aircraft and confused about the active FEP modes, the pilots stalled the aircraft, resulting in a fatal crash. The approaching stall would have been seen on the display in front of the pilots in a normal situation. Unfortunately, due to the sensor

failure, this visual information was not present. The stall was only communicated to the pilots with an aural warning, initially masked by a master caution warning. A buffet could occur, yet no clear mentioning of the buffet phenomenon is made in pilot training. The accident report indicates that "The addition of a visual signal to supplement the audible signal (warning) and the proprioceptive signal (the buffet), would provide the crew with additional information to enable them to escape from an erroneous understanding of the situation.". [7] Although not discussed in the accident report, this additional signal on the stall does not necessarily need to be a visual signal. Such a signal could be provided to the pilot using the sense of touch, which requires an active control device.

In another case, Air Asia 8501, an Airbus A320 rudder limiter malfunction, resulted in a degraded FEP mode which disengaged the autopilot. It took the pilots nine seconds before a correcting action was inputted, in which time the aircraft had reached a bank of 54°. The delayed response of the pilot was likely due to his attention not being on the primary display. Their subsequent actions resulted in extreme bank angles and a prolonged stall. This flight also crashed, resulting in loss of life for all crew and passengers. [8] In order to prevent similar accidents from re-occurring, the author feels that the pilot should be supported with information on the crucial states, especially during the initial bank where the attention of the pilot was distracted from the displays. One possibility to support the pilot during such a situation could be to use human sense other than the visual: an aural warning, or, again, a signal using the sense of touch.

In both cases, the aircraft, an A330 and A320, have a computerized system providing FEP, and a passive control device, providing pilots with no direct haptic feedback on that system. These examples indicate that not having this direct means of feedback might contribute to reduced pilot awareness on the aircraft state, which poses dangers especially when sensor failures occur and the FEP system reverts to less protected regimes.

A reason that this haptic information was not fully integrated after the introduction of fly-by-wire systems, is the device itself required to implement the forces. This used to be an issue because of the size, power and stability requirements resulting in certification problems. However, nowadays low-weight reliable force feedback for control devices offer the possibility to re-introduce haptic feedback in the fly-by-wire control systems. [9]

Connecting the pieces of the puzzle presented above, increased automation on the protection of the aircraft limits, and the loss of haptic information to the pilot, gives us a straight-forward possibility: use the haptic feedback to present pilots with information on the FEP. This could be done by replicating the force felt on the control device in the 'old' system. Nevertheless, since the fly-by-wire setup is present, the feedback is not limited to these options. Using the information of the latest FEP systems and their parameters, much more advanced and/or innovative designs are possible. This is the main topic of this thesis: how to supply the pilot with haptic feedback on the FEP.

First, the information which is being provided to the pilot is explained in Section 1.1. To explore the possibilities of using the sense of touch as feedback to the pilot, Section 1.2 discusses existing solutions, in which haptic feedback is provided to a human operator, followed by the haptic feedback design options considered in this thesis in Section 1.3. Finally, the scope for the current research and the outline of the thesis are shown in respectively Sections 1.4 and 1.5.

1.1. FLIGHT ENVELOPE PROTECTION

Although the haptic feedback concepts developed in this thesis are applicable to multiple environments, this thesis will use a setup similar to an Airbus A320. While the current A320 aircraft flight deck comprises a side stick, rudder pedals, and passive throttle levers, this thesis will focus solely on the side stick. As mentioned before, this is a fly-by-wire system, an (electrical) signal is sent from the side stick to the computers, no mechanical connection exists between the control surfaces and the side stick. During flight, the side stick is used to provide pitch attitude change commands to the flight control computers using a control law called C^* , which is further elaborated in Chapter 2. [10–13]

The flight envelope represents the allowable operating space of an aircraft, typically expressed in terms of limits on variables or combinations of variables. Different combinations of variables are used, for example the allowable angle of attack versus side slip angle, or roll rate versus velocity. When the aircraft is maneuvered to a state close to the edges of the nominal flight envelope, the computer can limit the inputs of the pilot to make sure the aircraft does not pass these limits, in other words, the computer keeps the aircraft state within the normal flight envelope. This thesis looks mainly at the flight envelope for longitudinal flight, which during both design and operation is typically represented by the allowable load factor versus velocity, and is used for the flight envelope protection system. [13]

The nominal limits of this flight envelope, as shown by the black line on Fig. 1.1, are determined by the maximal operation velocity (V_{MO}), the maximum and minimum load factor (respectively, n_{max} and n_{min}), and the velocity related to the maximum angle of attack ($V_{\alpha_{max}}$). When approaching these limits, the control law can be slightly altered, such that the pilot perceives the aircraft as having a natural tendency to stay away from those limits. The point at which these changes occur can be indicated in the flight envelope by a smaller region inside the nominal envelope, called the safe flight envelope, indicated by the dashed red line in Fig. 1.1. All states inside of this safe flight envelope can be considered as "safe" since there is sufficient margin to the limits.



Figure 1.1: Flight Envelope: allowable load factor versus velocity

This flight envelope protection is possible when all computers and sensors are working. When a sensor fails, or a computer malfunctions, the aircraft can revert to a less protected state. In such a state, some flight envelope protections might be lost, increasing the risk of unusual attitudes, and allowing more extreme maneuvers to the pilot. The control law change at the limits of the safe flight envelope, and the different levels of protection of the nominal flight envelope are implemented on current fly-by-wire aircraft such as the Airbus A320 or A330. They are further elaborated in Chapter 2.

The entire control loop is summarized in Fig. 1.2, which illustrates the flight envelope protection setup: the flight control computers take both the input of the pilot, as well as the information of the flight envelope protection system. Based on these inputs, the inputs to the aircraft using the control surfaces are calculated. The information on the flight envelope is communicated to the pilot using both visual and auditory information.



Figure 1.2: Pilot-aircraft control loop, combined with the haptic feedback as presented in this thesis in thick

It is, however, important to note that this information is not always clearly perceived by the pilots. Accidents have occurred, for example the crash of Air France flight 447 discussed before, where the lack of FEP and unawareness of the flight envelope limits might have been contributing factors. [7] It is therefore paramount to investigate whether more intuitive communication of this information is possible.

This thesis looks at the use of haptic feedback to provide the pilot with information on the flight envelope protection system. The implication of adding haptic feedback to the control loop is indicated in Fig. 1.2, using the thicker lines: an additional dependency of the side stick on the aircraft state and flight envelope protection system.

As the flight envelope protection system is directly linked to the distance of the aircraft state to the flight envelope limits, the information transmitted through the haptics will be the "relative distance of the current state to the limit". As such, when the aircraft state is clear of any of the flight envelope limits, the side stick will have nominal characteristics. When the aircraft state approaches the limits, the haptic feedback will become active together with the flight envelope protection system. In other words, the activation of the flight envelope protection can be observed by a possible change in control law *and* the haptic feedback, for which the haptic cues are elaborated in the next section.

1.2. LITERATURE ON HAPTICS FOR VEHICLE CONTROL

This section provides a brief overview of existing solutions which use haptic feedback to provide the human operator with information during the control of a vehicle. It provides the context of this thesis with some general applications, followed by some automotive and aerospace examples.

1.2.1. GENERAL APPLICATIONS

The first examples of haptic feedback use tactors, i.e., vibrating elements attached to some part of the body, to provide vibro-tactile feedback. The information transmitted by the tactors is mostly a warning and can be applied to almost any part of the body: fingers, arms, chest, legs, etc.

The literature involving tactors shows that it can be used to draw attention to a secondary task ([14–17]), spatial location of approaching traffic ([18]) and corresponding recommended action to avoid a collision ([19, 20]), act as an artificial horizon ([21–29]), and provide information on the state of the wing of an aircraft. [30, 31]

In summary, tactors are mostly used to provide the pilot with an alert of some system, or information on spatial orientation. They are not used to show the pilot information on the flight envelope limits, especially not on the proximity to these limits. Additionally, the tactors require the pilot to wear a specialist piece of equipment, either a vest or sleeves, which could be expensive, and personal to a pilot. If such parts are not worn correctly, the information might be encoded to a wrong part of the body resulting in wrong information. Combining all these possibilities and issues of tactile feedback, using only tactors was not deemed satisfactory for the intended application of this thesis.

Numerous examples exist where the control device is not just vibrating, yet the dynamical properties are actively changed. One example for the tele-operation of a UAV showed that the addition of artificial force can help the human controller avoiding hazardous areas. By providing the operator with forces to *avoid* certain areas, workload was reduced and safety increased. [32] Following this example, addition of forces on control device on the flight deck might be used to *avoid* areas outside the safe flight envelope. Artificial forces can also be used to *attract* a certain path, which was shown to be more effective for an abstract control task. [33] Such systems can be used to draw the attention of the pilot to a preferred region in the flight envelope.

Applications can have one path which is preferred, in the context of the thesis, one can think of a certain path through the flight envelope. When guiding the human operator along to that respective path, van Paassen et al. argue that such a design requires four choices: (i) the human-compatible reference path, (ii) the level of haptic support, i.e., feed-forward, (iii) the level of strength and strategy of haptic feedback, i.e., feed-back, and (iv) the overall level of haptic authority. [34]

1.2.2. Automotive applications

The driver of a car has similar objectives as a pilot: operate the car within the allowable limits the road. One simplified 'driving envelope' can define the allowable operating space by the side of the road and the distance to the car in front, the current state is the position car on the road. When operating the car within these limits, driving can be considered safe, and haptic feedback can be supplied to enhance safety and performance.

In order to protect the longitudinal element of the 'driving envelope', i.e., to avoid a head-on collision, haptic feedback can be provided through the gas pedal by increasing the stiffness when driving closer to the car in front. This was shown to improve car-following performance while reducing control activity. [35]



Figure 1.3: Example of haptic feedback applied to the steering wheel of a car in the HMI simulator at TU Delft

The lateral envelope can be protected by providing haptic feedback through the steering wheel as shown on Fig. 1.3, where vibrations can be used, so-called 'motor-priming', to provide a warning and direction similar to the vibrations shown before. [36] Active feedback which is more involving can be provided by using an offset force to indicate a required deflection, and changing the stiffness to indicate a criticality of the action. [37] This type of feedback can be used to provide support to either steer the car away from the boundaries, or steer the car to one specific path, comparable to the artificial forces discussed before. Within the automotive field, no clear preference is present and different techniques are tested. It was found that boundary-avoidance was susceptible to driver annoyance when the haptic system intervened too early, and steering towards a specific path might suffer from after-effects when no feedback is supplied. [38] Advantages and disadvantages for both types of feedback have to be kept in mind for designs in this thesis.

1.2.3. AEROSPACE APPLICATIONS

Within the aerospace domain, several haptic feedback designs have been investigated, where one design changed the control device position to communicate the autopilot commands to the pilot. This was preferred over a manual or fully-automated system. [39]

Most of the remaining research investigates whether the pilot can be supported in manual flying. As discussed before, two main groups can be distinguished: boundary-avoidance and path tracking. The latter group can be simplified as a system where both the autopilot and pilot operate the control devices, hence the aircraft. Such a 'haptic flight director' can improve pilot performance ([40]), and can be combined with guidance presented on a tunnel-in-the-sky display. [41] The direction of these forces can be changed: a pilot can be asked to follow the control device, i.e., exert no force, or can be asked to oppose the force, i.e. maintain position. Both improved control performance, yet the latter increased required physical effort. [42] These examples assume that there is a preferred path which is away from the boundaries. When the boundaries are exceeded, showing a preferred path to return the aircraft within its limits was found to be useful. [43]

The second group involves boundary-avoidance using haptic feedback. This was applied on a helicopter to provide the pilot with the limits on the engine using a visual indication and a soft-stop: a local step increase of force required to move the controls. It resulted in handling qualities benefits and reduced limit exceedances. [44, 45]

The controls in large passenger aircraft can be connected to a hydraulic system to adjust the control surfaces. This allows for larger forces, yet limits the control system by a maximum rate of change. When a limit on the rate of change is encountered, the pilot-vehicle system might develop an oscillatory response: a pilot-induced-oscillation. Haptic feedback has been used to prevent encountering such a rate boundary by increasing the static friction, or changing the natural frequency and damping. [46] Other designs increased the stiffness of the control device and lowered the gain on the pilot command when the hydraulic system ([47]), or, in the case of fly-by-wire, control law ([48]) was not able to keep up with the pilot command.

Older aircraft with reversible flight controls might still benefit from a retrofitted FEP system. This has been found to be useful, and as it acts directly on the cables connecting the control device and surfaces, the pilot can directly feel the system working. [49]

When a FEP is present, approaching the limits can be communicated to the pilot using an increased resistance force. This was shown to have a positive effect on flight safety. [50] Additionally, providing the pilot with a new visual display with information on the FEP can improve safety, yet might be difficult to add to the already visually loaded flight deck. [51] Another example involves haptic feedback that actively changes the reference point of the control input, and limits the deflection based on the remaining control space, complemented by a visual indication, showing promising results. [52]



Figure 1.4: Example of haptic feedback applied to the side stick on the SIMONA flight deck at TU Delft

The examples in literature show that haptic feedback can be applied using multiple cues. Additionally, they show that there are a few examples which translate the limits of the aircraft to the pilot, and those examples do not provide a conclusive design rationale yet. It proves that there is still a lot to learn about haptic feedback for flight envelope protection. As such, an initial design has been setup and tested on a side stick shown on Fig. 1.4. [53, 54] This was, as the other designs presented here, showing promising results and proposed several points of improvement. This initial design, together with the lessons learned, formed the starting point of this thesis.

9

1.3. HAPTIC FEEDBACK DESIGN PARAMETERS

Many different applications, and matching definitions, of haptic feedback are possible. To clearly frame how this thesis looks at haptic feedback, the haptic profile is defined here as the amount of static force on the control device (*F*) required to move the control device to a certain deflection (δ). For most applications, for example the steering wheel in a car, the rudder pedals on the flight deck, as well as the side stick, the haptic profile resembles a linear or piece-wise linear relation as shown on Fig. 1.5a.

When no force is applied to a control device, it returns to a position called the neutral point (δ_{np}). From this position, the breakout deflection (δ_{br}) indicates a region with increased stiffness, as such, a threshold force is required to break the control device out of the neutral position. Next, the amount of force required for a given deflection is usually increasing linearly with the deflection, defined by the spring coefficient, up to a maximum allowed deflection (δ_{max}). The spring coefficient can have different values for both positive and negative deflections, respectively k^+ and k^- .



(a) Default haptic profile: static force (b) A required to move the side stick

(b) Adding of stick shaker

(c) Increased stiffness for positive deflections and shifted neutral point

Figure 1.5: Haptic Profile: static force required to move the side stick

This conventional haptic profile provides the pilot with information of the input magnitude: larger inputs require larger forces. Following the examples from literature, additional haptic feedback is explored in this thesis by actively changing the haptic profile in two main ways: providing the pilot with vibro-tactile cues, and providing the pilot with force feedback. The first group, vibro-tactile cues, is a vertical shift of the haptic profile with a small effect in time. As such, to show what kind of vibrations are used, a time history of the vibration is shown next to the haptic profile on an inset graph, as on Fig. 1.5b. Such cues are typically used to indicate a critical region of the envelope. [36, 55]

The latter group, continuous force feedback, is a gradual change of the haptic profile, for example, changing the neutral point and stiffness of the side stick in one direction. This results in an altered haptic profile as shown on Fig. 1.5c. Such cues are typically used to indicate, respectively, required control inputs ([41, 56]), and unwanted deflections. [48]

The International Standarization Organization (ISO) defined that haptic feedback can be divided into both tactile and kinaesthetic feedback. [57] Since the cues shown here are considered to provide *both* tactile feedback through mechanical stimulation of vibro-tactile cues, and kinaesthetic feedback through torques on the body of the pilot induced by stiffness and neutral point changes, the more general term 'haptic feedback' is used to describe the type of feedback. These haptic cues will now be used throughout the thesis to provide pilots with feedback on the flight envelope protection system.

1.4. RESEARCH GOAL

The increased automation on current flight decks in the form of FEP, combined with the possibility of providing pilots with haptic feedback, presents a new research opportunity. Literature shows that haptic feedback can indeed be used to provide pilots with feedback on the vehicle/aircraft, nevertheless sources which provide the pilot with active feedback on the FEP are missing. Therefore, the main research goal for this thesis is given by:

Research goal

Within the current fly-by-wire flight deck, improve pilot situation awareness of the aircraft flight envelope protection system using haptic feedback.

Previous exploratory research by Ellerbroek et al. showed that a haptic feedback system for flight envelope protection can indeed be designed, yet that research lacks an in-detail description and a rigorous evaluation. [53] Therefore, the first task is to re-visit this design and to implement one of the main lessons learnt in that research: the addition of a discrete warning cue. Therefore the first research question of this thesis is stated by:

Research Question 1: Combining vibrations and guidance design

Does a haptic feedback design combining stiffness changes, neutral point shifts, stick shaker, and discrete cues improve pilot situation awareness?

Literature showed that information is best presented using multi-modal displays, i.e., using haptics *and* visuals. [58] The first iteration, the subject of Research Question 1, did not include a visual display and therefore the actions of the haptic support system might not always have been clear to the pilot. As such, providing a more transparent source of haptic information might be achieved when complementing it with a graphical, visual display. This is the topic of Research Question 2.

Research Question 2: Complementing visuals

What kind of visual display can be used to complement the haptic information?

Research Question 1 involves all haptic feedback cues elaborated in Section 1.3. Implementing a system which is capable of all cues might be difficult in terms of design or certification. A system which uses only a sub-set of those cues might be simpler to design/certify, for example a system using only vibrational cues, resulting in a system which only warns the pilot when approaching the limits. For this, a suitable cue has to be designed and it has to be evaluated whether such a simpler system may be equally useful. Both topics will be tackled with Research Question 3.

Research Question 3: Vibrations design

Does a haptic feedback design using only vibrations improve pilot situation awareness?

After a cueing system, a logical next system involves more active haptic feedback guiding the pilot near the limits. Such a system could use a change in stiffness to resist pilot inputs, and actively change the neutral point to indicate a required deflection. The investigation of such a guidance system is the topic of the last research question.

Research Question 4: Guidance design

Does a haptic feedback design combining stiffness changes and neutral point shifts improve pilot situation awareness?

1.5. OUTLINE

This thesis consists of seven chapters in two parts. Chapter two is published in a peerreviewed journal, all other chapters have been published in the proceedings of peerreviewed conferences. The details of these publications can be found at the beginning of each chapter. The chapters' content is equal to the publications, with small modifications for smoother transitions and more consistent terminology. As such, all chapters can be read independently.

After this introduction, the parts are structured to follow the research questions, shown in Fig. 1.6, and are elaborated in the following of this chapter. Part I presents the first iteration of the haptic feedback design following the initial research. Part II elaborates on the research to investigate specific aspects of the design. Finally, conclusions and recommendations are given in Chapter 8.

1.5.1. PART I: FIRST DESIGN ITERATION

The first part discusses the initial design, and applies the main lessons learnt from that evaluation. One main lesson learnt from the initial haptic feedback design: the pilots indicated that the transition into the FEP was not always clear and could perhaps better be indicated by just a tick-on-the-stick. [53] To have a clear overview of what the FEP can do, Chapter 2 starts with elaborating on these protections. Using this list, the discrete cue is incorporated into the initial design and all cues are elaborated in detail to fully disclose the working principles of the haptic feedback system.

To evaluate whether the system is effective in promoting awareness of the pilot on the flight envelope limits, an experiment and the corresponding results are shown in Chapter 3. This experiment invited eleven active Airbus pilots to fly approaches into Schiphol and Montpellier. In both airports, the pilots encountered one of two events, respectively icing, which makes the limits of the aircraft envelope shrink, and a windshear, the pilot has to maneuver the aircraft close to the flight envelope limits in the recovery.

1.5.2. PART II: EXPLORING THREE DESIGN IMPROVEMENTS

The initial experiment showed that pilots did not fully understand the haptic feedback, and that the scenarios provided pilots with much freedom, resulting in a large spread in data points. Hence, it was concluded that both the way the system is evaluated, as well as the system itself, were far from perfect. Therefore Part II investigates which improvements can be made to further improve the design of the haptic system and the sub-improvements are tested in smaller experiments.



Chapter 8: Conclusion and recommendations

Figure 1.6: Structure of this thesis

In the evaluation of the first design iteration, the pilots indicated that the reason for triggering the haptic feedback was not always clear. Therefore, a visual display was designed to support the haptic feedback as shown in Chapter 4.

Additionally, the first evaluation showed that adding the discrete cue, i.e., a tick-onthe-stick, when the FEP becomes active, was appreciated by the pilots, perhaps even more as all the other cues. Nevertheless, the cue was intended to not only show the activation of the FEP, but also give information on which control input is required to move away from this protection. This directional information was not always clear to pilots, which warranted more research. As such, Chapter 5 investigates different signals to determine which has the clearest direction information and pin-point a threshold of minimal force required for pilot to be still able to feel that direction. 1

To determine whether the design can use *only* these new cues, or parts of the first design iteration is required, Chapter 6 evaluates this new design with a reduced experiment, using the lessons learned from the experiment setup of the first design iteration.

This last design evaluation did not show a clear benefit of adding haptic feedback in terms of averaged results, it did show a potential training benefit. To further explore this training benefit, another haptic feedback system involving active guidance feedback was designed. This guidance design is evaluated in Chapter 7, where its training effects are compared to the cueing design.

Conclusions on the different designs presented in this thesis, together with recommendations for future design and implementation, are provided in Chapter 8.

FIRST DESIGN ITERATION

2

DESIGN OF A HAPTIC FEEDBACK SYSTEM FOR FLIGHT ENVELOPE PROTECTION

Several modern aircraft use a passive control manipulator, a spring-damper system which generates command signals to the flight control computers in combination with a flight envelope protection system which limits pilot inputs when approaching the aircraft limits. This research project aims to increase pilot awareness of such protection systems through the use of force feedback on the control device, i.e., haptics. This chapter describes in detail how the haptic feedback works, and when it triggers; the next chapter will discuss the results of an experimental evaluation. With the current haptic design, pilots can get five cues: first, a discrete force cue when approaching the limits. Second, an increased spring coefficient for control deflections which bring the aircraft closer to its limits. Third, a stick shaker for low velocities. Fourth, if a low velocities condition requires an input, the stick is moved forward to the desired control input. And finally, the stick follows the automatic Airbus-like 'pitch up' command during an over-speed condition. This novel system is expected to help pilots correctly assess the situation and decide upon the right control action. It will be evaluated in two scenarios close to the flight envelope limits: a windshear and an icing event.

This chapter is published as:

Title
 Design of a Haptic Feedback System for Flight Envelope Protection

Journal Journal of Guidance, Control, and Dynamics, vol. 43, pp. 700 – 714, Feb. 2020

Authors D. Van Baelen, J. Ellerbroek, M. M. van Paassen, and M. Mulder
2.1. INTRODUCTION

M^{ODERN} flight decks provide an abundance of information to pilots, primarily using the visual and aural communication channels. Examples of visual displays are the Primary Flight Display (PFD) for the most important aircraft states, and the Navigation Display (ND) for a planar, top-down overview of the environment. Aural signals are often used to provide urgent messages such as to warn pilots for excessive velocities, and to provide altitude read-outs and throttle back-commands on landing. [59]

But apart from these senses, pilots are able to perceive information in several other ways. This chapter will elaborate on the use of the pilots' haptic sense, by providing haptic feedback through the control device. As shown in Fig. 2.1, within the field of haptic research two main categories are identified: touch, stimuli to the skin, and kinaesthesis, stimuli to the receptors in the muscles, joints and tendons.[57, 60] The design discussed in this chapter uses both touch and kinaesthesis, hence the term haptic feedback is used.



Figure 2.1: Components of haptics [57]

In most aircraft of the 20th century the control manipulator 'feel' provided information on, for instance, aerodynamic forces, buffeting when close to a stall, actuator saturation through hard stops of the controls, and other control-related phenomena. With the introduction of fly-by-wire, however, the forces on the control surfaces and the control devices were decoupled, eliminating this potentially very useful haptic information. [61]

A reason that haptic feedback on the aircraft limits was not integrated after the introduction of fly-by-wire systems in the 1980s and 1990s, was the rather bulky device required to implement the haptic forces. These old devices had issues regarding their size, weight, power and stability requirements, resulting in certification difficulties. Currentday devices have become much smaller and lighter while still able to provide reliable haptic feedback. [62] This offers the possibility to re-consider this type of feedback in fly-by-wire control systems. [63]

Together with the advances in control devices, automation on flight decks is rising resulting in a more supervisory role for the pilot, instead of direct manual control. Despite these advances, pilots are still often required to take over manual control of the aircraft in landing, takeoff, or during emergency scenarios. An example of the latter could be a computer or sensor malfunction which was the case for Air France flight 447. The crew, startled by unexpected high-altitude dynamics, lost situation awareness despite the information available from the visual and aural displays. Unaware of the loss of the usual flight envelope protections due to the malfunction, pilots stalled the aircraft. The accident report indicates that the stall warning, which was only aural, should be complemented. [7] The control manipulator, the Airbus A330 side stick, did neither provide the pilots with direct feedback on their control actions nor the aerodynamic stall buffets, i.e., it did not help them in properly identifying the situation as a stall. As this tragic example shows, when manual control is needed the lack of haptic information through the control device might contribute to a reduced situation awareness.

Combining the ever-increasing sophistication of automation on the flight deck, and the current generation of small and powerful control devices provides designers a new opportunity, namely to increase pilot awareness through haptic feedback. Some aircraft already include 'augmented forces' on the control device, which can be provided on *both* control devices (on a two-pilot flight deck) linked to the surfaces, or fly-by-wire control systems. An example of this is the "Q-feel force", which changes the stiffness of the controls with changing dynamic pressure/velocity in Boeing type aircraft. [64] Another example is a stick shaker or pusher, which warns pilots of moving closer to extreme aircraft states. [4] The control device can also be loaded with two passive springs to create a change in spring coefficient when pilots exert large control deflections irrespective of the aircraft state, such as in Airbus aircraft. Active control can be used to have an increased (artificial) spring force when rolling beyond the safe roll limit, irrespective of the control surfaces, as used in a Boeing 777. [55]

Although examples of haptic feedback implementations exist, there is limited research published in open literature to prove the benefits of such a system. Within the field of aerospace, Schmidt-Skipiol and Hecker used a passive spring or an active counterforce to communicate the distance to the flight envelope limits. The latter gave best tracking performance increase compared to the no-haptics condition. [50] A second example is the work by Stepanyan et al. that showed the limit on the available control space both visually and haptically. [52] For the haptics, they changed the input neutral point and the maximum deflection, which was used by the pilots to operate the aircraft at the limits. A soft-stop, i.e. a local step in the force required for a certain deflection, can be used to indicate the engine limitations in the collective of a helicopter. It was shown in simulations that such a system can reduce the workload of the pilot ([44, 65]), this was implemented in an experimental helicopter of the German Aerospace Center. [66] Tactile feedback through the use of tactors on a vest enabled improved spatial awareness and reduced spatial disorientation. [29]

These examples use haptic feedback to inform the pilot about the flight envelope limits. Note that research in supplying the pilot with such information is not limited to haptic only, new visual displays are investigated as well and show positive results. [51] Aside from information on the flight envelope limits, the haptic channel can additionally be used to supply guidance support, of which a haptic flight director showed great potential to increase the pilot tracking error and reduce workload. [67] Other fields do have more open domain research interest, for example in tele-operation: the control of

an unmanned vehicle was supported by haptically showing the proximity to objects in its surroundings. It resulted in decreased workload and increased situation awareness for the given navigation task. [68] In the automotive field, haptics can be used on the gas pedal to show the proximity of a car in front, resulting in an increased performance while reducing input magnitudes ([35]); and to support curve negotiation. [37] Petermeijer et al. showed that automotive warning systems reduced the reaction time of the driver while have a potential to induce driver-annoyance, while guidance – for example to the center of the road – improved performance yet is subjective to after-effects. [38]

The aim of the current thesis is to investigate the use of haptic feedback to give the pilot more information on the augmentation with respect to the limits of the aircraft during manual control, within a modern fly-by-wire flight deck. In other words, the design presented in the following aims to provide feedback to the pilot on the proximity of the state to the flight envelope limits. Only longitudinal haptic feedback is considered here, lateral cues can be added in a future design using the same design ideas. This work builds on an initial study ([53]), which showed a potential benefit of such haptic feedback system. The goal of this chapter is to elaborate on a new iteration and give a thorough description on the *how* and *when* of the haptics, as well as the expected practical implications.

Section 2.2 will first discuss some basic flight dynamics and will introduce the control laws and flight envelope protection system present in fly-by-wire Airbus aircraft. Section 2.3 discusses the rationale of our haptic interface, designed to present some of the functions of these automated systems. We then discuss two operational scenarios where the flight envelope protection system will trigger, a windshear and an icing event, to explain in detail how our haptic interface works, Section 2.4. Finally, conclusions are given in Section 2.5.

2.2. FLIGHT DYNAMICS AND CONTROL LAWS

This section provides the background needed to understand the design rationale of our haptic interface. Subsection 2.2.1 covers some basic flight dynamics properties and variables. Readers familiar with aircraft flight dynamics can skip this subsection. As our haptic design focuses on supporting pilots in working with the complex Airbus-like control law and flight envelope protection systems, a brief recap of these systems is provided in Subsection 2.2.2. This recap discusses the (highly-coupled) protections, yet the level of detail is sufficient to support the design of the haptic feedback system in the following.

2.2.1. FLIGHT DYNAMICS

This subsection explains a basic set of flight dynamics variables which are essential to understand the aircraft control laws and the application of the haptics. A full discussion on flight dynamics can be found in literature. [69] The bank angle (ϕ), indicating how much the aircraft wing is tilted with respect to the horizontal plane, is the most important lateral variable and is depicted in Fig. 2.2a. The relevant longitudinal angles are shown in Fig. 2.2b: the pitch angle (θ) depicts the angle of the nose of the airplane relative to the horizon, the flight path angle (γ) gives the elevation of the true velocity vector (V) with respect to the horizon, the angle of attack (α) is the angle of incidence of the air with the

wing section. Accelerations are expressed in the aircraft body reference frame (index *b*); the vertical acceleration (a_z) is commonly expressed in load factor $(n = \frac{a_z}{g})$ and is also shown in Fig. 2.2b. Typical level cruise flight is performed with a load factor of one: lift is equal to weight. By pitching up, the load factor is increased, experienced as 'being pushed in the seat', and visa versa.



Figure 2.2: The aircraft with most important angles indicated (all positive)

Limits of the aircraft are typically expressed in a flight envelope. Different combinations of variables are possible, yet as Airbus pitch control laws are mostly load factordependent, this research considers only the relation between aircraft velocity (*V*) and load factor (*n*). This flight envelope is depicted by the solid black line in Fig. 2.3. The upper velocity limit (right-hand vertical line) is due to the maximum velocity (n_{max}) created by aerodynamic and vibration limits. Extreme load factor values are determined by static structural limits and indicated by the upper (n_{max}) and lower (n_{min}) horizontal lines. The lower velocity limits ($V_{\alpha_{max}}$), the left hand side of Fig. 2.3, follow a quadratic relation with velocity due to the lift equation shown in Equation 2.1, where ρ is the density of the air, *S* is the lifting surface of the wing, and C_L is the lift coefficient:

$$L = \frac{1}{2}\rho V^2 SC_L \tag{2.1}$$

The lift coefficient (C_L) depends on the wing shape and on the angle of attack: the higher α , the higher the lift coefficient, up to a maximum value (α_{max}) where this coefficient suddenly drops and a stall occurs.

2.2.2. AIRBUS CONTROL LAWS

To better understand when the haptic feedback is applied, it is important to understand how pilots control the aircraft. As this research focuses on control laws closely resembling an Airbus A320, the main control device used is discussed: the side stick. All information in the following is retrieved from the A320 Flight Crew Operation Manual (FCOM). [13]

The Airbus side stick is a passively loaded control device: the 'stick feel' is provided by springs and dampers. The device is not mechanically coupled to the control surfaces (ailerons, elevator), it produces an electrical signal to the Flight Control Computers (FCCs) as shown in Fig. 2.4. The latter are responsible for converting the side stick deflections to required control inputs and blending them with the autopilot control commands



Figure 2.3: Flight envelope, allowable load factor (n) versus allowable velocity (V)

to control surface deflections. As such, the FCCs can *override* the pilot inputs and, in doing that, provide an additional layer of safety to keep the aircraft states inside the allowed flight envelope region. This process is called FEP and is applied for both lateral and longitudinal inputs as will be elaborated in the following.





The FCCs consist of seven computers with three functions: 2 Elevator & Aileron Computers (ELACs), normal elevator and stabilizer control, constant aileron control; Spoilers & Elevator Computers (SECs): spoilers control, standby elevator and stabilizer control; 2 Flight Augmentation Computers (FACs): electrical rudder control. These computers are provided with information on the aircraft states by a number of systems and sensors: (i) Air Data and Inertial Reference Unit (ADIRU), (ii) Slat Flap Control Computer (SFCC), (iii) accelerometers, (iv) Landing Gear Control Interface Unit (LGCIU), (v) Radio Altimeter (RA) and (vi) Flight Management Guidance Computer (FMGC).

Five control laws with different levels of support are possible within the Airbus philosophy. This chapter will not discuss in detail when each of the control laws is active. The selection of control law is based on internal sensor validity checks for which more details can be found in the FCOM, Ref. [13]. Only a general description, together with the control laws, follows. We start with 'normal law' which provides the highest level of assistance to pilots, and then move to the configurations which provide less assistance ('alternate law 2

with reduced protections' and 'alternate law without reduced protections'). The 'direct law' and 'mechanical backup' are included for the sake of completeness but will not be used in the sections that follow. A summary of the control laws can be found in Table 2.1, and all the protections in Table 2.2.

NORMAL LAW

When all systems are functioning nominally, the FCCs operate in normal law, the default control mode. The pilot longitudinal and lateral control inputs are both interpreted as 'command' signals for the FCCs, as will be discussed in the following. In addition, the FEP applies a number of protections, to prevent the aircraft from moving outside the – what is considered safe – flight envelope. Examples are the bank angle limitation, the load factor limitation, the pitch attitude protection, the high angle-of-attack protection, and the high-speed protection.

Direction	Normal Law	Alternate Law
Lateral	Bank rate demand maximum 15 °/ s	Bank direct stick-to-surface: clean maximum 30 °/ s otherwise 25 °/ s
Longitudinal	$\overline{C^*}$ control law Autotrim for changing speed or configuration Automatic pitch compensation for $\phi \le \pm 33^\circ$	Control law equal to NL

Table 2.1: Summary of Airbus Flight Control Laws

Lateral control The FCC interprets lateral stick deflections as commands to change the bank angle. From zero to 33° of bank, the side stick lateral deflection is a bank angle *rate* command, whereas the bank from 33° up to 67° is a bank *angle* command. The maximum bank angle rate achievable with full deflection is 15°/s. The FEP in the FCCs limits the maximum achievable bank to 67° which is the first hard envelope limit. If the bank angle exceeds 33°, positive bank stability is present such that the aircraft automatically rolls back to 33° when the side stick is not deflected. Hence, in case the pilot intends to execute a steep turn, a constant stick deflection is required. To assist the pilot during horizontal turns, for bank angles up to 33°, an automatic pitch command is added, such that the pilot does not need to maintain back pressure on the stick to compensate for the required increase in lift.

Additionally, the autopilot disconnects when the bank angle exceeds 45°, at which point the Flight Director (FD) bars (indication of the guidance by the FCC on the PFD) disappear. The bars return when the bank angle reduces below 40°. To prevent excessive Trimmable Horizontal Stabilizer (THS) deflections due to the manual or auto-trim functionality, the deflection is limited between the value on entering of the protection and 3.5° nose-down. Finally, limits for the bank angle depend on the longitudinal protections, which is elaborated in the following. A visual summary of the lateral protections can be found in Fig. 2.5. [13]

Table 2.2: Summary of Airbus Flight Envelope Protection

Variable	Normal Law	Alternate Law		
	THS limited between entry value and 3.5° nose-down			
	Autopilot disconnects when $\phi > 45^{\circ}$			
	FD bars disappear when $\phi > 45^{\circ}$			
	FD bars return when $\phi < 40^{\circ}$	Autopilot disconnects $\phi > 45^{\circ}$		
Bank (φ)	Nominal maximum: 67°			
	Maximum with α_{prot} active: 45°	No other protections		
	Maximum with $V > V_{MO}$: 40°			
	Nominal protection: positive bank stability to $\pm 33^{\circ}$			
	Protection with $V > V_{MO}$: positive bank stability to 0°			
	No limit for autopilot			
	Nose-up maximum, flaps 0 till 3 30°			
Pitch (A)	Nose-up maximum, full flaps 25°			
$1 \operatorname{Herr}(0)$	Nose-down maximum –15°			
	FD bars disappear $\theta > 25^{\circ}$ or $\theta < -13^{\circ}$			
	FD bars return $\theta < 22^{\circ}$ or $\theta > -10^{\circ}$			
	THS limited between entry value and 3.5° nose-down			
	Autopilot disconnects when $\alpha > \alpha_{prot}$			
	Pilot input proportional in region: $\alpha_{prot} \rightarrow \alpha_{max}$			
	Protection deactivates: when 8° forward input			
Angle of attack (α)	or 0.5° forward for 0.5 s when $\alpha < \alpha_{max}$	—		
	Below 200ft, protection deactivates			
	when pilot input is half of previous nose-up input			
	or when $\alpha < \alpha_{prot} - 2$			
	When $\alpha > \alpha_{floor}$, thrust is set to TOGA ¹			
		$5-10$ kts above V_{SW}^2 :		
		a nose-down command is inserted		
Low velocity (V)	_	with bank angle compensation		
(`)		to keep $\rightarrow \alpha$ constant		
		An aural "STALL" warning is provided		
		with 'appropriate margin from stall'		
	THS limited between entry value and 11° nose-up	When $V > V_{MO}$: autopilot disconnects		
	When $V > V_{MO}$: autopilot disconnects	Nose-up command is inserted		
High velocity (<i>V</i>)	pilot nose-down authority is reduced	and can be overruled by pilot		
	an automatic nose-up command is introduced	When $V > V_{MO} + 4$:		
	command cannot be overruled by pilot	and "Overspeed" warning sounds		
Load factor (n_z)	THS limited between entry value			
	and 3.5° nose-down when $n > 1.25$			
	Maximum, clean: 2.5	Equal to normal law		
	Maximum, with flaps/slats: 2	1		
	Minimum, clean: -1			
	Minimum, with flaps/slats: 0			

¹ a_{floor} = 9.5° without flaps/slats (15° for configuration 1 and 2, 14° for 3, and 13° for full), or when pilot inputs 14° pitch up with either pitch or angle of attack protection active.
 ² Speed margin according to Airbus documentation: "Based on the aircraft gross weight and slats/flaps configuration."
 ³ Airbus documentation is unclear on the exact activation point: "at/or above V_{MO}"

2



Figure 2.5: Lateral control in normal law, based on the A320 FCOM [13]

Longitudinal control For longitudinal control, Airbus uses the C^* control law which is a combination of both pitch rate (*q*) and load factor (*n*). [10–13] In the low speed regime, up to approximately 240kts, the pilot stick deflections are predominantly interpreted as pitch *rate* commands; in high speed regions, the stick deflections are mainly interpreted as load factor commands. [12] Due to this setup, there is no need for the pilot to trim the aircraft for changing velocity or configuration.

On top of the C^* control law, protections are present on the pitch angle, the angle of attack, the load factor, and high velocities. The limit on the pitch angle and load factor is without any buffer zone: when approaching the limit, the FCC gradually reduces the pitch rate/load factor until the maximum value is reached and no further control can be achieved. For the other limit, angle of attack, there is a zone from a protected value (α_{prot}) up to the maximum value (α_{max}) where the control device is no longer commanding the C^* control law, yet control device deflections are directly proportional to an angle of attack command above the protection value, i.e., proportional to $\alpha - \alpha_{prot}$. Additionally, the autopilot disconnects when entering the protected region, and the maximum achievable bank is reduced to 45° to prevent asymmetric stall. The throttle input is automatically set to Take Off/Go Around (TOGA) when the angle of attack increases beyond α_{floor} (9.5° without flaps/slats, 15° for configuration 1 and 2, 14° for 3, and 13° for full), or the control device deflection is larger than 14° nose up with pitch or angle of attack protection active.

The angle of attack protection deactivates when the pilot pushes the control device more than 8° forward, or when (s)he pushes at least 0.5s with a deflection of minimal 0.5° forward when the angle attack is below the maximum value. Below 200ft, the protection is also deactivated by when the pilot uses less than half of the previous nose-up input, or when the angle of attack is less than $\alpha_{prot} - 2^\circ$.

For all three limits, the maximum value, and if applicable the size of the position control zone, depends on the particular flight conditions and the state of the aircraft. The pitch angle limits are between -15° and 30° (25° for full flaps/slats), load factor must remain between -1g and 2.5g without flaps/slats (0g and 2g for any other configuration), and the angle of attack must remain less than 12°, with a protection zone of 2°. The buffer zone for the angle of attack is shown on the PFD through the velocity indication,



Figure 2.6: Flight envelope with A320 longitudinal Flight Envelope Protection limits indicated

whereas no indication for the load factor is available in the current Airbus setup on the primary display. Informing the pilot that this buffer is reached and that pilot inputs are being limited, could result in a more transparent system, increasing pilot awareness of the flight envelope limits. Note that the pitch limits do not apply to the autopilot, and that the FD bars disappear when the pitch increases above 25° nose up or below 13° nose down and return when the pitch is between 22° nose up and 10° nose down.

To prevent structural damage when controlling the aircraft at high velocities, a highspeed protection is present. This protection triggers at (or above, depending on the configuration) the minimum of both maximum operational velocity (V_{MO}) and maximum operational mach (M_{MO}), disconnects the autopilot, and activates an automatic nose-up command while reducing the nose-down stick authority to reduce the airspeed below the maximum, effectively creating an artificial high-speed stability. The pilot is warned of the overspeed condition by an aural message, yet the nose-up command is *not* communicated to the pilot. Enabling the pilot to know (or feel) how it is implemented can be an addition to improve pilot awareness of the flight envelope protection system. Note that the nose-up command cannot be overridden by the input of the pilot, even with full forward deflection. In order to ease returning to normal operational velocities and to avoid high structural loads during an overspeed, the positive bank stability shown in Fig. 2.5 rolls the aircraft back to 0° and the maximum bank angle is limited to 40°. These additional limitations are present until the velocity drops below the maximum velocity.

As for the lateral control, here the THS is limited to prevent excessive deflections. During the angle of attack protection and load factor values above 1.25g, the limits are the entry value and 3.5° nose-down. When the high velocity protection is active, maximum values are the entry value and 11° nose-up.

Fig. 2.3 already showed the nominal flight envelope. Here, we discussed the angle of attack (related to velocity through Equation 2.1), load factor and high velocity protections present in the A320 control laws. The angle of attack protection can be visualized on the flight envelope as shown with the red-dashed line on Fig. 2.6, where every state where no protection is active is defined as belonging to the safe flight envelope. As can be seen in the figure, the zone where a protection is active provides a buffer for the pilot when approaching the limits.

ALTERNATE LAW

In case of sensor or computer failures, the FCC reverts to control laws which provide less support for pilots. The first of these degraded control laws is the alternate law, with reduced protections, which triggers, for example, when a dual computer failure is detected.

Lateral control Lateral control becomes a direct stick-to-control-surface-position relationship with maximum roll rate of 30 $^{\circ}$ / s for clean configuration and 25 $^{\circ}$ / s otherwise. Hence, positive bank angle stability and bank angle protections are lost. Furthermore, if the autopilot is engaged, it disconnects at 45 $^{\circ}$, requiring the pilot to take over control.

Longitudinal control The longitudinal control law is not changed. One major change with respect to normal law, when considering safety, is the loss of the angle of attack and pitch protection. This includes the buffer zone described before, as well as the protection against excessive control inputs. Load factor protection is present equal to normal law.

Too large angle of attack angles can lead to an aircraft stall event, and pilots are trained to avoid this event in all circumstances. Most of the time the aircraft flies in normal law, and the aircraft simply cannot stall. But in the very rare situation that the normal law is deactivated and the degraded control laws become active, the angle of attack protection is lost. Pilots may fail to notice this control law degradation, and the corresponding loss of protection, which could lead to a stall. This possibly catastrophic event will be taken into consideration in our haptic interface design.

To assist the pilot in this control law, a region with low speed stability is introduced by Airbus. Dependent on the configuration, 5 to 10kts above the stall warning speed (V_{SW}) a nose-down signal is introduced. Additionally an aural "STALL" warning is added with, according to the Airbus documentation, 'appropriate margin from stall'. Furthermore, during a turn, pitch compensation is present to maintain a constant angle of attack.

The high-speed stability from the normal law remains, yet the pilot is now able to overrule the imposed nose-up command. Autopilot disconnection occurs when the velocity exceeds V_{MO} , and at V_{MO} + 4 an aural "OVERSPEED" warning is present.

ALTERNATE LAW WITHOUT REDUCED VELOCITY PROTECTIONS

In some cases, for example when all three air data reference units fail, the control laws further degrade and have even less protections. Both lateral and longitudinal control laws remain equal to alternate law with the protections, except that the low- and high-speed stabilities are lost. The load factor limitation does remain available for the pilot.

DIRECT LAW

When all three inertial reference units fail, the RAs fail when the landing gear is down, or when flaps are selected while the LGCIUs disagree, the control law is reverted to direct law. In direct law, control surface deflections become equal to side stick inputs.

Lateral control Although direct stick-to-roll is low-level control, the FCC still aids the pilot inputting the right magnitude of inputs by scaling the control gains based on the configuration. Yaw damping and turn coordination are lost in this case, as is the maximum bank angle protection.

Longitudinal control Stick-to-pitch direct control is aided by scaling the control gains depending on the center of gravity location of the aircraft. In this control law, no protections are active and the pilot can therefore bring the aircraft outside the flight envelope limits.

MECHANICAL BACKUP

When a complete loss of electrical power occurs, the side stick is unusable due to the transducers used in the design, and the loss of the FCCs. Therefore a mechanical backup is available which is a very basic and crude control.

Lateral control Lateral control is achieved solely by operating the rudder pedals, without any direct bank control. Rolling is achieved due to the coupling of yaw and roll, but as this is a slow response and Airbus indicates in Chapter OP-20 of Ref. [70] to: "*Gently apply an input and wait for the response*". Care should be taken to not exaggerate the input as to not over-control the aircraft.

Longitudinal control The mechanical backup for the pitch control is by manually trimming the horizontal stabilizer. Again, this provides a slow control method and should be executed with caution.

Now that the basic flight dynamics are discussed, and the Airbus flight control philosophy has been summarized, we can move to the design of our haptic interface. That is, how can we use haptics to assist pilots in maintaining situation awareness of the state of the aircraft and the automation, especially in high workload situations when the aircraft operates close to the flight envelope limits?

2.3. HAPTIC DISPLAY DESIGN

This section describes the haptic display that is used to show the flight envelope boundaries to the pilot. First, the definition of haptic feedback in this research is shown, followed by the goal of the support system. Next, the information is used to elaborate on *how* and *when* haptic feedback is provided in the current design. Note that the values for all tuning parameters introduced in the following are summarized in Table 2.3.

2.3.1. HAPTIC FEEDBACK DEFINITIONS

Haptic feedback can be considered as a process that deliberately changes the feel of the control device. This research focuses on changing the haptic profile, i.e., the relation between the deflection of the control device (δ) and the amount of force required to do so (*F*). A default profile for many sticks (and other control manipulators such as rudder pedals) is a piece-wise linear relation as shown in Fig. 2.7. Here, δ_{np} is the position of the control device when no force is applied, referred to as the neutral point. The location of the break-out zone is given by $\pm \delta_{br}$, here the stick has a spring coefficient k_{br} . The breakout zone is included to haptically show pilots where the 'zero stick deflection' position lies. Outside this zone, k^+ and k^- are the spring coefficients for, respectively, positive and negative control device deflections. The default case for this design, as

for the Airbus side stick¹, is a symmetric profile using a nominal stiffness (k_{nom}) for positive and negative deflections until a maximum deflection (δ_{max}). Such a haptic profile provides the pilot with information on the input magnitude: larger inputs require larger forces. Deviations from this default haptic profile can be used to provide the pilot with additional feedback through the control device. Although not considered here, haptic feedback can also be considered by changing the dynamic properties of the control device such as the natural frequency, damping coefficient, static friction (force required to move from a stand-still), dynamic friction (friction due to movement), or other nonlinear phenomena. [57]



Figure 2.7: Nominal control device profile: required force exerted (*F*) versus stick deflection (δ)

Literature shows different ways of changing the haptic profile: in the automotive field there is a strong focus on using a forcing function which can be used both as a *warning* signal ([36, 71]), or as a *guidance* force. [35, 38]. Aerospace applications show examples which adds a soft-stop (a local step in the amount of force required), a hardstop (a change in maximum deflection, [62]), forcing functions ([42]), changes in the stick neutral position ([52]), and changes in nominal stick stiffness. [46, 47] An example of haptic feedback in the current Airbus A320 flight deck is the detent present on the thrust levers: the controls 'clicks' in the important thrust positions (such as maximum continuous trust, of take off/go around setting) and requires a threshold force to move away from this position.

2.3.2. GOAL OF SUPPORT SYSTEM

We aim to use haptic cues to provide pilots with information on whether the aircraft approaches the limits of the FEP: increase situation awareness. In Subsection 2.2.2 we discussed how moving the aircraft outside the safe flight envelope, shown in Fig. 2.9, can lead to changes in the control law. For instance in normal law, when exceeding the protected angle of attack, the control laws change from C^* to α -command control. So in principle the fact which control law is active does provide some information on the proximity to the boundaries of the flight envelope. Nevertheless, pilots must *infer* this from the changing aircraft reaction to control inputs or from the velocity indication on the

¹After publication it was note that the Airbus side stick has two spring coefficients: extreme deflections have an increased k^+ / k^- . This is included in the designs in Chapter 5, 6 and 7.

PFD when flying in normal law. To present this more clearly, the haptic support system will include the Airbus protection features expressed with haptic cues. In addition, we will explore how potential mitigation control strategies can be suggested by the haptics, e.g., by making clear what control actions are desired or undesired. The following haptic cues are added and will be discussed in detail in the indicated subsections:

- 1. A square pulse displaying the transition from in- to outside the safe flight envelope, i.e., crossing the red-dashed line indicating the safe flight envelope on Fig. 2.9 (Subsection 2.3.4).
- 2. Change in spring coefficient for positive or negative positions relative from the distance from the safe flight envelope to the limit, i.e., the distance between the red-dashed and the black line on Fig. 2.9 (Subsection 2.3.5).
- 3. Changing the neutral point to indicate the automatic control input by the FCC in case of overspeed, i.e., right of the high velocity protection line on Fig. 2.9 (Subsection 2.3.3).
- 4. Changing the neutral point to indicate a neutral stick position is not sufficient for low velocities, near the location of the inset on Fig. 2.9 (Subsection 2.3.3).
- 5. A stick shaker at critical low velocity, i.e., left of the green dash-dotted line on Fig. 2.9 (Subsection 2.3.4).

The result on the system architecture is a dependency of the control device properties on the aircraft states and FEP through the Haptic Feedback Law (HFL), shown on Fig. 2.8. Note that the HFL is not dependent on the current control device state: the haptic display shows *when* the limits are near, not the control device position *where*. Information on the limits is assumed to be calculated by an external model and are therefore not discussed in this chapter. For this research project, a proprietary Airbus A320 model created by the German Aerospace Center (DLR) has been made available. More information on the control laws, flight envelope and the corresponding protections can be found in Refs. [54, 72].



Figure 2.8: Block diagram representing the Airbus control loop combined with the Haptic Feedback Law (HFL)

This chapter discusses only longitudinal haptic feedback. Furthermore, it is assumed that the pilot is flying with hands on the controls, which is verified in conversations with pilots to be a common airline procedure below an altitude of 10,000ft, and in emergency situations. Additionally, Airbus specifies three phases in flight with different control modes: on the ground, during flare and in flight. [13] In this research, only 'flight mode' is considered. The transitioning modes during flare and on the ground, as well as lateral haptic feedback are left to a next iteration.



Figure 2.9: Flight envelope, load factor (n) versus velocity (V), inset for Fig. 2.11

Note that the flight envelope used for the design of the haptic display presented in Fig. 2.9, has three differences with respect to the flight envelope for Airbus control laws shown in Fig. 2.6. First, we decreased the upper aircraft velocity limit in the safe flight envelope, and provided a buffer of 20kts ($V_{MO_{prot}} = V_{MO} - 20$). Second, to complete the buffer zone towards the hard flight limits, we added a buffer on the load factor of 0.5g. Third, we implemented a critical low velocity zone, which will be communicated through the use of forcing functions.

In normal operations, the aircraft is operated within the safe flight envelope in the normal control law. In case of abnormal situations, as discussed in Subsection 2.2.2, the aircraft reverts to an alternate law in which fewer protections are active and the pilot has more control to move outside the flight envelope. In the current stage of our project, the haptic display is designed such that in both cases – normal and alternate law – the haptic settings are identical. The full haptic display can still be applied in alternate law because the intensities of the cues will be chosen such that pilots can always overrule the haptic signals: they have the final authority of the side stick. Hence in both conditions, in case the aircraft is maneuvered outside the safe flight envelope, the haptic cues are designed such that they should *support* the pilot in identifying the situation, and deciding on an effective mitigation strategy, to keep the aircraft safe.

The remainder of this section elaborates more on *how* and *when* the haptic cues are provided.

2.3.3. Change the position of the neutral point

The position of the neutral point can be changed through manipulating the value of δ_{np} . If applied, the information provided by the haptic display is directly proportional to a required control command, and in principle the pilot can 'just follow the position'. Previous research showed that using such an approach increased tracking performance while reducing the physical effort. [42] If the pilot does not agree, however, (s)he can choose to override the cue, and keep the stick position fixed by actively counteracting, using co-contraction of the muscles. [37] Nevertheless, the shift in neutral position gives a clear message to the pilot on what (s)he should do. The effect of this change in neutral position on the profile can be seen in Fig. 2.10 by the shift of the entire graph to the right.



Figure 2.10: Haptic profile with a positive shift in the neutral point position

In the Airbus' philosophy, a zero stick deflection commands load factor of one. This is a safe and desired load factor for most of the flight, but in some cases a different load factor is needed to return to the safe flight envelope. To indicate this, the neutral point can be altered. Looking at the flight envelope in Fig. 2.9, two such regions can be identified: (i) in case of overspeed an active pull-up is required, and (ii) at g-loadings for low velocities since the maximum safe load factor is below one. The next sections therefore investigate this required load factor (n_{req}) for both situations respectively, followed by the translation of the required load factor to the required change in stick neutral position.

OVERSPEED

When an overspeed occurs, the speed has to be reduced actively by the pilot by either reducing the throttle, or by pitching up such that kinetic energy is exchanged for potential energy. The Airbus flight envelope protection system will implement a forced nose-up command (see Subsection 2.2.2), which could be translated to a change in neutral point. Nevertheless, the actual implementation of this signal is not known for this research and is approximated as described below. The main reason for this cue is to inform the pilot that maintaining the stick at zero deflection does *not* solve the flight envelope violation, and action needs to be taken. Note that here our research deviates from the A320 FEP: the nose-up command is not activated when crossing V_{MO} , it is already activated when crossing V_{MO} , it is already activated when crossing V_{MO} .

For this research, the nose-up command, and therefore the magnitude of the neutral point shift, is governed by the change in load factor required to bring the positive accel-

eration to zero. It is determined by starting from the longitudinal equations of motion ([69]), where we assume engine thrust to be parallel to the x axis of the aircraft body frame:

$$T\cos(\alpha) - D - W\sin(\gamma) = m\frac{dV}{dt}$$
(2.2)

From all variables in this equation, the pilot can manipulate the aircraft flight path (γ), through moving the stick. Here, the neutral point is shifted to obtain a flight path angle such that there is no positive acceleration, $\frac{dV}{dt} = 0$. Since the aircraft is accelerating, the left part of Equation 2.2 is not zero and can be rewritten to obtain a steady flight path:

$$\gamma_{\text{steady}} = \arcsin\left(\frac{T\cos\left(\alpha\right) - D}{W}\right)$$
 (2.3)

Thrust and drag cannot be measured directly, their effects can be measured through accelerometers, mounted on the aircraft body, which therefore must be rotated to the aerodynamic reference frame:¹

$$T\cos(\alpha) - D = ma_{x_a} + W \cdot \sin(\gamma)$$

= $m(a_{x_b}\cos(\beta)\cos(\alpha) + a_{y_b}\sin(\beta) + a_{z_b}\cos(\beta)\sin(\alpha)) + W \cdot \sin(\gamma)$ (2.4)

Combining Equation 2.3 with Equation 2.4 then yields the required change in flight path angle for zero acceleration ($\gamma_{\text{steady}} - \gamma$), all expressed in measured quantities.

As discussed above, the side stick gives load factor commands for high velocities and therefore also a relation between the change in flight path angle and load factor is required. By assuming that the steady state pitch rate is predominantly determined by a change in flight path, the required load factor can be expressed by the required flight path angle and a tuning factor ($\tau_{overspeed}$) which is chosen as a measure of the recovery speed: [73]²

$$n_{\text{req}} = \frac{V}{g} \cdot q + 1 \approx \frac{V}{g} \cdot \dot{\gamma} + 1 = \frac{V}{g} \cdot \frac{\gamma_{\text{steady}} - \gamma}{\tau_{\text{overspeed}}} + 1$$
(2.5)

G-LOADING FOR LOW VELOCITIES

As mentioned before, the stick neutral position commands a load factor of one. In case the aircraft velocity becomes too low – that is, too far to the left on Fig. 2.9, and zoomed in shown by Fig. 2.11 – returning to load factor 'one' is not sufficient to re-enter the safe flight envelope, and the pilot has to be informed that action is required.

This is done by shifting the stick neutral point. The prerequisites for this cue are that: the current safe load factor is below one (the green circle on Fig. 2.11) and the current load factor is above the safe load factor. Note that the current load factor is measured by sensors, and it is assumed that the aircraft (model) calculates the safe load factor. The required load factor to return to the safe flight envelope therefore *is* the safe load factor itself ($n_{max_{neut}}$).

¹The term due to gravity was not included in the journal publication. After publication, it was found that the lack of this term resulted in an incorrect calculation of the required flight path angle when climbing/descending.

²The journal publication contained a mistake in the equation derivation and is correct for this thesis.



Figure 2.11: The amount of load factor change required when flying at low velocities, inset from Fig. 2.9

CHANGE IN NEUTRAL POINT IMPLEMENTATION

For the two cases discussed above, a required load factor is calculated, which needs to be shown to the pilot using a change in stick neutral deflection. Since zero stick deflection indicates a required load factor of one, the required shift in neutral point ($\Delta \delta_{req}$) given a required load factor (n_{req}) can be determined using:

$$\Delta \delta_{\rm req} = \frac{\delta_{\rm max}}{n_{\rm max} - 1} \left(n_{req} - n \right) \tag{2.6}$$

In case this required change in neutral point would be implemented immediately, abrupt changes in the control feel can be observed. The change in neutral point would then be perceived more as an 'alert', rather than a guidance cue. Therefore the required change in neutral point is ramped-in linearly using an iterative formula which can be easily implemented in software. With the previous neutral position (δ_{np}_{prev}), the time difference with the previous step (Δt), and the rate ($\dot{\delta}$), the current neutral position (δ_{np}) is calculated using:

$$\delta_{np} = \min\left(\Delta\delta_{\text{req}}, \delta_{np_{\text{prev}}} + \Delta t \cdot \dot{\delta}\right)$$
(2.7)

2.3.4. ADD A FORCING FUNCTION TO THE DEVICE

When a forcing function is added to the control device, the whole force/position profile is shifted vertically up or down. Depending on the magnitude of the cue, and whether the pilot is holding the stick or not, it can change the control device deflection as shown in the illustrative example in Fig. 2.12a. In our design we intend to use this cue mainly to alert or warn the pilot, and not to impose a required control input. Hence, the forcing function should be of small period, or small amplitude. The effect is therefore short and not pre-defined in terms of deflection: the effect can be difficult to grasp in one snapshot of a haptic profile. As such, the time trace of the forcing function is visualized by an added graph as shown in Fig. 2.12b, where zero time is current, and times to the right represent past times. A pragmatic approach is used to evaluate whether the cue complies with the assumption of small period or amplitude. In the current design, two forcing functions 2

are used: a discrete cue to communicate that the safe flight envelope is exited and a stick shaker to alert for low velocities.



Figure 2.12: Haptic profiles showing the addition of a forcing function

Note that the addition of a forcing function and a pure change of neutral point both result in a change of position of the control device. Nevertheless, their driving principle is different: a forcing function does not have a predefined effect on the control position. The effect depends on the position of the control device in the haptic profile, and the pilot's arm stiffness. In contrast, a pure neutral point shift results in one desired control input, to guide a pilot through a maneuver.

DISCRETE CUE

Discrete cues are limited in time and can have a wide variety of shapes, ranging from a square block signal to a noise input. They can be a useful tool to warn the pilot of entering a certain region, while not giving a constant signal. The intent of this cue can be compared to a softstop: an indication of entering a region where caution is required. For example, a softstop can indicate a position *where* the maximum engine limits are exceeded. [44, 74] In contrast, a forcing function is added to the controls *when* the (protection) limit is exceeded. The forcing function is chosen in the design as it does not have a dependency on the state of the control device, as do all haptic cues used in the design.

One region which can be entered, with or without the intention of the pilot, is the protected region close to the edge of the flight envelope shown on Fig. 2.9 by the dashed line, corresponding to the buffers created on α , n, and V. An example where entering this zone can go unnoticed by the pilot, is when he/she is busy scanning the instruments, or involved in other tasks. Therefore, to provide a clear transition cue when exceeding the safe flight envelope, a warning cue in the form of a square pulse signal (width 0.1s, magnitude 10N, shown in Fig. 2.13a) is given. This shape and intensity of the forcing function was chosen based on a preliminary test with a single test-pilot, future research is needed for the further definition of this shape. By adding this cue, the pilot is triggered about the safe flight envelope departure and the attention is drawn to the event.

The direction of the cue should indicate the direction of the 'correct action' for the pilot to perform if (s)he intends to solve the limit violation. For this reason a stick forward

cue ('pitch down' indication) is given for extreme positive load factors, high angles of attack for positive load factors, and low velocity violations. A stick backward cue ('pitch up' indication) is given when crossing all other boundaries.

STICK SHAKER

A periodic cue is a signal which repeats itself in time, and can be used as a persistent way to alert pilots of an imminent critical state. An example is the motor priming used by Navarro et al. ([36]) to warn drivers of a lane departure. Analogous to this event, in aerospace exceeding the maximum angle of attack should be avoided at all times. Hence, to bring extra attention to the proximity to stall, a second forcing function is added: a stick shaker following a sinusoidal forcing function with a frequency of 20Hz and 5N amplitude, shown in Fig. 2.13b. The frequency and amplitude is tuned to match the stick shaker present in other aircraft (such as Boeing [55]), and was initially designed to represent the aerodynamic buffeting on the control surfaces.

The stick shaker is activated when the aircraft velocity drops below half of the protected range (hence $\frac{V_{a_{max}} - V_{a_{prot}}}{2}$). In terms of the flight envelope, this means that close to the left-hand limits of the flight envelope, indicated on Fig. 2.9 with a stripe-dotted green line, the stick shaker activates. This cue is additional to the existing flight envelope protection as described in Subsection 2.2.2, yet is intended to clearly indicate to pilots that the aircraft is moving closer to the lower velocity limit.





(b) Stick shaker

(a) Stick push function

Figure 2.13: Continuous Forcing functions used

2.3.5. CHANGE IN SPRING COEFFICIENT

Previous research increased the spring stiffness to indicate that continued control inputs would result in a hazard, effectively reducing the occurrences of imminent pilot-induced oscillation ([46, 47]), signal a lagging adaptive controller ([75]), or indicate a helicopter main rotor setting below the limit. [74] In our design, a continued control input results in a hazard when it brings the aircraft closer to the limit.

Looking at the flight envelope, it is not just *any* input that poses a hazard, it is *one direction* of input which worsens the situation. For example, when the aircraft is close to an overspeed condition, pushing the stick results in a state closer to the actual overspeed, pulling on the controls is a possible mitigation strategy. Therefore, to show the undesired input, a continuous single-sided spring cue is used resulting in a haptic profile as shown in Fig. 2.14c. In this figure, the positive (push) deflection requires more force indicating an unwanted input as in the examples above, a negative (pull) deflection is easier to obtain as the spring stiffness is equal to the nominal value. Note that a change in spring

coefficient is only noticeable to pilots when they move the stick away from the neutral point, hence when the pilot is actively controlling. This haptic cue does not necessarily change the control input itself.

Similar to the further spring coefficient increase when the adaptive controller increases lags ([75]), increasing the spring coefficient can additionally be used to communicate the magnitude of the safe flight envelope excursion. As such, starting at the edge of the safe flight envelope, the red dashed line in Fig. 2.9, up to the the edge of the nominal flight envelope, the solid black line in Fig. 2.9, the stiffness is increased. For the load factor, the velocity and the angle of attack, we use v as generic symbol, the default stiffness is multiplied with a factor K_k , determined by the gain K_v and the severity of the violation:

$$K_{k} = \begin{cases} 1 & \text{if } v < v_{prot} \\ 1 + K_{v} & \text{if if } v > v_{nom} \\ 1 + K_{v} \frac{v - v_{prot}}{v_{nom} - v_{prot}} & \text{else} \end{cases}$$
(2.8)

The severity is defined as the ratio of the violation of the safe flight envelope, $v - v_{prot}$, where v_{prot} is the value at the edge of the safe flight envelope, and the distance between the safe and nominal flight envelope, $v_{nom} - v_{prot}$, where v_{nom} is the value at the edge of the nominal flight envelope. To guarantee that the pilot has the final authority of the side stick, the stiffness does not increase when the state exceeds the nominal flight envelope. The haptic display is defined to trigger on the maximum (α_{max}) and protected (α_{prot}) angle of attack instead of the lower velocity, nevertheless these variables are related through Equation 2.1.

To illustrate the working principle in the overspeed condition mentioned before, Fig. 2.14 shows three instances where the velocity is outside the safe flight envelope. Fig. 2.14a represents a situation where the severity is 0.2: the velocity is slightly over $V_{MO_{prot}}$. If the situation gradually evolves, an increased velocity results in Fig. 2.14b which shows an increased single-sided stiffness with a severity of 0.5. Finally Fig. 2.14c shows a condition at or above V_{MO} where the spring stiffness is maximal, severity is one.



Figure 2.14: Haptic profiles showing progressively increasing positive spring stiffness (k)

The direction of the stiffness cue is inversed from the discrete cue. That is, the stick feels 'stiffer' for *backwards* movement in cases of extreme positive load factors, high angles of attack for positive load factors, and low velocity violations. All other violations of the safe flight envelope lead to increased stiffness for *forward* movements. As such, the direction of the stiffness cue informs pilots of control actions which bring the aircraft closer to its limits, the discrete cue informs which action can resolve the current situation.

As the stiffness changes with continuous variables, no sudden changes in stiffness should occur. Nevertheless, if it occurs, a large change in stiffness could be observed by pilots as a forcing function, an alert, not a continuous guidance cue. Therefore, to guarantee a smooth change in stiffness, the change is ramped-in linearly, similar to Equation 2.7.

Table 2.3: Summary of design parameters for control device and haptic settings

Kα	K _n	$K_{V_{MO}}$	k_{nom}	δ_{max}	k_{br}	δ_{br}	$\tau_{overspeed}$
3	2	2	1 N/deg	18 deg	50 N/deg	0.05 deg	15

2.4. OPERATIONAL TEST SCENARIOS

Two relevant operational scenarios will be discussed in this section, which were chosen because we expect that our haptic interface can provide pilots intuitive and useful information to deal with these events, both scenarios are based on Ref. [54]. The first example describes a case in which pilots are required to maneuver close to the edges of the flight envelope limits: a wind shear. The second example shows how pilots can use the system when the flight envelope is shrinking, and the envelope limits approach the current aircraft status, ultimately limiting pilots in their control: icing.

For both scenarios we will discuss the origin of the event, the required (or: desired) actions to be taken by the pilot, and how we expect that the new haptic system supports the pilot in deciding and performing the necessary actions. In addition, Subsection 2.4.3 discusses some possible undesired actions, as a brief introduction to what our experimental depend measures may look like.

2.4.1. WIND SHEAR: AIRCRAFT OPERATES NEAR THE LIMITS

A wind shear is a meteorological phenomenon in which locally wind velocities are rapidly changing. Such event can be caused by multiple sources such as strong surface winds, weather fronts, and convective storms. Severe wind shears can occur when a large cylinder of air suddenly "drops" towards the earth due to convective weather conditions, sometimes referred to as microbursts. [76] When this cylinder plunges on the earth surface, the air spreads out as illustrated in Fig. 2.15, with the numbers in circles corresponding to those used in the text below, and figures that follow. If an aircraft flies through the windfield, the headwind initially causes its airspeed to increase as in (2). When the pilots do not recognize the windshear and fail to take action, the downwind that follows will push the aircraft towards the ground (3) and (4). The next tailwind drastically reduces the velocity (5). Near the final stage of the recovery, the aircraft is flying with high throttle settings and almost level flight, a potential problem is an overspeed (6). At the end of this event, the pilots hopefully are able to return to normal flight (7). All things considered, windshear forms a severe risk to the safety of the flight, especially during take-off or landing when already close to the ground. [77] Throughout the recovery of a severe windshear, is vital that pilots use all available aircraft performance to climb irrespective of forward velocity, with one catch: the aircraft should not be stalled.



Figure 2.15: Weather structure during wind shear

If this event occurs with the autopilot active, most actions are handled automatically while the pilot maintains a close watch on the autopilot actions. Here we focus on manual flight control, and the autopilot is assumed to be turned off. The pilots must perform a set of actions, put forward by the manufacturer as described in the FCOM. [13]

The initial warning for the pilot of the oncoming event is a visual and an aural warning: a red "WINDSHEAR" message on the PFD and a synthetic voice which announces "Windshear" three times. At this point the FCOM states that the pilots must take the following six actions:

- A Do not change configuration (flaps, slats, gear) until out of the windshear,
- B Throttle levers at go around position,
- C Initial pitch attitude of 17.5°,
- D Increase pitch if necessary to minimize loss of height above terrain,
- E Closely monitor flight path and speed, and
- F Recover smoothly to normal climb out of shear.

The first step is a straightforward command, to make sure that no time is lost before starting the recovery. Next, one must assure that maximum energy is available, step B, followed by an initial pitch attitude to start increasing altitude, step C. Then, steps D and E are crucial to the safety of the aircraft: here we see a trade-off between on the one hand reducing altitude loss and on the other hand maintaining sufficient airspeed. In case of an extreme wind shear, this recovery procedure might require pilots to move dangerously close to the limits of the flight envelope, namely at very low velocities as to use all available energy to climb out of the shear. The final step, F, assures that, when clear of the dangerous winds yet still with high throttle settings, the aircraft velocity does not exceed the upper limit.

Throughout the procedure, pilots are likely to work under high workload levels and could develop a mental state of "cognitive tunnelling", heavily monitoring the loss of height. [78] A support to improve the attention division of the pilot is of crucial importance and we show below that our haptic interface can enable this.

The trajectory in the flight envelope of the seven selected time frames is shown in Fig. 2.16. For each frame, the left column of Fig. 2.17 shows the aircraft flight envelopes, with the current aircraft state shown using a circle. The center column shows the cor-

responding PFDs, with velocity (left), altitude (right), and attitude (center). The right column of Fig. 2.17 shows the haptic profile. These frames are used here to show how the haptic interface is working during operations. For example, the first of these frames is the starting point when the windshear-warning becomes active, the corresponding flight envelope, PFD and haptic profile can be seen in, respectively, Fig. 2.17a, 2.17b and 2.17c.



Figure 2.16: Trajectory through the flight envelope during windshear recovery with frames indicated in Fig. 2.15

After the warning is given, our haptic interface is expected to help in the following steps (corresponding to the list of actions stipulated by the FCOM):

- C. The pilot must pitch-up the aircraft and this increases the load factor somewhat (Frame 2). If this maneuver is executed too fast, the pilot is informed of the g-loading limit through the load factor protection cues.
- D. During this step as much energy as possible should be used to climb, and the haptic system is expected to help pilots to operate at or close to the flight envelope limits. The initial cue of approaching limits is the discrete cue, corresponding to Frame 3.
- E. As the pilot has to divide attention over two elements of the PFD (the velocity and altitude indicators) and possible cognitive tunneling may develop on the vertical speed, the haptic system is expected to serve as a velocity monitoring aid. This can be achieved by both the continuous spring cue and the change in neutral position for low velocities, as illustrated in Frames 4 and 5 in Fig. 2.17, respectively. Additionally, we expect pilots to use the stick shaker as a possible control aid to "ride the stick shaker": adjusting the input such that the stick shaker remains on the verge of activation.
- F. When approaching the upper limit on velocity, the high velocity cues alert the pilot of imminent limit violation with an extra control aid by where the pilot can follow the stick backwards position, shown in Frame 6 in Fig. 2.17.

In general, for each of the steps the discrete haptic pulse cue (Frame 3) is expected to first *alert* the pilot that a flight envelope limit is approaching, and then the continuous spring cue (Frame 4) follows to clearly communicate the distance left to the ultimate flight envelope boundaries.



42

(a) Flight envelope for Frame 1



(d) Flight envelope for Frame 2



(g) Flight envelope for Frame 3



(j) Flight envelope for Frame 4



(m) Flight envelope for Frame 5



(b) PFD for Frame 1

(e) PFD for Frame 2



(h) PFD for Frame 3



(k) PFD for Frame 4



(n) PFD for Frame 5



(c) Haptic profile for Frame 1



(f) Haptic profile for Frame 2



(i) Haptic profile for Frame 3



(l) Haptic profile for Frame 4



(o) Haptic profile for Frame 5

Figure 2.17: Flight envelopes, PFDs, and haptic profiles for the windshear recovery according to the frames from Fig. 2.15



2.4.2. ICING: LIMITS MOVE TOWARDS THE AIRCRAFT

The second scenario we will use to evaluate our haptic interface is an extreme form of ice formation on the aircraft wings. Especially when flying through cold humid air, the risk of such an event is severe. [79] The effect of ice formation is a degradation of the aircraft aerodynamic performance, resulting in a reduced lift from the wings, and with that an increase in the aircraft minimum velocity. Here, it is assumed that the FEP has an updating algorithm which is presented with icing implementation in Ref. [54]. The decrease in minimal velocity highlights the main difference with the previous scenario: in this case the flight envelope *shrinks*, the flight envelope limits 'approach the pilot', and (s)he must identify this situation properly and act on it.

An example case of such an event is during a manual instrument landing in which the landing is performed in the clouds. If extreme ice accumulation is present, or when the de-icing system is not working properly, the ice formation is inherently a slow yet detrimental process. It is very likely that, for considerable time, pilots may not be aware of the deteriorating aerodynamic properties.

In principle, pilots can notice the degradation of the aerodynamic properties due to icing through two clues. First, the increase in drag requires a higher throttle setting, and second, the decrease of lift requires a higher angle of attack.

Especially when the pilot is flying with the auto-thrust active, the increase in throttle setting can be more difficult to notice and, as Airbus aircraft by default do not have an angle of attack indicator, pilots might be unaware of the creeping danger. Nevertheless, the haptic feedback system uses information on the angle of attack sensor, and therefore the pilot will get new information without adding another element on the, already comprehensive, visual display.



(a) Flight envelope for Frame 1



(d) Flight envelope for Frame 2



(g) Flight envelope for Frame 3



(j) Flight envelope for Frame 4



(m) Flight envelope for Frame 5



(b) PFD for Frame 1

(e) PFD for Frame 2



(h) PFD for Frame 3



(k) PFD for Frame 4



(n) PFD for Frame 5

 $F[N] \xrightarrow{F[N]} \delta_{br}$ $k^{+} \xrightarrow{k^{+}} \delta_{[deg]}$

(c) Haptic profile for Frame 1



(f) Haptic profile for Frame 2



(i) Haptic profile for Frame 3



(l) Haptic profile for Frame 4



(o) Haptic profile for Frame 5

Figure 2.18: Flight envelopes, PFDs, and haptic profiles for an icing event illustrating shrinking of the flight envelope

To illustrate how the state is developing and how the haptic feedback is supplying flight envelope information, Fig. 2.18 shows five frames of an icing event in which the flight envelope is shrinking. The left column shows the aircraft flight envelopes, with the current aircraft state shown using a circle. The center column shows the corresponding PFDs, with velocity (left), altitude (right), and attitude (center). The right column shows the haptic profile. Starting from the nominal condition in Frame 1, icing forms and the minimal velocity is increasing as stated before. If the pilots do not react to this, the first signal from the haptic display is the discrete cue when exiting the safe flight envelope as in Frame 2. At this point, the pilots should become aware that something is going on. Additionally, they have received the correct action by the direction of the cue: reduce the angle of attack. When the pilot would keep controlling in the low velocity region, the increased spring coefficient for negative deflections (pull) in Frame 3 indicates them that pulling should be executed with caution. Crossing the stick shaker activation threshold gives a clear cue that a stall is imminent, shown by Frame 4. Finally, if the pilot still did not react, the state in Frame 5 is at the upper angle of attack limit where the stiffness is maximum, the stick shaker is active, and the neutral point shift is most observable, all cues which inform the pilot of the proximity of the flight envelope to the state.

2.4.3. Possible undesired actions

The previous sections discussed the intended use of the proposed haptic feedback system. We now look at possible undesired actions, which are discussed using the concepts of misuse, disuse and abuse as proposed by Parasuraman. [80]

Misuse is the use of the automation for an unintended goal, typically due to overreliance on the system. In the case of the haptic feedback system, overrelience can result in a lack of scanning the instruments: the pilot might expect the haptic feedback to signal an approaching limit and focus on other tasks besides the primary flight duty. As the feedback system is reliant on sensor measurements, if these sensors fail, the haptic system might not trigger whereas a scan of the instruments might show the erroneous measurement. In an evaluation experiment, presence of overreliance on the haptic feedback system might give different results in the scenarios discussed above: in case of windshear the pilot is actively maneuvering the aircraft closer to the limits and more likely to be aware of closing limit, in case of icing the limits move to the current state and in case of overreliance this event can surprise the pilots.

Disuse is the deliberate *not* using of the automation available, commonly caused by a distrust in the system due to a significant false alarm rate. Looking at the haptic feedback system while assuming that it functions as intended (no false positives), pilots might still consider the haptic feedback as false when it would be perceived as out of tune with respect to the magnitude of the flight envelope protection zones. For instance, the haptic feedback might signal a limit as close, whereas the pilot experiences it not as such. In that case, the haptic feedback might be considered as distracting when controlling the airplane, in a worse case, the pilots can feel that they are fighting the haptic feedback system. An evaluation of the system therefore has to check that the workload of the pilots does not increase, and that pilot actions and haptic feedback are in line.

Abuse is the automation of functions by designers without due regard for the consequences for human performance. In the haptic feedback design, part of the design parameters (for example the magnitude of the discrete cue) are heuristically tuned using one test pilot. Due to this heuristic tuning, the haptic feedback might be experienced by some pilots as intrusive. As such, an evaluation has to investigate whether the current setup does not increase workload, and allows the pilot to keep performing the nominal mission.

This provides a set of criteria for the haptic display of flight envelope limits: the haptic feedback system should not increase workload and should not hinder the primary pilot tasks. Furthermore, it should be investigated that overrelience is not present, and that pilots are not fighting with the system. Additionally, performance and safety metrics of pilots flying both windshear and icing scenario's should improve. In conclusion, we expect that the new haptic display presented in this chapter increases the knowledge of the pilot on the edges of the flight envelope and helps identifying abnormal situations. This hypothesis is tested with an experimental evaluation: the subject of the next chapter.

2.5. CONCLUSION

This chapter describes the design of a haptic feedback system, i.e., using force feedback through the control device, to provide intuitive information on the state of the aircraft relative to the Flight Envelope Protection. The system (i) informs the pilot about an approaching limit using a discrete cue, (ii) indicates a non-desired control direction using the spring coefficient, (iii) warns the pilot of a dangerously low velocity using a stick shaker, (iv) shows a desired control input during an over-speed event by moving the control device, and (v) indicates the required control input at low velocities when a stick neutral position is not sufficient by moving the control device.

3

EVALUATION OF A HAPTIC FEEDBACK SYSTEM FOR FLIGHT ENVELOPE PROTECTION

Flight envelope protection systems in modern fly-by-wire aircraft support pilots in staying within the safe flight envelope, however the actions of these systems may not always be clear to pilots. To support situation awareness, the aircraft proximity to the limits of its flight envelope can be communicated using haptic feedback, through providing forces on the control device. The haptic feedback design was discussed in detail in the previous chapter; this chapter reports on its experimental evaluation. Professional airline pilots were invited to fly an airliner model similar to an Airbus A320 in a research flight control law, with the haptic feedback system turned on and off. Results show that the haptic feedback does not lead to significant changes in performance or safety metrics. It also does not interfere with normal pilot control actions. Pilots did express a clear preference for the haptic system, however. Recommendations for future work include more research in haptic feedback tuning, the addition of visual support to complement the haptic cues, and the redesign of the scenarios.

This chapter is published as:

Title	Evaluation of a Haptic Feedback System for Flight Envelope Protection
Conference	AIAA Guidance, Navigation, and Control Conference, San Diego (CA), 2019
Authors	D. Van Baelen, J. Ellerbroek, M. M. van Paassen, and M. Mulder

3.1. INTRODUCTION

MODERN flight decks provide an abundance of information to pilots, using mainly the visual and aural communication channels. Examples of visual displays are the PFD for the most important aircraft states, and the ND for a planar, top-down overview of the aircraft trajectory and flight plan. Aural signals are used to provide urgent messages, such as to warn pilots for excessive aircraft velocities, and to provide altitude read-outs and throttle back-commands on landing. [59]

But apart from these senses, pilots are able to perceive information in several other ways. This chapter will elaborate on the use of the pilots' haptic sense, by providing haptic feedback through the control device. As shown in Fig. 3.1, within the field of haptic research two main categories are identified: (i) touch, stimuli to the skin, and (ii) kinaesthesis, stimuli to the receptors in the muscles, joints and tendons. [57, 60] The design used in this chapter uses both touch and kinaesthesis, hence the term haptic feedback is used.



Figure 3.1: Components of haptics [57]

In most 20th century aircraft, the control manipulator 'feel' provided information on for instance aerodynamic forces, buffeting when close to a stall, control surface limits through hard stops of the controls, and other control-related phenomena. With the introduction of fly-by-wire, however, the forces on the control surfaces and the control devices were decoupled, eliminating the potentially useful haptic information channel. [61]

A reason that haptic feedback on the aircraft limits was not integrated after the introduction of fly-by-wire systems in the 1980s and 1990s, was the rather bulky device required to implement the haptic forces. These old devices had issues regarding their size, weight, power and stability requirements resulting in certification difficulties. Currentday devices have become much smaller and lighter, while still able to provide reliable haptic feedback. [62] This offers the possibility to re-consider this type of feedback in fly-by-wire control systems. [63] Together with the advances in control devices automation on flight decks is rising, leading to a more supervisory role for the pilot, instead of direct manual control. Despite these advances, pilots are still often required to take over manual control of the aircraft in landing, takeoff, or during emergency scenarios. An example of the latter could be a computer or sensor malfunction, which was the case for Air France flight 447. Here the crew, startled by unexpected high-altitude dynamics, lost situation awareness, despite the information available from the visual and aural displays. Unaware of the loss of the usual flight envelope protections due to the malfunction, pilots stalled the aircraft. The accident report indicates that the stall warning, which was only aural, should be complemented. [7] The control manipulator, the Airbus A330 side stick, did neither provide the pilots with direct feedback on their control actions, nor the aerodynamic stall buffets. The passive stick did not help them in properly identifying the situation as a stall. As this tragic example shows, when manual control is required the lack of haptic information through the control device might contribute to a reduced situation awareness.

Combining the ever-increasing sophistication of automation on the flight deck, and the current generation of small and powerful control devices provides designers a new opportunity, namely to increase pilot awareness through haptic feedback. Some aircraft already include 'augmented forces' on the control device, which can be provided on *both* control devices (on a two-pilot flight deck) linked to the surfaces, or fly-by-wire control systems. An example of this is the "Q-feel force", which changes the stiffness of the controls with changing dynamic pressure/velocity in Boeing type aircraft. [64] Another example is a stick shaker or pusher, which warns pilots of moving closer to extreme aircraft states. [4] The control device can also be loaded with two passive springs to create a change in experienced stiffness when pilots exert large control deflections irrespective of the aircraft state, such as in Airbus aircraft. Active control can be used to have an increased (artificial) spring force when rolling the aircraft beyond the safe roll limit, irrespective of the control surfaces, as used on a Boeing 777. [55]

Although examples of haptic feedback implementations exist, there is limited research published in open literature to prove the benefits of such a system. Within aeronautics, Schmidt-Skipiol and Hecker used either a passive spring or an active counterforce to communicate the distance to the flight envelope limits, with the latter yielding better performance. [50] A second example is the work by Stepanyan et al. that showed the limit on the available control space both visually and haptically. [52] For the haptics, the input neutral point and maximum deflection were manipulated, to be used by pilots to operate close to the aircraft limits. A soft-stop, i.e., a local step in the force required for a certain deflection, can be used to indicate the engine limitations in the collective of a helicopter, which was shown in simulations to reduce pilot workload ([44, 65]), and is implemented in an experimental helicopter of the German Aerospace Center. [66] Tactile feedback through the use of tactors on a vest enabled improved spatial awareness and reduced spatial disorientation. [29] These examples all use haptic feedback to inform the pilot about the flight envelope limits. Note that research in supplying the pilot with such information is not limited to haptic only, new visual displays are investigated as well and show positive results. [51] Aside from information on the flight envelope limits, the haptic channel can be used to supply guidance support, of which a haptic flight director showed potential to increase pilot control performance while reducing workload. [67]

Other fields do have more open domain research interest, for example in automotive. Here, haptics were successfully applied on the gas pedal, to show the proximity of a lead car, with better performance while reducing driver input magnitudes. [81] Haptics for the steering wheel are currently being developed, to support curve negotiation. Implementations as either a 'warning system' (to reduce the driver reaction time) or 'guidance system' (to keep the car on the road) exist. [38] In tele-operation, the control of an unmanned aerial vehicle was supported by haptically showing the proximity to objects in its surroundings, leading to safer flight and increased operator awareness. [32]

The aim of the current thesis is to investigate the use of haptic feedback to give pilots more information on the aircraft FEP limits during manual control, within the modern fly-by-wire flight deck. In other words, the haptic feedback aims to provide feedback to the pilot on the proximity of the aircraft state to the limits adhered by the FEP system. The full design and rationale is described in detail in the previous chapter. This chapter presents the evaluation of this new design using two operational scenarios flown by commercial airline pilots. This design only involves longitudinal haptic feedback, lateral cues can be added in the future.

Section 3.2 will first provide a brief description of the control laws and elaborate the cues which can be perceived using the haptic feedback. Section 3.3 presents the experimental setup, designed to test two operational scenarios: windshear and icing. Results are shown in Section 3.4 and discussed in Section 3.5. Recommendations are given in Section 3.6, followed by conclusions in Section 3.7.

3.2. HAPTIC DISPLAY

The haptic feedback design rationale, full implementations details, and the Airbus A320 control structure on which it is based, are extensively discussed in the previous chapter. This section provides the essential information on the control laws, Subsection 3.2.1, and haptic feedback design, Subsection 3.2.2, to understand the working principles and provided information during the experiment, discussed in Section 3.3.

3.2.1. AIRBUS A320 CONTROL STRUCTURE

Modern day Airbus aircraft, such as the A320 and the A330, employ a fly-by-wire system. This means that there is no mechanical connection between the control surfaces and the control device, yet the latter acts as an interface between the pilot and the FCCs. The FCCs augment the pilot control input with a control law, to move the control surfaces with hydraulic actuators. The FCCs also include a Flight Envelope Protection (FEP) system, which can limit pilot inputs, such that no flight envelope limits are violated.

Longitudinal control in a Fly-By-Wire (FBW) Airbus, when all sensors are functioning, is the normal law. It uses C^* -control, a combination of both pitch rate (q) and load factor (n). [10–13] On top of this control law, a hard envelope limit is defined which protects pilots from exceeding pre-defined limits on angle of attack (α), load factor (n), and maximum velocity (V_{MO}). This protection is shown in Fig. 3.2, where the nominal flight envelope is the extreme limit which can not be exceeded, and the safe flight envelope is the point where protections start acting. The envelope is defined by the maximum (n_{max}) and minimum (n_{min}) load factor, their protection limits ($n_{max_{prot}}$ and $n_{min_{prot}}$, respectively), the maximum operation velocity (V_{MO} , and protection $V_{MO_{prot}}$), and minimum velocity ($V_{\alpha_{max}}$, and protection $V_{\alpha_{prot}}$). $V_{\alpha_{max}}$, $V_{MO_{prot}}$



Figure 3.2: Flight envelope, load factor (n) versus velocity (V)

When multiple FCCs fail, or when a sensor failure occurs, control is reverted to a degraded control law. In this research, besides the normal law, the Airbus alternate law without reduced protections is considered. In this law the same protections apply as before, only the angle of attack protection is lost, and therefore, while in alternate law, the aircraft can be stalled.

These control laws and envelope protections were implemented in a proprietary model by the German Aerospace Center (DLR) which is made available for this research, see Ref. [54, 72]. Although this model in not in the public domain, the principles behind the haptic feedback can be extrapolated to any model which provides flight envelope information. Note that the model does not include the automatic 'pitch-up' command during high speed conditions, which raises the aircraft nose when its velocity is too high.

3.2.2. HAPTIC FEEDBACK DESIGN

For this research, haptic feedback is provided using two methods: (i) by *changing the haptic profile* of the control device, and (ii) by *adding deterministic forcing function signals* to the control device. The haptic profile is defined here as the amount of force required (*F*) to achieve a certain deflection of the control device (δ). A typical nominal haptic profile is illustrated in Fig. 3.3a: a breakout center (δ_{br}) with an increased spring coefficient (k_{br}), followed by symmetric spring behaviour (k_{nom}) until maximum deflection (δ_{max}). Such a haptic profile provides the pilot with information on the input magnitude: larger inputs require larger forces. By changing this haptic profile, certain deflections require pilots to exert a different force input, which informs them about (non-)desired inputs.

The goal of the haptic feedback is to communicate the FEP limits in an intuitive way. When presenting the evaluation scenarios in Section 3.3, we will discuss how we expect pilots to use the following five cues:

1. When the aircraft leaves the safe flight envelope (anywhere outside the red dashed line in Fig. 3.2), a discrete, unit pulse *forcing function is added* to the stick resulting in a perceived 'tick on the stick'. This is illustrated on the haptic profile with the inset graph in Fig. 3.3b.



(c) Increased spring coefficient for positive deflections

(d) Positive shift in position of the neutral point

Figure 3.3: Control device profiles

- 2. For aircraft velocities close to the lower velocity limit V_{prot} , a stick shaker (i.e., a sinusoidal forcing function) activates, to communicate the increasing risk of stalling the aircraft.
- 3. To communicate the relative distance to the limit, the *spring coefficient increases* when moving from the safe flight envelope to the actual limit (the black line on Fig. 3.2). Maximum spring coefficients are determined by the tuning parameters K_{α} , K_n and $K_{V_{MO}}$ for angle of attack, load factor and maximum velocity, respectively, as discussed in our design chapter, Chapter 2. The increased spring coefficient results in a situation where pilots must apply a larger force to move the stick in a particular direction, illustrated for positive (push) deflections in Fig. 3.3c.
- 4. When the aircraft has a critically low velocity, and bringing the stick back to its neutral position is not sufficient to return to the safe flight envelope, the required stick deflection is communicated to the pilot by *a change in neutral point* of the stick, as illustrated in Fig. 3.3d.
- 5. During an overspeed situation, the automatic 'pitch up' command is communicated to the pilot by a change in neutral point of the stick, similar to Fig. 3.3d, but now using a negative neutral point position. This command is discussed in the previous chapter and is defined by one tuning parameter ($\tau_{overspeed}$).

All implementation details are provided in the design chapter, Chapter 2. For example, there we discussed that both the discrete and periodical forcing functions are
empirically tuned to be noticed by pilots, without interfering with actual control inputs. We also discussed two scenarios with intended use, and potential disuse/abuse/misuse of the haptic feedback. The goal of this chapter is to show the results of these scenarios when flown by professional pilots in a human-in-the-loop experiment evaluation, discussed next.

3.3. METHOD

An experiment was performed to test the haptic feedback system. Professional airline pilots were instructed to fly standard approaches, in which emergency situations were introduced.

3.3.1. INDEPENDENT VARIABLES

The experiment had three independent variables. First, the haptic feedback was either present (HF), or not (NH). Second, two levels of protection in the control laws were used: a control law resembling the the default Airbus normal law (NL) where all protections are present, and one resembling the Airbus degraded alternate law (AL) where a stall is possible (see Subsection 3.2.1 for more details). Third, two scenarios were used: windshear (WS) and icing (IC), as will be discussed in detail below.

A factorial 2*x*2*x*2 design resulted in eight experiment conditions. However, experimenting with the first participants revealed that the flight dynamics simulation model did not work well during the icing scenario in normal law, and it was decided to abandon these two conditions. The experiment was continued with the remaining six conditions for all remaining participants. Each of the six conditions was flown by each participant, in both a training and measurement phase. The order of the conditions for each participant was determined using a latin-square.

3.3.2. PARTICIPANTS AND INSTRUCTIONS

Participants (all males) were commercial airline pilots with a current Airbus license (either A330 or A320); their experience is shown in Table 3.1. Pilots did the experiment alone, no other crew members were either present or simulated. Instructions were to fly as they would in the actual aircraft, and to voice their thoughts as much as possible. Additionally, the pilots were told that each run would stop at 50ft of altitude, irrespective of any other events or their performance. Because all participants (and experimenter) were native Dutch speakers, the questionnaires introduced below were all provided in Dutch; they have been translated to English for this publication.

3.3.3. APPARATUS

The experiment was performed in the Research Facility for SImulation, MOtion and NAvigation (SIMONA) at Delft University of Technology. It is a full-motion capable, near 180° outside field-of-view, generic flight deck, used in the first officer position of which an inside-picture is shown in Fig. 3.4. Since the pilot was seated in the first officer position, the display to the front-left was the ND showing a top-down view, Fig. 3.5a. The display right in front of the pilot was the PFD showing the critical flight states, Fig. 3.5b for normal and Fig. 3.5c for alternate law.

Participant	Age	Flight hours	Airbus flight hours
1	45	13,500	500
2	47	15,000	4,500
3	57	17,000	2,000
4	41	10,000	6,000
5	56	14,000	7,500
6	30	4,800	4,600
7	27	5,200	5,000
8	49	14,000	7,000
9	34	5,000	4,000
10	29	3,000	1,900
11	26	2,250	2,100
Mean	40.1	9,431.8	4,100
Std. Dev.	11.5	5,466.6	2,258.8

Table 3.1: Experiment participants



Figure 3.4: Inside view of the SIMONA flight deck (picture by Thierry Schut)



(b) PFD for normal law



(c) PFD for alternate law

Figure 3.5: Flight deck display setup

A control-loaded, two degrees-of-freedom electrical Moog FCS Ecol-8000, with a haptic profile resembling a conventional Airbus side stick was the main control manipulator. The nominal control device settings, and the haptic feedback settings for this experiment, all defined in the previous chapter, are given in Table 3.2. Forcing functions used were a discrete unit pulse function (duration 0.1s, magnitude 10N) when exiting the safe flight envelope, and a stick shaker for low velocities as a sinusoid (frequency 20Hz, magnitude 5N). Furthermore, a Boeing 777 center console with throttles and flap-lever provided engine and high-lift device inputs, and a Boeing 737 Mode Control Panel (Airbus terminology: Flight Control Unit) was the interface with the heading/velocity/altitude references on the displays.

Parameter	Value
Kα	3 -
K_n	2 -
$K_{V_{MO}}$	2 -
k_{nom}	1 N/ deg
δ_{max}	18 deg
k_{br}	50 N/ deg
δ_{br}	0.05 deg
$\tau_{overspeed}$	15 s

Table 3.2: Control device and haptic settings in the experiment

3.3.4. EXPERIMENT SCENARIOS

The haptic feedback system is designed to communicate the 'proximity of the flight envelope limits' to the pilot. It was expected that it supports pilots by providing both a control aid when maneuvering close to the limits, as well a warning aid when moving (unknowingly) to the limits. To investigate these two potential use cases, two experiment scenarios were designed: (i) a situation where the pilot is forced to maneuver close to the aircraft limits: a windshear event; and (ii) a situation where the flight envelope shrinks, and the limits slowly move to the normal maneuvering range of pilots: icing.

For both scenarios, a training and evaluation design was extensively discussed and prepared with the help of an experienced A330 captain (and instructor). They are elaborated in the following; we also discuss how we expected pilots to use the system.

WINDSHEAR

A wind shear is a meteorological phenomenon where wind velocities are locally rapidly changing, and can be caused by a large cylinder of air suddenly "droping" towards the earth. During such a wind shear event, the pilot has to control the aircraft as close to the stall limit as possible, to prevent further height loss. This can be a dangerous situation especially when it occurs when flying close to the ground, such as on final approach. When a windshear is detected by the aircraft sensors, this is announced using visual and aural warnings.

Following these warnings, pilots must apply the windshear recovery procedure as stipulated by the aircraft manufacturer. For Airbus aircraft, the procedure is shown in Section B.5 and includes the following items ([13]):

- 1. Do not change configuration (flaps/slats/gear),
- 2. Set thrust levers at go-around position,
- 3. Set pitch initially at 17.5°,
- 4. Increase pitch to eliminate descent rate,
- 5. Closely monitor flight path and speed, and
- 6. Recover smoothly when clear of the windshear.

As can be seen in this checklist, the pilot might need to increase the aircraft pitch angle to extreme values if a severe windshear occurs and thereby the aircraft can possibly reach the limits of the flight envelope. This latter property makes this scenario of great interest for our evaluation. Following the FAA training manual, the windshear in this experiment was modeled by both a head-on and top-down component as shown in Fig. 3.6, where *x* represents the along track distance traveled from the trigger point. [76]



Figure 3.6: Windshear component distribution

Training Phase To make sure the pilots became sufficiently familiar with the windshear procedure, it was first trained using an approach to Nice (France). A full approach was flown since it improved the immersive feeling of the simulation. The flight path, with the approach chart shown on Section B.3, was an agile GNSS approach starting at an altitude of 2,500ft, velocity of 190kts, heading 310, and 2NM South-East of point SOVEX. From there, the pilot had to fly along the navigation points in the right order, with speeds and altitudes as indicated on the chart:

- 1. NANAX: minimal 2,000ft, maximum 150kts
- 2. MN410: at 1,500ft, maximum 145kts
- 3. MAP22: descent to 1,000ft after this point, maximum 145kts

Between waypoints MN410 and MAP22, and when the velocity and altitude were 'stable' as judged by the experimenter, the windshear was initiated and the pilot had to execute the recovery procedure. The missed approach procedure consisted of turning South, climbing to 2,000ft, 185kts and reporting to Air Traffic Control (ATC).

Measurement Phase In this phase a tear-drop approach was flown into Montpellier, to runway 12L. A tear-drop was chosen to increase pilot workload, as it requires a rather accurate timing of the maneuvers. The approach started at 3,000ft altitude, 180kts, heading 300, 1.6NM South-East of the airfield, and the pilot had to perform a procedure as indicated on the approach chart in Section B.4:

- 1. Fly overhead the airfield while descending to 2, 500ft,
- 2. Follow radial 332 outbound of the FJR VOR,
- 3. Slow down to 150kts,
- 4. At 7.0NM radial distance from FJR, make a left turn to heading 121,
- 5. Perform the turn using a 'rate one turn' (2 minutes per 360°), using the rule-of-thumb: "roll angle equals speed divided by 10, plus seven" ([82]), and
- 6. On completion of the turn, maintain heading/velocity/altitude and await ATC instructions.

When the turn to heading 121 was completed, and when the velocity and altitude were 'stable' as judged by the experimenter, the windshear is initiated. The missed approach consisted of maintaining runway heading 121, climbing to 2,500ft, and informing ATC.

Intended Use In our design chapter, we proposed that the haptic feedback system would help in informing the pilot on extreme inputs which result in an aircraft state close to the limits during the initial stage of the windshear recovery procedure. Next, when the pilot has to increase pitch to climb, the initial discrete cue can help the pilot being aware of the envelope limit. The stick shaker can be used to operate the aircraft near the limits by 'riding the stick shaker': adjusting the input such that the stick shaker remains on the verge of activation. Additionally, at the end of the recovery, with high throttle settings, the pilot can follow the pitch-up cue to understand the high-speed protection and can feel that the distance to the limit is decreasing with the increased local stiffness. In summary, the haptic feedback system was expected to help in alerting the pilot that the aircraft is approaching the flight envelope by means of the discrete cue, and to communicate the distance left to the ultimate flight envelope boundaries using the increasing spring stiffness.

ICING

When flying through cold, humid air, there is a risk that ice starts forming on the leading edges of the aircraft and degrades its aerodynamic performance. [79] This can result in reduced lift from the wings, increasing the minimum velocity and reducing pilot control authority. If this is not anticipated by the pilot, icing can result in a loss of control.

The aircraft model included a simulation of the effect of icing on the aerodynamic parameters, which was used to test this scenario. For both the training and measurement

runs, the simulation was initialized slightly off the instrument approach path to a runway. The instructions for the pilots in this scenario were to follow the localizer/glide slope, to fly as slow as possible due to heavy traffic, until an altitude of 50ft where the simulation automatically stops. It was the pilots' responsibility to decide what the slowest velocity was, and the information they had was the minimum velocity indicated on the PFD and felt through the haptics (if enabled).

Training Phase During training, an Instrument Landing System (ILS) approach to Rotterdam (EHRD) for runway 24 was performed. The approach started at 2,500ft, 140kts, 15.4NM radial distance from touch down point, slightly left of the localizer, and below the glide slope. From this point on, the pilot had to intercept the localizer and glide slope as indicated on Section B.1. Approximately 60s after the start of the run, the aircraft was stabilized on the localizer/glide slope and icing was gradually ramped in. The stable approach after 60s was checked visually by the experimenter, and was indeed the case for all runs and for all pilots.

Measurement Phase Here, a similar instrument approach was flown into Amsterdam (EHAM) for runway 36C. The approach started at 3,000ft, 150kts, 10.3NM radial distance from touch down point, slightly right of the localizer, and below the glide slope. The pilot was again asked to intercept the localizer and glide slope as indicated on Section B.2 and the icing initiation point is approximately 100s after the start of the run.

Wind variation During the approach, there were stable and variable wind components. The stable wind was different for each of the runs the pilots performed: values can be found in Table 3.3. As the runway in use for Rotterdam is 24 (heading West South-West, actual heading 237), and for Amsterdam is 36C (heading North, actual heading 003), all wind instances gave a headwind with a slight crosswind. The four wind states were distributed using a (four-by-four) Latin square, which was repeated every four pilots.

	Stable	Stable	Variable
Wind realization	heading [°]	intensity [kts]	heading [°]
Training set 1	272	7	262
Training set 2	259	11	249
Training set 3	214	11	224
Training set 4	201	7	211
Measurement set 1	321	13	331
Measurement set 2	332	11	342
Measurement set 3	027	11	017
Measurement set 4	038	13	028

Table 3.3: Wind components during the icing scenario

Variable horizontal wind (w_{var}) was added to the experiment to mask the change in aircraft aerodynamic properties, and to increase pilot workload. It was modeled with a sinusoidal shape, Equation 3.1, where *I* is the amplitude, $T_{duration}$ the duration of one period, and $T_{trigger}$ the time between the end of one shape and the start of the next. When $t > (T_{trigger} + T_{duration})$, time was reset to zero, and the parameters were re-initialized. It was converted to North-East components using the direction specified in Table 3.3.

$$w_{var} = \begin{cases} 0 & \text{if } t < T_{trigger} \\ \frac{I}{2.0} \cdot \left(1.0 - \cos\left(2 \cdot \pi \cdot \frac{(t - T_{trigger})}{T_{duration}}\right) \right) & \text{if } T_{trigger} \le t \le \left(T_{trigger} + T_{duration}\right) & (3.1) \\ \text{reset properties and time} & \text{else} \end{cases}$$

Parameters were re-initialized to prevent recognition of the variable wind, and taken from a random distribution. *I* was taken from a Weibull distribution with $\lambda = 4.9$ and k = 2.3, $T_{duration}$ from a Weibull with parameters $\lambda = 9.0$ and k = 5.5, and $T_{trigger}$ from a normal distribution with $\mu = 6.5$ and $\sigma = 1.0$. To enable reproducibility of the results, and equal variable wind over the participants, one realization of the stochastic process is made for each of the eight cases listed in Table 3.3.

Intended Use The haptic feedback system was expected to help pilots in identifying the performance degradation, especially in a manual control situation, by the progressing severity of the haptic cues: an initial tick when the protection value is exceeded, a changing neutral point since the load factor corresponding to the protection can be below one, an increasing stiffness with the flight envelope boundary approaching, and a stick shaker when crossing the respective activation velocity.

3.3.5. Secondary task: ATC requests

As a secondary task, and used as an objective measure for pilot workload, pilots were instructed to listen and, if required, respond to ATC requests throughout the simulations. Each of the ATC requests is characterized by three variables: a callsign, a command, and a trigger time. All callsigns used had the company name of which the pilot is an employee. At the start of the experiment day, each pilot was asked what his preferred flight number is, and if no preference, flight number '107' was used. Two other flight numbers were used: 685 and 713, which resulted in three possible combinations of callsigns.

The commands issued by ATC were requests on state information. These could be: 'report altitude', 'report speed', or 'report heading', which are variables easily retrievable from the PFD. Of course, these requests had to be answered only for the correct callsign.

To prevent recognition and anticipation of the requests, random realizations were made of the above; one realization was coupled to one condition. Using a uniform distribution, a callsign and a command was selected. Next the trigger time was determined by a normal distribution (μ = 20s and σ = 2.5s) indicating the time after the previous command. At the trigger time, the command was presented to the pilots. The eight realizations made of this random process resulted in a slightly different number of ATC calls made during each condition, as shown in Table 3.4.

To exclude any effects of the experimenter, messages were produced with a text-tospeech generator: the 'festival' library by The University of Edinburgh. [83] This latter was setup to provide the requests with a synthetic American-English, female voice.

3.3.6. EXPERIMENT ORDER

None of the participating pilots had any experience with haptic feedback on the side stick (since current Airbus aircraft do not provide this), nor with the particular behaviour of this aircraft model. Therefore, even before training a familiarization phase was required to

	IC AL		WS			
			AL		NL	
Condition	HF	NH	HF	NH	HF	NH
Total ATC calls during training	32	30	30	31	31	31
ATC calls to participant during training	12	9	6	13	14	12
Total ATC calls during measurement	31	30	29	31	31	30
ATC calls to participant during measurement	8	12	13	16	17	8

Table 3.4: Number of ATC calls per run

provide pilots with sufficient time to get used to the haptics, aircraft model, and simulator. We discuss the familiarization, and the setup of the training and measurement runs in the following.

FAMILIARIZATION

After a briefing on the simulator safety procedures, controls and displays, the aircraft model was introduced to the pilots by requiring them to fly a traffic pattern to a final approach at Schiphol (EHAM), see Fig. 3.7. Pilots were asked to follow the instructions as indicated, unless company policy deviates. This to focus on the model familiarization, not procedures. The traffic pattern was flown three times in visual approach weather conditions: twice in normal law without wind, and once in alternate law with variable wind similar to the winds used in the icing scenario.



Figure 3.7: Traffic pattern flown to runway 36L at Schiphol (Schiphol layout from AIP [84])

After the model familiarization, the pilots were offered the opportunity to *feel* the design rationale behind the haptic feedback design. This was done by presenting both the flight envelope (computer implementation of Fig. 3.2), haptic profile (computer implementation of Fig. 3.3), and the PFD (Fig. 3.5b) to the pilot. In this static setup, no aircraft model was used, yet the flight envelope state was changed directly (hence changing the velocity and load factor) and all cues of the haptic design were elaborated in detail.

Following the static demonstration, the pilots *flew* the haptic feedback design using the full simulation model. Pilots were asked to fly a set of maneuvers (stall, overspeed, high load factor, climbing stall) three times. This was flown once in normal law without haptics, such that pilots could see how the aircraft reacts. Next the manoeuvres were repeated in normal law, with haptic feedback enabled, such that pilots could feel the feedback. Finally, it was flown once more in alternate law, with haptics enabled. After this last set of maneuvers, the pilot was asked to pitch to 5° nose down, followed by closing his eyes and pitching up as far as possible while not crossing the envelope limits. The rationale behind the latter was to see whether the pilot could indeed feel the limits of the aircraft. If (s)he stalled the aircraft, the haptic cues were not understood, if the pilot was able to keep pitching up while not exceeding the limits, the pilot understands how the haptic support communicates the limits. Each pilot was able to pitch-up the aircraft while not stalling the aircraft, indicating that at this stage in the evaluation the haptic feedback was indeed understood. After this familiarization phase, a lunch break was provided.

TRAINING AND MEASUREMENTS RUNS

The training and measurement runs used the six conditions as described above. These were flown in a randomized fashion, using a latin-square distribution. The pilot had to respond to ATC requests during the run, as explained before. After each run, pilots were asked to give a Rating Scale Mental Effort (RSME) rating ([85]), and fill-in the post-run situation awareness questionnaire (both are discussed below). Once the task was completed, the appropriate performance score was communicated to the pilot (discussed with the dependent measures). After all runs were finished, pilots were asked to complete a post-experiment questionnaire, which contained a number of questions on how they experienced the haptic feedback system (discussed below).

As mentioned to the pilots in the briefing, each experiment run was stopped at 50ft above the ground. This limitation was added because the aircraft model did not include a wheels-on-ground reaction model.

3.3.7. DEPENDENT MEASURES

The dependent measures are split into objective and subjective measures.

OBJECTIVE MEASURES

The objective measures were directly retrieved from the experiment data and focused on performance, safety and workload.

First, pilot performance depends on the scenarios flown. During windshear recovery, performance was defined by the altitude lost, the difference between the altitude at which the windshear was initiated, and the lowest altitude flown during the recovery. During

icing, performance was defined as the root-mean-square of the position deviation from the ILS glide slope/localizer, starting at the initiation of the icing event until the end of the simulation.

Second, safety was defined as the remaining margins to the limits during the windshear recovery or icing event, using three sub-metrics based on angle of attack: (i) to determine *how close* the pilot operated to the limits, the 5% closest distance to the limit $(\alpha - \alpha_{max})$ was taken and the average is calculated, (ii) in order to see *how long* the pilot operated near the limits, the amount of time spent in the protected zone was calculated (hence time where $\alpha > \alpha_{prot}$), and (iii) to differentiate between short, close encounters with the limit, and sustained flight in the neighbourhood of the limit, a combination of distance and time was calculated by the integral of the angle of attack in the protected zone (hence $\int (\alpha - \alpha_{prot}) dt$). A visual summary of the safety margins is shown in Fig. 3.8.



Figure 3.8: Safety metrics indicated on time trace

Related research often reveals a trade-off relation between performance and safety, referred to as risk homeostasis: the perceived level of risk is kept constant with increasing support. [86] An example is the increase in driving speed when being supplied with haptic feedback for lane keeping in an automotive study. [87]. In this experiment, risk homeostasis was considered to be present when performance would increase, while safety margins would degrade.

The third metric used to evaluate the system was an objective measure for workload: the performance with respect to the secondary task introduced above (responding to ATC requests). When workload due to the primary flying task was high, responses to the ATC requests were expected to be less accurate, and vice versa. Note that this follows the aviation order of priority: '*aviate, navigate, communicate*'.

In addition to these system evaluation metrics, a design evaluation metric was introduced to validate one of the design assumptions. As mentioned in the design section, it still remains to be seen whether the forcing functions used for the haptic feedback, cueing the pilot about approaching the FEP, would interfere with nominal pilot inputs. Therefore, the ratio of force exerted by the haptic feedback system (F_{haptic}) was compared with the total force exerted on the control device (human F_{human} and haptic feedback F_{haptic}) as shown by Equation 3.2. Here the integration is approximated by the summation of the discrete simulator states.

$$r_{\text{haptic}} = \frac{\int |F_{haptic}| dt}{\int (|F_{human}| + |F_{haptic}|) dt}$$

$$\approx \frac{\sum |F_{haptic}| \cdot dt}{\sum (|F_{human}| + |F_{haptic}|) \cdot dt} = \frac{\sum |F_{haptic}|}{\sum (|F_{human}| + |F_{haptic}|)}$$
(3.2)

SUBJECTIVE MEASURES

The subjective metrics included workload, situation awareness, and pilot experience. Workload was measured after each run using the RSME which presented the participants with a scale ranging from zero to 150 to indicate how mentally demanding the last run was. To aid the participants with choosing a score, several verbal indications are added (for example 'a little effort', or 'great effort') at validated locations along the scale. [85]

Situation awareness was measured by presenting the participants with two questions after each run: 'Did you have the feeling you were in control of the situation?' and 'Did you have the feeling you missed critical information?'. Both questions had a linear ratio scale (0–100) ranging from left 'Never' (0), to right 'Always' (100).

Finally, to ask participants for their experience with the haptic feedback system, and to streamline the debriefing, every pilot was asked to fill in a questionnaire after the experiment. It proposed thirteen statements, see Table A.1, using five point Likert-scales where the extreme and middle points are labeled.

3.3.8. HYPOTHESES

The haptic feedback system aims to improve pilot situation awareness with respect to the flight envelope limits and protections. To determine whether this goal is achieved, the following hypotheses were tested on the 'measurement phase'-data described above:

- 1. With haptics enabled, risk homeostasis is present, that is, performance metrics will improve while safety margins reduce.
- 2. With haptics enabled, the pilot awareness of the aircraft critical flight states improves, as indicated by better scores for the situation awareness questions asked after each run, and in the debriefing questionnaire.
- 3. With haptics enabled, pilot workload, in terms of both the secondary task performance and RSME ratings, decreases.
- 4. The haptic feedback is equally effective when maneuvering the aircraft towards the edges of the flight envelope, the windshear scenario, as compared to a shrinking flight envelope, the icing scenario.
- 5. The haptic feedback is equally effective in both normal and alternate law in terms of performance and safety.
- 6. The haptic feedback does not interfere with nominal pilot behavior (executing normal tasks such as navigation and communication).

In the following, the results are shown and thereafter discussed to see whether we can find evidence in the experiment data to reject any of the proposed hypotheses.

3.4. RESULTS

In the experiment, some irregularities with the aircraft model came up which will be discussed first. In the next subsection, it will be illustrated how a pilot used the haptic support system, using typical time histories. A full description of all experimental results is split in objective measures, presented in Subsection 3.4.3, subjective measures in Subsection 3.4.4, and pilot answers to the questionnaire, Subsection 3.4.5.

3.4.1. SIMULATION OBSERVATIONS

Throughout the experiment campaign, four unexpected events are noted. The first two were simulation-related, the latter two were pilot-related.

First, a problem with the control law simulation occurred which resulted in oscillating aircraft responses (sometimes enlarged by pilot control inputs) in the runs indicated in Table 3.5 with an 'x'. The effects were mostly a worse performance; when the oscillation occurred close to the flight envelope limits, also the safety margins reduced. To avoid any misinterpretation of our data, these runs were all excluded from further analysis.

This first category of events is a consequence of using the flight dynamics (and control laws) model in regions close to the flight envelope limits, for which these models were originally not developed. This can be regarded as an inevitable problem for our research, as good and validated models of aircraft operating close to their limits are scarce and difficult to obtain.

	WS			IC		
	NL		AL		AL	
Participant	NH	HF	NH	HF	NH	HF
1	0	х	Х	0	i	i
2	0	х	0	0	0	0
3	0	0	0	0	0	х
4	0	0	0	0	0	0
5	0	0	0	0	0	0
6	0	0	х	0^1	0	0
7	0	0	0	0	0	0
8	0	0	0	0	с	0
9	0	х	х	х	0	0
10	0	0	0	0	0	0
11	0	х	0	0^1	0	0
comparisons	7	7	8	}	8	}

Table 3.5: Unexpected events: 'x' indicates control law oscillation, 'i' indicates too low icing, 'c' indicates crash

¹Angle of attack increased above protection value and pilots were informed by the haptic feedback.

Second, during the measurements for the first participant, it was found that the icing severity was too small. Hence, the icing level was increased for all participants that followed. All data on the icing conditions for Participant 1 were discarded, indicated in Table 3.5 with an 'i'.

Third, after dropping below the glide slope in the icing condition without haptic feedback, Participant 8 tried to climb again but stalled the aircraft, resulting in a crash. This is indicated in Table 3.5 with a 'c'.

Fourth, in the windshear scenario with haptic feedback, Participants 6 and 11 slowed the aircraft down *without* increasing the flap setting. The resulted in an angle of attack above the protection value and therefore the haptic feedback informed them as such. On the haptic cue, the pilots noted the low velocity and adjusted the flap setting. These events did not have any influence on the performance nor safety margin metrics. They are a good example of other unintended use of the haptic feedback system.

In the evaluation of the results, discussed below, our main focus will be to investigate the effects of activating the haptic feedback *within* a scenario, (windshear or icing) using one control law (normal or alternate law). That is, it will be investigated for each of the three conditions what the effects will be of activating the haptic feedback, and we will compare the 'no haptics' versus 'haptics' runs pair-wise. Since these comparisons could be affected by any of the first three simulation events mentioned above, all comparisons with 'issues' will be discarded. For example, Participant 2 encountered a control law oscillation with the windshear scenario, using normal law when provided with haptic feedback, as such data for this participant on windshear in normal law are not presented below. The other data points for this participant are shown in all figures and analysis.

When excluding all runs with simulation or pilot issues, a total of 7, 8 and 8 comparisons can be made for the windshear+NL, windshear+AL and icing+AL conditions, respectively. Note that any of the comparisons is statistically questionable, as the experiment runs are inevitably dependent, and any data do not come from independent samples. One should further note that, in making a comparison with just 8 samples, applying conservative statistics means that at least 7 of these comparisons should have the same sign for the comparison to be significant at the 0.05 level. Hence, with the limited number of pilots participating, most comparisons will be statistically insignificant.

3.4.2. TIME TRACE

Time histories for Participant 3 are shown, in the windshear scenario, flown in alternate law with the haptics enabled (WSALHF). Time traces for angle of attack, load factor, side stick deflection, force exerted on the control device, pitch angle and velocity can be found in Fig. 3.9, which all start at the time of the windshear warning which is initiated approximately 250s into this flight. Three time frames are distinguished, defined in the text below. Combining the control device deflection and force, Fig. 3.10 shows the changes in the haptic profile, for the same three time frames.

At the start of the windshear recovery procedure, the pilot observed the aircraft states, and pulled the stick (Fig. 3.9e) to increase the pitch angle to approximately 17.5°, Fig. 3.9c. The pilot then maintained that pitch angle using slight variations in input. At Frame 1, Fig. 3.10a, the pilot encountered the protections and was informed as such with a discrete haptic tick (indicated with the star (*) symbol). Nevertheless, to stop the aircraft from descending, the pilot maintained the pitch angle. When the stick shaker was activated at Frame 2 (indicated with the solid dots), the pilot reduced his inputs and pushed slightly on the side stick until the stick shaker de-activated. Following the de-activation, the pilot intended to move the side stick backwards again, yet encountered once again the stick



(e) Side stick deflection

Figure 3.9: Time traces for Participant 3 in condition WSALHF, thrust was set to full after 6.4s

shaker (solid dots), Frame 3, and had to push the side stick, as shown in Fig. 3.10c. Note that the way the pilot used the haptic feedback system in this example, effectively 'riding the stick shaker, is as expected.



Figure 3.10: Haptic profile during windshear recovery of Participant 3 in condition WSALHF. The current state is indicated with a cross, and a preview of the control device of 1.5s, frame numbers correspond to Fig. 3.9

3.4.3. OBJECTIVE MEASURES

The objective measures are determined based on the time traces of the experiment. Here the performance for both scenarios is shown, followed by the metrics for safety margins, secondary task scores, and an analysis of the forces applied by the haptic feedback system. In all figures, individual pilot data are shown with and without the haptic feedback enabled, for both the training and measurement phases. Although the metrics for the measurement phase is leading in the evaluation of the hypotheses, the training data is additionally shown due to the lack of data points in the measurement phase, and to show initial pilot behavior.

To facilitate quick reading, we counted and show the number of times the dependent measure increased more than 5% (\uparrow), decreased more than 5% (\downarrow), or remained equal (=), for each individual pilot. These indications should be considered with care, however, as the 5% threshold is arbitrary, and as mentioned above statistically the comparisons cannot be regarded as independent.

The data for each condition is summarized in a boxplot, where the median is indicated with a horizontal line, outliers with a cross. It will be the medians that are central in our analysis, as these indicate what the 'average pilot' would do with the haptics enabled.

PERFORMANCE

Performance in the windshear scenarios, Fig. 3.11a, and icing scenarios, Fig. 3.11b, do not indicate any visible trend in the median of the measurement phase data. Some larger differences are present between participants: each participant has a certain performance-level indifferent from the control law, or haptics state used. But overall one can safely say that performance during the measurement phase is not affected by the haptics system.

The trends which are present in the measurement phase, are more prominently present in the training phase: during the windshear scenario, enabling the haptic for normal law seems to worsen performance with normal law, whereas in alternate law the performance doesn't change. The icing scenario clearly shows an improvement when haptic feedback is enabled during training.

Aside from the general trend, some participants do stand out: for Participant 3, the haptics-enabled run in normal law during the measurement phase was one of the final runs, and a sign of fatigue showed: his initial reaction after the windshear warning was a reverted control input; using his captain (left-hand) seat-routine he closed the throttles and pitched down, resulting in a low performance. Participant 7 shows consistently the



Figure 3.11: Performance scores, lower values indicate improved performance

lowest performance for windshear. During the debriefing, he indicated that he prioritized the windshear over the stall warning and haptic cues by maintaining a pitch-up command despite the visual/aural/haptic low velocity warnings.

On the icing plot in Fig. 3.11b, Participant 11 is the only participant who has a noticeable decrease of performance when enabling the haptic feedback, no confounding factors (such as the control law or fatigue) have been identified. Nevertheless, the pilot was not paying much attention to the velocity/altitude causing a larger deviation from the glide path and resulted in considerable effort to rejoin the glide slope.

SAFETY

Objective metrics for safety margins are defined in relation to pilot behavior at or near the limits. The first metric, presented in Fig. 3.12, looks at the 5% highest values for angle of attack near the windshear or during icing, and takes the average of these data as the evaluation criterion. More negative numbers means more distance to the limit, hence a larger safety margin.



Figure 3.12: Safety metric: Closest distance to the limit, positive values indicate (aural) stall warning

Similar to the performance in windshear, and now also for icing, no trend is present in the measurement data, yet considerable differences between participants are observable. The windshear scenarios clearly show the control strategy of Participant 7: by ignoring the high angle of attack warnings, worse safety margins are obtained. The training data of windshear shows a similar lack of trends, for icing, a slight improvement is present following performance.

Fig. 3.13 shows the time outside the safe flight envelope. For the windshear data, during both training and measurement, the median time spend outside is marginally lower when enabling haptic feedback. In the case of icing, although the median is slightly decreasing, more participants have an increase of this metric.







Figure 3.14: Safety metric: integral of α in protected zone

The integral of angle of attack in the protected zone for both scenarios is shown in Fig. 3.14 and shows no clear trends, yet does show the performance level per participant. Similar observation for windshear can be made as with the previous safety margin metric: the control strategy by Participant 7 results in the worst obtained metrics. During the training for icing, Participant 8 maintained a velocity around the haptic feedback triggering point. This results in small, but sustained, excursions of the safe flight envelope.

SECONDARY TASK

During each run, the answers of the pilots to the requests of ATC were recorded. The ratio of incorrect answers is shown in Fig. 3.15. It shows that the pilots in each case were triggered by the ATC command, and gave correct answers. No statistical significant results are obtained. A trend might be visible, yet keep in mind that the number of calls addressed to the participant (see Table 3.4), was not the same for all conditions.



Figure 3.15: Secondary task: ratio of incorrect responses to ATC

DESIGN EVALUATION

For each of the conditions where haptic feedback is provided, Fig. 3.16 shows the amount of force exerted on the side stick by the feedback system, relative to the total force exerted. It includes all flights shown in Table 3.5, except for those where a problem with the control law occurred. The figure indicates that, in general, the haptic feedback system exerts a small portion of the total force: the medians are between 0.3% and 2.6%.



Figure 3.16: Ratio of force exerted by haptic feedback system

3

3.4.4. SUBJECTIVE MEASURES

After each run, participants were asked to provide a number of ratings, these include a measure of how well they were in control, Fig. 3.17, a measure on whether they had all the critical information they needed, Fig. 3.18, and a rating for their mental workload, Fig. 3.19. As for the previous results, no statistically significant trends show.



Figure 3.17: Subjective situation awareness: is the pilot in control?

When asked whether the pilot was in control, enabling the haptics in the windshear scenarios does not seem to influence most pilots: scores remain equal and the number of pilots indicating higher/lower is almost the same. In contrast, switching on the haptics in the icing scenario improved the feeling of being in control during training, yet lowers the perceived level of being in control during the measurement phase: five participants indicated feeling less in control. This hints that haptics can be used to provide pilots with information during training, yet is less helping when accustomed to the task and dynamics.



Figure 3.18: Subjective situation awareness: is the pilot missing information?

The information missing according to the pilot, Fig. 3.18, has again no clear trends. During windshear, in Fig. 3.18a, two participants stand out: Participant 6 during training, and Participant 2 during measurement. Participant 6, in normal law during training, had the no-haptics run, followed by the haptic-enabled run, this sharp improvement of this metric indicates that the haptic feedback added information for this pilot. Participant 2 during the measurement phase experienced the run with haptic had less information missing, leading also him to conclude that the haptics provided the information required as it was not available on the display anymore. In the icing condition, Participant 4 consistently missed information on the icing level.

Multiple pilots, such as Participant 3 and 5 during the icing scenario, indicated that the triggering data of the haptics (mostly α_{prot} and $V_{MO_{prot}}$ for this case) were not clear, especially in alternate law as that triggering data is not available on the PFD. Therefore, they were not able to properly understand the haptic cues and indicated an increased amount of information missing: the reason for the haptic cues.



Figure 3.19: Workload: RSME

Mental effort required to operate the aircraft for each condition is shown in Fig. 3.19. These figures show once more that each participant has a personal base line, and little variation is present between condition. Even more, including the between-participant variability on the box-plots indicates that very little spread is present around the median scores.

3.4.5. POST-EXPERIMENT QUESTIONNAIRE

After the final run, the participants were asked to complete a questionnaire which queried them about their experience with the system. It involved thirteen questions shown in Table A.1, and the answers are shown in Fig. 3.21. Note that in this figure, the horizontal axes are defined in such a way that the right-hand side of the subfigures indicate ratings which are favorable towards the haptics system enabled, and the left-hand side less favorable ratings.

The questions contain one entry for each pilot (hence eleven in total) for the windshear scenarios, and ten answers for the icing scenario since one pilot did not want to complete the questions for the icing scenario as he did not recognize it as such an event. Statistical analysis between both scenarios is performed using a Wilcoxon Signed Rank test with continuity correction.



74

(a) What is your general feeling on the haptic feedback as an information cue about the flight envelope? -2) Negative 0) Neutral 2) Positive



(d) The haptic interface distracts me. -2) Agree 0) Neutral 2) Disagree

No

Yes (g) Do you expect any adverse impact

on outcomes when using this technol-

10

5

ogy?



(b) The haptic interface affected my workload.





(e) The haptic interface changes the likelihood of human error. -2) Increased 0) Neutral 2) Decreased



(h) Using the haptic system, my knowledge on the edges of the aircraft performance changed. -2) Less 0) Neutral 2) More



(c) If a critical event occurs, the haptic interface helps to mitigate the consequences.





(f) I was fighting the haptic interface. -2) Always 0) Sometimes 2) Never



(i) The haptic interface changed my behavior.

-2) Agree 0) Neutral 2) Disagree

Figure 3.20: Results of the post-experiment questionnaire (1); blue indicate answers for windshear, red for icing

The first question, Fig. 3.20a, asked pilots what their general idea on the haptic system is. For the windshear scenario, the ratings are statistically significant more towards the positive side, whereas for the icing scenario the results are more centered around neutral (V = 28, p = 0.021).

Another set of questions was asked to investigate the expected consequence of using the system according to the pilots. With a statistical significant difference from the icing answers, the pilots answered in case of windshear that with haptics their workload was less (Fig. 3.20b, V = 0, p = 0.011), the system has the ability to help mitigate consequences of critical events (Fig. 3.20c, V = 15, p = 0.057), and did not distract them (Fig. 3.20d, V = 1.5, p = 0.040). For the icing scenario, the haptic system gave a neutral to increased workload, a more neutral opinion for the system to mitigate consequences of critical events, and a larger distraction. In both scenarios, pilots expect the possibility of human error to decrease (Fig. 3.20e), in general were not fighting the haptics (Fig. 3.20f), and expected no adverse impact (Fig. 3.20g). Overall, a neutral to positive change on their knowledge on the edges of the flight envelope was observed (Fig. 3.20h), which should be kept in mind when considering the goal of the design. Pilots did not provide a uniform answer to whether they changed their behavior (Fig. 3.20i).

In order to evaluate whether the simulation was adequately representing an actual Airbus aircraft, four questions on realism were asked see Fig. 3.21. The displays and weather implementation were positively received, whereas the aircraft dynamics did lack realism. The realism of the controls, in Fig. 3.21b, is more important as the experiment considers the design of the controls itself. Most pilots perceived the (nominal) feeling of the controls to be acceptable.



(a) How would you grade the A320 dynamics? -2) Unrealistic 0) Acceptable 2) Perfect









(b) How would you grade the controls? -2) Unrealistic 0) Acceptable 2) Perfect



(d) How would you grade the weather? -2) Unrealistic 0) Acceptable 2) Perfect

3.5. DISCUSSION

The discussion is split in the objective, subjective and post-run criteria. We conclude with the overall haptic feedback system evaluation.

3.5.1. OBJECTIVE MEASURES

Statistical analysis of results for the objective performance metrics indicated there was no significant difference beteen the scenarios, nor for the different control laws. Nevertheless, in the icing scenario the haptic feedback system did lead to an improvement in terms of performance during training. Note that this scenario is only evaluated for alternate law, in which case the display shows minimal velocity, and no velocity protections. With the addition of the haptic feedback angle of attack information is added, as the haptics use the protection angle of attack to trigger the discrete cue. This improvement of performance could be caused by this inclusion of angle of attack information, and not directly in the way of communicating it. Hence, one can only conclude that showing angle of attack information can be beneficial for the performance.

From the safety margins in the icing scenario, no significant differences appear, yet the medians of the integral of the angle of attack and the time outside the safe flight envelope tend to indicate that more margin is kept. As for the performance metric, this tendency is most likely caused by the inclusion of the angle of attack data, not present on the PFD. An extra argument that the information presented by the haptics is required, is the occurrence of a crash in the haptics-disabled run of Participant 8.

Safety margins for windshear showed no statistically significant results, nevertheless, enabling haptics in alternate law tends decrease the time spent above the protection value of angle of attack and decrease the integral of angle of attack outside the safe flight envelope. The latter indicates that pilots, on average and not statistically confirmed, moved less into the protected zone when given the (haptic) protection information.

Combining both scenarios in alternate law show no indication for risk homeostasis: performance is not improving and slightly more margin is kept to the limits. This indicates that the current data does not support our first hypothesis

In normal law, pilots can see the velocity corresponding to the angle of attack protection on the display, but do not move closer to the limits. This can indicate that the perceived level of risk is not changed by adding the haptic feedback, and pilots are still relying on the information on the displays. A possible reason for not using the haptic feedback information could be a lack of experience with the haptic system while having ample experience with the visual system, or a dislike for the haptic feedback information channel. The first can be mitigated by re-evaluating the training, probably a longer training with the haptic system is needed. From the questionnaire, discussed below, we inferred that pilots do not disapprove on the system. Nevertheless, to avoid dislike of the system, the limits used by the haptics can be perhaps more clearly communicated to the pilot on the display. An example for this is the overspeed protection (V_{MO} – 20kts) used by the haptics, yet not visible on the PFD.

Results for both the performance and safety metrics show a large differences in metrics between participants: the metric per participant is in most cases of similar value, irrespective of control law used or status of the haptic feedback system. This shows that the largest variation is the participant, not the intervention used. Hence, the hypotheses stating that the haptic feedback is equally effective in normal and alternate law, and also equally effective when moving towards the flight envelope limit as compared to the limit approaching by itself, are comparing two conditions between which no significant differences are found. Therefore these hypotheses cannot be rejected.

The ratio of correct answers to ATC did not show any difference over the conditions. Therefore this secondary task cannot confirm that the workload of the pilot changes. It did show that the addition of haptic feedback still allowed pilots to aviate, navigate the approaches, and communicate with ATC. Hence, these data supports the hypothesis that the haptic feedback does not interfere with nominal behavior.

Finally, the design evaluation parameter showed that on average 0.3 % to 2.6% of the force on the side stick is delivered by the haptic feedback system. The metric which is used here, takes into account all absolute forces by the haptic feedback system imposed on the side stick. For example, a sustained stick shaker could inflate this metric since it is a high frequency oscillation yet it provides no actual input. As this example shows, we can consider this metric to be a worst possible case. Therefore the metric indicates that the haptic feedback system is indeed 'cueing' the pilot on the limits, and not imposing any input, following our intended design goal.

3.5.2. SUBJECTIVE MEASURES

The subjective measure for situation awareness involved two questions: "Is the pilot in control?" and "Is the pilot missing information?". As stated, pilots indicated that they feel more in control during the training phase, yet slightly less in control in the icing event. Following the debriefings with the pilots, this is traced back to the mis-match of the visual display, showing the aircraft minimum velocity, and the haptic system giving information on the angle of attack. The latter was largely influenced by the icing, caused the haptic cues to be triggered at much higher velocities. During training this helped the pilots to be aware that something was going on. Nevertheless, when more accustomed to the model and task, those cues did not match the information on the visual display, and pilots found this situation to be unclear, resulting in a reduced feeling of being in control during the measurement phase.

Looking at whether pilots were missing information, our participants did not seem to experience that they had more information available in the scenarios with haptics. The objective results before did indicate that in the windshear scenario, when using alternate law, they moved slightly less into the protected flight envelope, and they performed better with the icing scenario, which could only be caused by the added information from the haptics. Nevertheless, this was not experienced as such by the pilots. To make the haptic feedback more clear, it should either be redesigned, or the information should be more clearly presented, for example, using multiple modalities. [58]

Ideally, a system improving situation awareness should give the pilot the feeling of always being in control and never missing information, as is the hypothesis for this system. Nevertheless, the answers to the questions after each run do not confirm that the haptic display increases the situation awareness of the pilots.

Looking at the results for the workload, a grouping per subject is present, again indicating that the intervention used is not the major variation. This information is currently not sufficient to reject the hypothesis on decreasing workload.

3.5.3. POST-RUN QUESTIONNAIRE

The questionnaire asking the pilots for their experience with the system, showed a clear difference between both scenarios (already rejecting Hypothesis 4). For the windshear scenario, pilots clearly experienced the haptic feedback system as positive, whereas more neutral for the icing scenario, which corroborates the metrics discussed before and the answers in the following.

With respect to the windshear scenario, although the objective and subjective metrics before did not show this, the pilots after the experiment did experience a decrease in workload, supporting Hypothesis 3. Although most pilots do not indicate a change, some indicate that their knowledge on the flight envelope improved. Additionally, pilots indicate that the haptics help to solve problems and it deceases the human error possibility. This implies that the pilots expect the haptic feedback system to transfer some knowledge about the situation, partly supporting Hypothesis 2. Nevertheless, as stated in the results, this unexpected answer of the pilots on the knowledge on the boundaries should be further investigated, more elaborate interviews with the pilots might indicate points of improvement.

Pilots did not provide a clear answer whether the haptic feedback changed their behavior. This might be caused by the way the question is posed, the intention of the question was to query 'a change in nominal behavior', yet the pilots might have interpreted this to 'a change in their behavior to incorporate the haptics in their loop'. As such, depending on how the question is interpreted, different answers are possible. Nevertheless, as the haptics were not experienced as distracting, nor did pilots report to 'fight' them, the pilot's answers support the hypothesis that it does not change nominal behavior.

No clear preference for the haptic feedback system was present for icing. Pilots indicated that the haptics slightly increased their workload, distracted some of them, and made them 'fight' the system sometimes. This can be partially explained by (i) the icing scenario might have been slightly unclear: one pilot did not recognized the situation as icing, and (ii) a biased comparison of available information (as no angle of attack indication present in no haptics case). The latter is confirmed by the debriefing: pilots remarked that the haptic cues were sometimes unclear. When a haptic cue was provided, it was not always supported by an item on the display. An example of this is the haptic cue for angle of attack, while the visual display does not show this, leading to a clear recommendation for follow-up designs.

In the debriefing, some pilots remarked that from all haptic cues provided, they especially liked the discrete cue provided when crossing one of the protection values. Although it was intended to communicate a direction to move away from this protection, pilots did note that this direction was not always clear. It therefore served as a clear trigger that something is going on and they have to be vigilant. The discussion also showed that pilots experienced the other cues as useful, yet less salient.

From all the post-run questions, one which clearly stands out in favor of this new system is the answer to the expected negative outcomes: this clearly shows that pilots do not expect an immediate negative result. Combining this with their general feeling about the system, and the debriefing, shows that almost all the pilots are in favor of haptic feedback for flight envelope protection. Some did even make a remark along the lines: 'Why is this system not implemented yet?'.

3.5.4. SIMULATION EVALUATION

The four questions concerning the level of reality of the simulation indicated that pilots did perceive the displays and weather to be adequate. For the latter, mostly the occurrence of icing was problematic. Concerning the model, there was a significantly lower drag present compared to the actual aircraft, and, noted in the objective results, some oscillations occurred due to the flight control law in the proprietary model. This model behavior might have resulted in a more conservative control strategy of the pilots, making the difference due to the haptic feedback less prominent.

Considering this research investigates haptic feedback through the control device, one of the most important realism questions was the realism of the nominal feeling which was experienced as acceptable, slightly leaning to an off-nominal feeling. The major point of improvement is the default stiffness of the control device. In an actual aircraft, the default control stiffness is higher, making our simulation more responsive. Nevertheless, as the pilot received time to get acquainted to the controls, this probably had only minor influence on the results.

3.5.5. OVERALL SYSTEM EVALUATION

As some of the hypotheses could be evaluated by both objective and subjective measures, this subsection summarizes the results for all of them. Hypothesis 1 proposed that risk homeostatis is present and is fully based on the objective results. Those results tend to show that risk homeostasis is not present as pilots do spent slightly less time in the protected zone for the haptics-enabled windshear condition with alternate law and during the icing condition. Therefore this first hypothesis is not supported by the data.

The pilot awareness of the aircraft critical flight states does not seem to increase nor decrease based on the subjective questions asked after each run, nevertheless, in the debriefing questionnaire the pilot's answers indicate that they have an increased perceived situation awareness. Therefore, the current data partly supports Hypothesis 2 for an increased situation awareness.

Workload was measured objectively with the ratio of correct answers to ATC and subjectively with a RSME evaluation, giving no difference when enabling haptics. Looking at the post-run questionnaire, the icing scenario showed an increase in workload due to the unclear cues, yet in the windshear scenario it was deemed that workload decreased. Hence, Hypothesis 3 is not supported with the current design because of the lack of evidence.

To support Hypothesis 4, no differences should be found between the windshear and the icing scenario. Mainly the post-run questionnaire indicates that the current haptic feedback design is not equally effective for both scenarios. This might be due to both the remaining simulation problems, as well as unclear reason of the haptic feedback in the icing scenario as stated by the pilots.

It was shown for both scenarios that the visual displays in alternate law did not show the pilots the protection zones, whereas the haptic feedback system did. As such a skewed comparison was performed. In alternate law for windshear, the pilots tend to operate less in the protected zone with haptic feedback enabled. Nevertheless this trend was not visible for normal law. As such the effect for both control laws was different and Hypothesis 5 is not supported. As pilots were able to aviate, navigate and communicate throughout all runs, the haptic feedback system does not interfere with nominal behavior and Hypothesis 6 is supported. The results shown above give the indication that the haptic feedback system in the current design does not lead to the expected result in terms of our hypotheses. Nevertheless, all pilots indicated that they see the potential benefit of such a system.

Chapter 2 discussed possible use, misuse, disuse and abuse following the principles proposed by Parasuraman and Riley. [80] Misuse is the use of automation for an unintended goal, which in this experiment would be shown by overreliance on the system. As explained in the simulator events, two participants started a windshear run with decelerating without the proper flap setting. This brought them close to the limit, and were informed as such by the haptic feedback. This could be considered as a lack of scanning the instruments when haptic feedback is available, yet it happened only once for these two participants. This leads us to conclude that there is no consistent overreliance on the haptic feedback system.

Disuse is deliberately not using the haptic feedback, commonly caused by a distrust in the system. Some participants commented, as discussed before, that the haptic cues were not always clear in the icing scenario. Although the safety margins increased for the icing scenario indicating intended use of the haptic feedback, this scenario might be at risk of disuse as the limits are not visible on the visual display.

Finally, abuse by designers would result in a system that provides pure guidance due to the large cues, instead of the intended goal of informing the pilot of the limits. The design evaluation metric showed that only a small portion of the input was imposed by the haptic feedback system, leaving the control actions solely to the pilot. This metric therefore proves that our intended design goal, without abuse, was achieved.

3.6. RECOMMENDATIONS

From this haptic feedback system evaluation, several points of improvement can be listed. One of the most straightforward recommendations for a next iteration of a haptic feedback system is the inclusion of more participants. Due to simulator and pilot availability, this evaluation included the results of eleven participants, where several runs had to be excluded due to a control law issue. Future evaluations can always benefit from more participants to capture smaller differences and reduce the dependency of the results on one individual.

Another recommendation to tackle the lack of differences is to re-consider the scenarios. Currently, the windshear scenario asked pilots to operate the aircraft as close to the angle of attack limit as possible, the icing scenario tried to operate at the same boundary. Providing pilots with a scenario in which they have *more freedom in their solution*, could result in more prominent use of the haptic cues. Nevertheless, this makes analysis more complex as one single evaluation metric might not be available for a wide range of solutions. On the other hand, *restricting the operating space* might make the initial conditions more uniform over all participants, enabling simpler evaluation. Doing so might reduce the reality and make the simulation less immersive, reducing the validity of the scenarios. The simulation environment itself can be improved by resolving the oscillatory response, and by improving the default stiffness to better resemble the existing A320 side stick. Furthermore, inclusion of angle of attack state and limit data proved to be useful in the icing scenario, therefore it is recommended to provide this information to pilots. Note that the limit data requires information on the current flight envelope, yet this can be obtained using update algorithms. [88]

During the debriefing, some pilots indicated a preference for the discrete cue that indicates crossing a protection value. A next design iteration might therefore explore the use of discrete signal as it proved to be useful in the automotive field. [36] Additionally, to improve acceptance of the haptic feedback itself, the haptic display can be augmented by a visual display as multi-modal information proved to be more clear. [58]

Finally, as the current system does not interfere with nominal pilot behavior, it should be investigated whether haptic feedback for flight envelope protection can be combined with haptic feedback for tracking tasks, such as during an instrument landing. [41]

3.7. CONCLUSION

To increase pilot situation awareness about flight envelope limits we investigated haptic feedback – force feedback through the control device. Our haptic interface communicates the proximity to the flight envelope limits using several cues, and was evaluated in the SIMONA research simulator by eleven professional Airbus pilots. The results showed no significant changes in performance and safety margins for any of the conditions or control laws used. Nevertheless, all metrics showed that the haptic feedback did not hinder them to perform their tasks. Finally, the debriefing indicates that some pilots experienced an increased situation awareness, and most pilots see a potential benefit of implementing the haptic feedback system on a modern fly-by-wire flight deck.

II

DESIGN IMPROVEMENTS

4

SUPPLEMENTING HAPTIC FEEDBACK THROUGH THE VISUAL DISPLAY OF FLIGHT ENVELOPE BOUNDARIES

This chapter describes the design and evaluation of a visual display in supplementing haptic feedback on the side stick as a way to communicate flight envelope boundaries to pilots. The design adds indications for the limits in airspeed, load factor, angle of attack and angle of bank to a standard Airbus Primary Flight Display (PFD). The indications not only show the limits of the flight envelope, but also indicate magnitude and direction of the haptic cues. Fifteen professional Airbus pilots and one Airbus simulator instructor participated in an experiment in the SIMONA Research Simulator at Delft University of Technology. Several approaches in three different scenarios were flown in alternate law with the old and new PFD, while haptic feedback was always enabled. Although the time spent outside the flight envelope is slightly reduced, performance with the new display was not improved significantly. Subjective results indicate a preference, however, for the new display and an increased understanding of the haptic feedback. Further research is recommended to focus on improving the design by removing unused indications and setting up an experiment with a bank scenario that allows the use of operational bank limits rather than artificially reduced limits.

This chapter is published as:

Title	Supplementing Haptic Feedback Through the Visual Display of Flight Envelope Boundaries
Conference	AIAA Guidance, Navigation, and Control Conference, Orlando (FL), 2020
Authors	G. de Rooij, D. Van Baelen, C. Borst, M. M. van Paassen, and M. Mulder

4.1. INTRODUCTION

▲ VIATION is one of the safest forms of transport, yet there are still improvements to $oldsymbol{\Lambda}$ be made. In the last decade, loss of control in-flight (LOC-I) has been the primary cause of fatal commercial jet airplane accidents. [89] There are multiple slightly different definitions of LOC-I, but a common factor is that it involves flying outside the flight envelope with the potential of making it impossible for the pilot to control the aircraft. [90] Modern fly-by-wire (FBW) aircraft can be protected from such flight envelope excursions, but when automation fails, the pilots find themselves in a stressful situation and do not always know what to do to keep the aircraft within the envelope. An example of such an occurrence is Air Asia flight 8501 in 2014. [91] Due to a fault in the Rudder Travel Limiter Unit (RTLU) of the Airbus A320 and subsequent actions by the crew, the aircraft switched from normal to alternate control law, losing most of its protections and disconnecting the autopilot. The RTLU fault made the aircraft bank to 54°. Startled by this, the crew responded incorrectly, banking the aircraft to even more extreme angles and eventually pulling the aircraft into an unrecoverable prolonged stall. All 162 people on board perished when the aircraft crashed into the Java Sea. This and other incidents like Air France 447 ([7]) show that once protections are lost, pilots lack clear cues on their position with respect to the flight envelope and how to return to the envelope if they exceed it.

Chapter 2 presented that haptic feedback can be used as a way to communicate information to human operators and has shown that haptic cues might close this information gap and decrease LOC-I incidents. Nevertheless, Chapter 3 found that pilots were unsure as to what triggered the haptic feedback and what corrective action to take. It was recommended that a visual representation of the haptic cuing and flight envelope is developed to help pilots understand what the haptic feedback is telling them. In combination with haptic feedback, this may assist pilots in recognizing the edges of the flight envelope and acting accordingly. Research on an unmanned aerial vehicle collision avoidance system indeed suggests an increase in user acceptance when adding visualizations to a haptic system. [92]

Several research projects have looked at the design of displays that can show (more) flight envelope information. The primary flight display (PFD) seems to be the preferred location to integrate such information, although some projects designed stand-alone displays. A common factor in most existing solutions is the separation of output and input space, showing either the limits of the envelope ([54, 93–95]) or the limits in control inputs that would otherwise bring the aircraft outside that envelope. [96] No previous research is known on an aircraft display specifically integrating the limits on the input and output space together with information on associated haptic feedback.

This study builds on the foundations of the aforementioned research by investigating the design of a display that integrates the input and output space, while also showing the force and direction cues of the haptic feedback. The chapter starts with background information on flight envelopes and haptic feedback in Section 4.2. Section 4.3 presents the display design and explains the rationale behind it. The display was tested in a humanin-the-loop simulator experiment involving 16 professional pilots as explained in Section 4.4, to asses the added value of said display. Section 4.5 lists the results of the experiment, which are then discussed in Section 4.6 together with some recommendations for further research. Finally, Section 4.7 concludes this chapter.

4.2. BACKGROUND

A basic understanding of the flight envelope is required to grasp the working of the haptic feedback and consequential display design choices. This section provides a short introduction to these concepts, together with a couple of implementation details that are specific to this research.

4.2.1. FLIGHT ENVELOPE

The longitudinal performance limits of an aircraft are often captured in a flight envelope that relates velocity (*V*) to load factor (*n*). A common flight envelope shape is depicted by the solid line in Fig. 4.1. The upper velocity limit is dictated by the maximum velocity V_{MO} that can be attained by the aircraft respecting aerodynamic and vibration limits. Structural limits, indicated by horizontal lines, put a minimum n_{min} and maximum n_{max} on the load factor, independent of airspeed. At low speeds, a quadratic relation limits the minimum velocity $V_{\alpha_{max}}$. Flying below $V_{\alpha_{max}}$ at a too high load factor will stall the aircraft. With extended flaps, both $V_{\alpha_{max}}$ and V_{MO} decrease, leading to a much smaller flight envelope (Fig. 4.1b). Airbus in addition moves the lower and upper load factor limits to 0 and 2g respectively when the slats and flaps are extended ([13]), but the model from this experiment keeps the load factor limits at -1 and 2.5g in order to match the haptic feedback.



Figure 4.1: Typical flight envelopes with velocity (*V*) versus load factor (*n*); [97] augmented with load factor data for 10,066 A320 flights; [98] the actual envelopes depend on the aircraft's configuration and loading

Safety margins are added to the flight envelope to create a so-called safe flight envelope, indicated by the red dashed line in Fig. 4.1. The associated *protection* margins are chosen such that pilots have sufficient time to steer the aircraft away from the boundaries after being alerted of leaving the safe flight envelope. The load factor margins are 0.5 g, lower speed margins vary along the envelope, and high-speed margin is fixed at 20 kts below V_{MO} . Another margin can be distinguished near the lower velocity indicated by the dashed green line, and showing critically low velocity close to a stall.

The envelopes in Fig. 4.1 are overlaid with maximum and minimum load factors encountered in 10,066 Airbus A320 flights. [98] Note that the envelopes shown here are for illustration purposes only and do not precisely match the actual envelope corresponding to those flights. In flaps up, aircraft in general stay well away from the boundaries, nevertheless some flights do get close to the limits of the safe flight envelope. On the contrary, with the significantly smaller flight envelope corresponding to a flaps 3 configuration, the majority seems to operate outside the safe flight envelope. This can be explained by the fact that the fixed 20kts overspeed margin of the safe flight envelope is not used in real-life operations.

4.2.2. HAPTIC FEEDBACK

The working principles of the haptic feedback system are best explained using the haptic profile, shown in Fig. 4.2a, which gives the stick deflection δ and the force required F. Break-out zone δ_{br} and associated spring coefficient k_{br} give the pilot a haptic feeling of the neutral point δ_{np} . Outside this break-out zone the spring coefficients are related to the negative (k^-) or positive (k^+) deflection of the stick. The full description can be found in Chapter 2. While only longitudinal haptic feedback was considered, lateral feedback based on the same principles, has since been implemented and both were used in the present research.



(c) Increased spring coefficient for negative deflections

(d) Positive shift in neutral point

Figure 4.2: Control device profiles

The system can be summarized with five haptic cues to communicate the flight envelope to the pilot. First, when the aircraft leaves the safe flight envelope, a discrete force cue warns the pilot, depicted by the in-set graph in Fig. 4.2b. Second, continuing to steer the aircraft out of the safe flight envelope, results in a progressively increased stiffness as shown by the asymmetric profile on Fig. 4.2c. Third, when zero stick input is insufficient to return to the safe flight envelope for low velocities, the neutral point moves as shown on Fig. 4.2d. Fourth, a stick shaker activates when crossing the critical lower velocity indicated on the flight envelope in Fig. 4.1 by the dashed green line. Fifth, the neutral point of the stick during an overspeed situation shifts backwards to indicate the automatic pitch up command.
4.3. DISPLAY DESIGN

The haptic feedback system from Subsection 4.2.2 was tested with professional pilots in the Chapter 3, resulting in a recommendation to investigate the addition of a display to visualize the haptic cues. Combining haptic feedback with a visual display could fulfill the important principle of multiple resources when presenting information. [99] To address the shortcomings of existing displays, such as the lack of integration of input and output space, a new display was designed. It should show the pilots which envelope limit is triggering the haptics, where the aircraft is with respect to the (safe) flight envelope and what forces are acting on the stick. This section first elaborates on the principle behind a design that fulfills all of these requirements and then explains the look and feel of the various new display elements.

4.3.1. DESIGN PRINCIPLE

In order to support the haptic system, the indications on the display have to match with the forces felt through the side stick in both magnitude and direction. From the cues discussed in Subsection 4.2.2, the discrete cue and changing stiffness can be visualized by an ordinary spring (upper part of Fig. 4.3) that is positioned next to the side stick. When the aircraft approaches the edge of the safe flight envelope as discussed in Subsection 4.2.1 the spring moves towards the stick. Upon leaving the safe flight envelope the free end of the spring – visualized by the left-most vertical line – barely touches the stick. At this point the haptic feedback gives a discrete tick on the stick to grab the pilot's attention. When the aircraft gets further into the protection zone, the spring is progressively compressed, its width increases and so does the force exerted by the spring. This force acts in the direction opposite to the movement of the stick, making it harder for the pilot to maintain a stick input in that direction. If the compression is relaxed, the spring lengthens again while its thickness and force decrease. Like any spring, the force is only felt when the spring starts getting compressed. The maximum compression is reached when the two vertical lines touch each other. Beyond that maximum the spring coefficient does not change any further.



Figure 4.3: Spring (top) and piston (bottom) symbols with increasing levels of compression

To ease implementation in the display, improve clarity and reduce clutter, the spring can instead be visualized in the form of a piston cylinder whose thickness is similar to the width of the spring (lower part of Fig. 4.3). Apart from visualizing the 'feel' from the haptics in both magnitude and direction, these indications also show the pilot in which direction he should provide control inputs to alleviate the required force and return the aircraft to the safe flight envelope. All of this is known to help pilots understand and consequently appreciate haptic feedback better. [92]

The other cues from the haptic system are not explicitly visualized. The stick shaker is a trigger to bring the pilot's attention to low speed rather than an actual limit, so no extra indication is added. The neutral position shift is neither explicitly visualized, as it comes in combination with an increased stick stiffness and thus another visual indication.

The piston analogy is used throughout the enhanced display. The symbols and colors are kept uniform over the various indications to adhere to Wicken's design principle of consistency. [99] In line with industry recommended color coding, yellow is used to indicate the protection limit, beyond which the aircraft is outside the safe but still within the nominal flight envelope. [100] The actual flight envelope limits are indicated in red.

In order to help pilots quickly determine what flight parameter is driving the haptic feedback on their control inputs, the various axes (bank, load factor, angle of attack and airspeed) are displayed separately. Where possible the new indications are placed on parts of the display that are already showing the related parameter(s) according to the proximity compatibility principle. [99] Fig. 4.4 shows the PFD with all of the flight envelope indications in place. The various elements are discussed in greater detail below.



Figure 4.4: Wireframe view of the Airbus A320 PFD with additional load factor indicator (1) and flight envelope limits for airspeed (2), bank angle (3) and angle of attack (4)

4.3.2. NOVEL DISPLAY INDICATIONS

This section elaborates on the new display elements implemented within the PFD.

LOAD FACTOR

The first addition is a load factor indicator to the left of the airspeed tape (Fig. 4.5). The new indicator consists of a tape showing the load factor currently acting on the aircraft. Similar to the speed and altitude tapes, the indicator is of the inside-out style: the aircraft is fixed and the reference scale is moving. The reference scale has major tick marks every 1g and minor tick marks every 0.5g. The flight envelope limits are indicated by horizontal lines that attach to vertical lines running away from the fixed reference line. The flight envelope limit is shown in yellow. When the aircraft leaves the safe flight envelope, the thickness of the vertical line on

the associated side increases linearly according to the piston principle. The horizontal yellow and red lines stay fixed at their positions on the moving scale to provide a quick indication of the distances to the flight envelope boundaries. An example of an excessive load factor maneuver is shown in the sequence of Fig. 4.5. The big red line at the top of the rightmost figure gives a clear 'pitch down' cue to the pilot. Approaching and crossing the lower limit exhibits a similar but mirrored sequence on the lower part of the scale.



Figure 4.5: Load factor indicator progressively reaching and eventually exceeding the upper limit

AIRSPEED

The haptic system provides speed cues on the pitch axis of the side stick, because pitching up or down is an effective method to control airspeed (next to controlling the throttle). In order to make it clear to the pilots that the pitch cue is actually a speed cue, an indication is added to the speed tape rather than the pitch ladder (Fig. 4.4). For the overspeed protection, the standard overspeed barber pole at V_{MO} is replaced by a protection and maximum limit indication similar to that of the load factor. The protection is always 20kts below the maximum speed. Once the aircraft crosses the protection limit, a gentle nose up command is encouraged by the haptics.

A similar indication on the lower side of the speed tape corresponds to the low speed part of the flight envelope, where the haptics will eventually encourage a nose down command. Midway between the yellow and red limit, the stick shaker will activate to alert the pilot of an impeding stall.

One potential issue with the above described representation is that the nose has to go up for the speed to go down and vice versa. The way the speed tape is oriented, leads to indications that are not adhering to the principle of the moving part. [99] A big red line at the top of the speed tape might be interpreted as a nose down cue while the proper thing to do is to pull the nose up. The other indications (bank, load factor and angle of attack) do give cues in the correct direction. However, since the speed tape on the A320's current PFD already has an indication for overspeed that is similar in direction to this new piston-symbol, it can be considered an acceptable design.

BANK ANGLE

For the bank angle protection, the piston-like indications are added below the bank indicator scale (Fig. 4.4). The limits move with the horizon – in-line with the insideout design of the PFD – while the reference aircraft symbol stays fixed. When the aircraft approaches a bank limit, this gives the pilot the sensation that the limits move towards the center of the display from the side that the aircraft is banking to. According to Wickens' principle of the moving part this helps pilots interpret the direction of the limit that matches the directional cue given by the side stick. [99] In the example from Fig. 4.4, the pilot should roll left to lower the bank angle.

ANGLE OF ATTACK

An indication for margin to stall angle of attack (AoA) is added to the PFD as shown in Fig. 4.6. The distance from the 'whisker' indications to the fixed aircraft symbol equals the margin of the current AoA to the stall AoA, similar to Boeing's pitch limit indication (PLI). [101] At the red whiskers, the aircraft is flying at its maximum AoA. A vertical line in the center of the display grows in width analogous to the piston indication from the design principle. To put additional emphasis on the importance of unloading the wing by pitching down, the lower end of the piston progressively changes to an arrow as it grows wider. The indications do not rotate with bank, to ensure that the indications are always visible and always match a pitch down command. The whiskers are placed beside the pitch ladder to not obstruct the ladder.



Figure 4.6: AoA indicators relative to the fixed aircraft symbol progressively reaching the AoA limit

4.3.3. TYPICAL WINDSHEAR RECOVERY

To illustrate the synergy between the flight envelope, display and haptic feedback, Fig. 4.7 shows the display indications during a typical windshear escape procedure side-by-side with the flight envelope and haptic profiles. The series of four frames follows the actions a pilot would typically perform.

- Frame 1: The windshear is triggered, indicated by a red windshear text on the PFD and a synthetic voice repeating 'windshear' three times. The pilot initiates the windshear procedure by applying full thrust and pitching the aircraft to 17.5° of pitch. [13]
- Frame 2: The pilot receives a tick on the stick's pitch axis, as well as an increased stick-back stiffness, to alert him that the speed is decreasing outside the safe flight envelope. On the speed tape this is shown by the current speed protruding into the yellow part of the low-speed piston. At the same time, the load factor indication shows that the aircraft is above the safe load factor limit for the current airspeed. And finally the angle of attack indication on the pitch scale starts growing in width, as the angle of attack approaches its maximum.
- Frame 3: When the aircraft continues the deceleration, the stick shaker is enabled as an additional low velocity warning. The aircraft is now very close to a stall and the big red arrow on the pitch ladder of the PFD urges the pilot to push the nose down. This is felt in the stick by an increased stiffness on the nose-up side. Additionally the neutral point of the stick is shifted forward to help the pilot lower the nose.

Frame 4: After the initial windshear recovery, the aircraft is now accelerating. When approaching the maximum velocity limit as shown here, a tick warns the pilot of an imminent excursion and the stick moves backwards to help the pilot bleed of airspeed. The spring stiffness of the stick is increased to inform the pilot of the distance to the ultimate flight envelope limit, as visualized by the widening of the piston on the speed tape.



Figure 4.7: Typical windshear recovery procedure. The left column shows the flight envelope, the center column shows an excerpt of the PFD and the right column shows the associated haptic pitch profile

4.4. METHOD

Since pilots are expected to interact with the display, its design was tested in a humanin-the-loop simulator experiment. The goal of the experiment was to evaluate the interaction of pilots with the display, see what it does to their control strategy and whether it improves their subjective perception of the haptic feedback.

4.4.1. PARTICIPANTS

Fifteen professional Airbus pilots, all male, from four airlines and one male Airbus A320 synthetic flight instructor (SFI) participated in the experiment. The experience of all participants is shown in Table 4.1. They were divided over two groups (A and B) that experienced a different display order. Four pilots previously participated in our haptic feedback evaluation, namely A5, A6, B1 and B6. It is worth noting that the second officers – while not certified to operate the aircraft below 20,000ft – did receive a complete flight training and all had first officer Boeing experience from previous positions. Of the pilots, 14 had experienced windshear on a real aircraft, of which nine in an Airbus. All pilots had received upset recovery and prevention training (UPRT) and had experienced alternate law in simulator training.

Pilot	Age	Flight hours	Airbus flight hours	Position	Type rating
A1	52	13,500	$2,000^2$	SFI	A320
A2	48	13,500	700	First officer	A330
A3	27	2,800	2,300	First officer	A320
A4	56	10,000	6,000	Captain	A330
$A5^1$	57	9,500	9,000	Captain	A320
$A6^1$	47	15,000	1,500	Captain	A330
A7	28	1,200	600	Second officer	A330
A8	25	2,300	200	Second officer	A330
$B1^1$	48	16,000	5,000	Captain	A330
B2	50	16,000	5,000	Captain	A330
B3	43	12,500	7,500	Captain	A320
B4	30	3,000	400	Second officer	A330
B5	49	13,000	2,000	Captain	A330
$B6^1$	31	5,500	5,300	Captain	A320
B7	47	13,950	3,300	Captain	A330
B8	39	8,787	6,178	Captain	A320
Mean	42	9,784	3,561		
Std. dev.	11	5,235	2,758		

Table 4.1: Participants in the experiment

¹Pilot participated in previous haptic feedback research.

²These are simulator hours. Participant is a former Boeing pilot and current SFI for the Airbus A320.

4.4.2. APPARATUS

The experiment took place in the Research Facility for SImulation, MOtion and NAvigation (SIMONA) at Delft University of Technology. The simulator's exterior and interior are shown in Fig. 4.8 and 4.9 respectively. SIMONA is a six degrees of freedom motion simulator with a full fledged flight deck shell. The interior can be configured to resemble any modern glass cockpit transport aircraft. For this particular experiment the motion system was not used.



Figure 4.8: Exterior of SIMONA at TU Delft

Figure 4.9: Interior of SIMONA at TU Delft

An electrically controlled Moog FCS Ecol-8000 side stick with force feedback capabilities as described in Chapter 3 was located on the right hand side of the pilot, who was seated in the right seat. The pedals were not used. A Boeing 777 pedestal with throttle quadrant and flaps lever, and a Boeing 737 Mode Control Panel (Flight Control Unit in Airbus terminology) complemented the interior. The outside visuals were provided by FlightGear¹ and showed the airport infrastructure, terrain and important buildings at the airport. A proprietary A320 flight dynamics model including control laws from the German Aerospace Center (DLR) was used as the simulated aircraft. Since the model did not include a landing gear, all flights were automatically stopped upon reaching 50ft above ground level (AGL).

The entire simulation was run using the Delft University Environment for Communication and Activation (DUECA) software. DUECA is a framework written in C++ allowing for easy real-time distributed simulations. [102] The PFD and ND were drawn using OpenGL (see Fig. 4.15) and very closely resembled the real Airbus displays.

4.4.3. PROCEDURE

All participants engaged in the procedure outlined in Table 4.2. They were divided over two groups, with an equal distribution of aircraft types. Group A first used the original PFD and then the new PFD, denoted as PFD+, while the order was reversed for group B. The complete experiment took circa five hours per pilot.

1. **Briefing** – At the start of the day, the pilots received a short introduction, signed a consent form and were asked to fill in a pre-experiment questionnaire on their flying and previous research experience.

Inside the simulator, the pilots were seated in the right seat. After a safety briefing, the various controls and standard displays were explained, as some of them were

¹Open source flight simulator available at http://flightgear.org

	45 min	40 min	30 min	30 min	60 min	60 min	20 min
Group A	Briefing	Famili -	Training	Lunch	PFD flights	PFD+ flights	Debriefing
Group B		arization	flights		PFD+ flights	PFD flights	

not completely resembling their Airbus counterparts. For instructional purposes, the original PFD was temporarily moved to the left screen – the normal location of the ND – while the right screen showed the haptic profile and the flight envelope.

Without the model in the loop, hence by the simulator operator changing the state of the aircraft directly, all haptic cues were explained. The pilots were asked to close their eyes while experiencing all cues once again to check whether they had understood the explanation of the various cues. Next, the PFD+ was shown and all cues were thoroughly presented and experienced once again.

2. **Familiarization** – For familiarization with the model and controls, a simple righthand circuit to runway 36L at Amsterdam Airport Schiphol (EHAM) was flown twice with the baseline PFD. Note that this is a non-standard approach and therefore no instrument landing system or precision approach path indicator was provided.

Next, the pilots performed the following exercises over the North Sea to experience the haptic cues:

- (a) Pilot induced stall by maintaining altitude with idle throttle.
- (b) Overspeed by full throttle and pitching down.
- (c) Nose-dive followed by a strong back stick input to reach the high-g region.
- (d) Rolling to the left and right.
- (e) Pitching up as far as possible with closed eyes, while keeping the aircraft at the onset of stall.

Upon completion of these exercises, the same circuit as before was flown once more, this time with the PFD+. The haptics were left unchanged with respect to the previous circuit. After one circuit the same manoeuvres were flown over the North Sea as before, apart from the closed-eyes exercise.

- 3. **Training** The training phase was setup to more closely resemble operational flights and prepared the pilot for the actual measurement flights. Four approaches were flown towards runway 16R of Seattle Airport (KSEA), for which the layout is shown in Fig. 4.10. The baseline PFD was used on the first two approaches, while the novel PFD+ was present on the latter two approaches. The conditions per flight are shown in Table 4.3. After each flight, the pilots were asked to fill in a questionnaire, identical to those used in the measurement runs.
- 4. **Measurement runs** For the measurement runs, the pilots were divided into two groups. Group A flew the first set of measurements with the old PFD, followed by the PFD+, Group B vice versa.

Run	Airport	Scenario	Display
1	KSEA	Windshear	PFD
2	KSEA	Runway sidestep	PFD
3	KSEA	Runway sidestep	PFD+
4	KSEA	Windshear	PFD+

Table 4.3: Training phase flights



Figure 4.10: Airport diagram of KSEA [103]

At the start of each block of six measurement runs, a go-around scenario was flown into KSEA with the PFD variant corresponding to that block of flights. This 'refreshment' run was used to give the participants a chance to re-familiarize themselves with the model, haptic feedback and when applicable PFD+ after a (lunch) break.

Thereafter, the six measurement runs were flown. Each ended with a questionnaire, followed by the presentation of a score. The airports and scenarios for these flights were assigned according to a balanced Latin square distribution. After the six flights, the pilots were asked to fill in another questionnaire about the complete set of six flights.

5. Debriefing – At the end of the experiment the pilots received one more questionnaire about their overall experience throughout the day as well as the realism of the simulator. Once the questionnaire had been filled in, the pilots were debriefed. The research question was revealed to them and any open questions that could not be answered before in order to not influence the experiment outcome were discussed at this point.

4.4.4. SECONDARY TASK

Apart from flying the approach, the pilots were given a secondary task in the form of ATC calls that they had to reply to. Each pilot's callsign reflected the company that the pilot was employed at: '{Company} 107'. To ensure the pilots had to pay attention to the ATC calls, two other aircraft from the same company were introduced with flight numbers 685 and 713. ATC could ask to 'report heading', 'report speed' and 'report altitude'. Random realizations were made for each condition, to ensure that all pilots received the same

ATC commands in the same condition. A callsign and command were selected from a uniform distribution. These were then triggered at a delay after the previous command, determined by a normal distribution ($\mu = 20$ s and $\sigma = 2.5$ s). The texts were read out loud by a female American-English accent from the Festival¹ text-to-speech generation library, developed by the University of Edinburgh.

4.4.5. INDEPENDENT VARIABLES

Three independent variables were used in the experiment: the airport (two levels), the scenario (three levels) and the display (two levels). In total there were therefore 12 different conditions. To reduce variance in the data, all pilots experienced the same conditions. However, to mitigate order effects, a randomized balanced Latin square was used. The airport and scenario were varied constantly, while the display variant was fixed during a series of six consecutive flights in order to prevent pilots from having to re-adapt to the available cues all the time.

AIRPORT

Approaches were varied between runway 26L at Hartsfield–Jackson Atlanta International Airport (KATL) and runway 09L at London Heathrow (EGLL). Both airports have runways on either side of the terminals, with comparable spacing (KATL: 1340 m, EGLL: 1420 m) and more or less adjacent thresholds. The layouts can be found in Fig. 4.11 and 4.12. An instrument landing system (ILS) was available on the approach runway, with corresponding indications on the PFD. The pilots were provided with approach charts including a schematic of the runway layouts.





Figure 4.11: Airport diagram of EGLL [103]

Figure 4.12: Airport diagram of KATL [103]

Each flight started circa 12 NM from the airport in trimmed flaps-up condition at 215 kts and an intercept heading of circa 45°, towards the final approach fix (FAF) on the localizer. At EGLL the starting position was circa 3 NM right of the localizer, while it was circa 4 NM left of the localizer at KATL. Fig. 4.13 shows a typical trajectory towards EGLL.

¹Available at http://www.cstr.ed.ac.uk/projects/festival/



Figure 4.13: Flight plan for EGLL (not to scale). Start (1), localizer interception at FAF (2), scenario triggering point (3) and end of flight (4)

SCENARIO

The pilots were presented with three scenarios. These were automatically triggered upon descending through a pre-determined altitude given in Table 4.4. In all scenarios, a stable and variable wind was introduced according to the distribution used in Chapter 3. This wind was identical for all pilots.

1. **Windshear** – The windshear was implemented using the standard take-off wind model from the FAA with wind components as shown in Fig. 4.14. [76] An approach windshear model was not used because it was found not to ensure that the aircraft would fly near the limits of the flight envelope. In the training runs, the strength of the windshear was reduced while keeping the same distances, as indicated by the dashed lines in Fig. 4.14.



Figure 4.14: Windshear model, based on Ref. [76]; the dashed profile was used in the training runs

In accordance with the FCOM procedure, the pilots were told to not change the configuration of the aircraft, apply full thrust, pitch up to an initial attitude of 17.5° and adjust pitch as necessary to control altitude loss. [13] The lack of Speed Reference System (SRS) pitch guidance upon windshear encounter was explicitly briefed. When out of the shear pilots were asked to climb to the missed approach altitude at which the simulation was halted.

- 2. **Runway sidestep** ATC would make either of the following calls, depending on the airport: '{Company} 107, sidestep right to runway 09 right, cleared to land' (EGLL) or '{Company} 107, sidestep left to runway 27 right, cleared to land' (KATL). Pilots were briefed to try to line up with the new runway as quickly as possible without using extreme bank angles.
- 3. **Go-around** When ATC made the following call '{Company} 107, go-around', pilots were supposed to perform the go-around procedure by climbing to the missed approach altitude with a climb rate of 2000 ft/min at runway heading.

Table 4.4: Scenario triggering altitudes

Scenario	Airport	Triggering altitude			
		ASI	L [ft]	AG	L [ft]
		PFD	PFD+	PFD	PFD+
	EGLL	1500	1500	1420	1420
Windshear	KATL	2500	2500	1475	1475
	KSEA	1700	1650	1270	1220
	EGLL	1200	1200	1120	1120
Sidestep	KATL	2100	2100	1075	1075
	KSEA	1500	1500	1070	1070
	EGLL	800	1000	720	920
Go-around	KATL	1700	1900	675	875
	KSEA	1200	1300	770	870

DISPLAY

Two variants of the PFD were used in the experiment (Fig. 4.15): the original PFD was a replica of the PFD on the real A320, while the new PFD+ had several new indications as discussed in Section 4.3. The A320-like ND was the same throughout the experiment and always showed the final approach fix and runway threshold as waypoints (Fig. 4.15c). This display also showed the current throttle and flap settings to compensate for the absence of their normal indicators in the simulator.



Figure 4.15: Displays used in the experiment

4.4.6. CONTROL VARIABLES

The aircraft model and haptic feedback settings were the same in all flights. The aircraft had a total mass of 64,841.7 kg and was in clean configuration at the start of each flight. All flights took place in alternate law. In terms of haptic feedback, the protection and maximum limits in roll were set to 15° and 30° respectively on all flights. These are considerably smaller than the 33° and 67° used by Airbus ([13]) and have been chosen to ensure that pilots would actually encounter the (artificial) limits, as pilots do not bank beyond circa 30° in normal operation. To ease recognition of these adjusted limits, the crosses on the PFD's bank scale that normally indicate the limit at 67° were moved to 30° for the experiment. Pilots were briefed on these stricter limits, but also asked to fly like they would normally do.

4.4.7. DEPENDENT MEASURES

Objective and subjective measures are used to assess the display in terms of performance, safety and pilot appreciation.

OBJECTIVE MEASURES

Data from the simulator were automatically logged at a rate of 100Hz. All objective measures were afterwards computed from this data, except for the secondary task score which was indicated on paper during the run.

- Control activity Root mean square of the stick deflection angle in degrees.
- Margins to flight envelope limits Both the flight envelope limits and aircraft states were measured in terms of airspeed, angle of attack, load factor and roll angle. The flight envelope margin was computed off-line.
- Performance scores, dependent on the scenario, were used for two reasons. First
 and foremost to assess whether the pilot's performance changed in the experiment
 and second to communicate to the pilots in order to encourage them to improve
 themselves throughout the experiment. The scores were defined as follows:
 - Windshear Total altitude loss in feet from start of windshear till lowest point during recovery.
 - Sidestep Smallest distance in nautical mile to the threshold of the new runway at which the aircraft was more than 300ft offset to either side of the localizer of that runway.
 - Go-around Ratio of time during climb at which vertical speed was between 1500 and 2500ft/min, measured from 100ft above the trigger altitude till 100ft below the missed approach altitude.
- Workload through a secondary task: ratio of correct responses to ATC requests.

SUBJECTIVE MEASURES

Subjective measures were collected through questionnaires at various times throughout the experiment.

- After each flight:
 - Workload through a RSME questionnaire. [85]
 - Situation awareness through two questions on a linear scale ranging from 'Never' (0) to 'Always' (100):
 - 1. Did you have the feeling you were in control?
 - 2. Did you have the feeling you missed essential information?
 - Usefulness of each haptic axis (pitch and roll) and when flying with the PFD+
 each new display element in helping the pilot to stay within the limits of the flight envelope through a five-point Likert scale question per item labeled as *not at all, slightly, moderately, very* and *extremely*.
- After both consecutive sets of six flights:
 - System acceptance through Van der Laan rating ([104]) and Modified Cooper-Harper rating. [105]
 - Five-point Likert scale questions on three statements, with labels at the minimum (*disagree*), middle (*neutral*) and maximum (*agree*).
 - Questions on usefulness of individual haptic and display properties in helping the pilot to stay within the flight envelope limits. Five-point Likert scale labeled as *not at all, slightly, moderately, very* and *extremely*.
 - Open question on what haptic cue(s) and/or display element(s) to add to the system, if any.
 - Open question on what haptic cue(s) and/or display element(s) to remove from the system, if any.
- At the end of the experiment:
 - Question on the pilot's display preference (PFD or PFD+) in combination with the haptic system.
 - Five-point Likert scale statements on the haptics, display and experiment with a minimum (*disagree*), middle (*neutral*) and maximum (*agree*) label.
 - Five-point Likert scale question on the safety effect of the system, with a minimum (*unsafer*), middle (*unchanged*) and maximum (*safer*) label.
 - Five-point Likert scale questions on the realism of various simulation aspects with a minimum (*unrealistic*), middle (*acceptable*) and maximum (*perfect*) label.

Apart from the questionnaires, pilots were actively encouraged to verbally communicate any questions, remarks and thoughts throughout the day. Since all pilots were native Dutch, all questionnaires and instructions were in Dutch.

4.4.8. HYPOTHESES

Based on the dependent measures, the following hypotheses are formulated:

- H1 Workload Workload in terms of control activity is expected to be lower with the PFD+ compared to the original display since the pilot can anticipate the limits. With a lower workload for the primary task, secondary task performance is expected to increase.
- H2 **Performance** In a similar fashion it is also predicted that the addition of a visual display will improve the overall performance of pilots flying with haptic feedback.
- H3 **Safety** Safety metrics are expected to follow risk homeostasis theory (a trade-off between performance and perceived level of risk). [86] However, it is assumed that pilots consider the edge of the safe flight envelope as the maximum allowable risk. It is therefore hypothesized that the margins to the ultimate flight envelope limits will be larger when flying with the PFD+. Additionally, pilots can anticipate the limits in contrast to the haptic *feedback*.
- H4 **Pilot appreciation** On a subjective level, pilots are expected to show greater appreciation for haptic feedback when combined with the PFD+ as the display should help them understand the haptic cues that they receive, one of the issues raised by pilots in the previous haptic system evaluation.
- H5 **Indicator usefulness** It is expected that the load factor display brings the least improvement compared to the old display as the respective limits are mostly encountered in combination with other limits. The angle of attack indication is expected to be most useful as it provides critical information that is currently not directly communicated to the pilot.

4.5. RESULTS

Several events warranted the selection of data, as some flights could not be used for the main analysis. Subsection 4.5.1 elaborates on this selection. The results are then split in objective results as shown in Subsection 4.5.2 and subjective results in Subsection 4.5.3 that stem from the questionnaires. Whenever statistical tests are performed, these are Wilcoxon signed-rank tests with a 95% confidence interval, unless explicitly stated otherwise.

4.5.1. DATA SELECTION

Sixteen pilots participated in the experiment, each flying 12 measurement conditions. Some flights in which a simulator hiccup, before reaching the scenario trigger point, prevented proper execution were restarted. Two pilots crashed their aircraft during the measurement flights by not recovering from a stall upon windshear occurrence. Pilot B2 crashed on the first measured windshear, while pilot B5 crashed on his second windshear. Both where flying with the PFD+ when they crashed and had already experienced two successful windshears in the training flights. B2 indicated after the flight that he did not follow the procedure from the FCOM, but relied on the AoA indication on the PFD+. B5 did not provide an explanation but showed similar behavior. Those flights have been started over without telling the pilots that they would encounter the same condition again. One other PFD+ flight was re-started when the pilot (A5) entered a stall while turning to final, before reaching the scenario trigger point. According to his own analysis he lost his concentration. The crashed and canceled flights are excluded from the results, unless explicitly mentioned.

4.5.2. OBJECTIVE RESULTS

All flown tracks for both airports are shown in Fig. 4.16. The freedom of the pilots to choose their flight path is clearly visible. Some pilots steered away from the localizer to give themselves a smaller intercept angle, while other pilots steered towards the localizer to overfly the FAF while lined up with the runway. While the intercept angle varied per pilot, this was found to be constant irrespective of the display. Furthermore, in the go-around and windshear scenarios many pilots did not maintain runway heading even though that was instructed. Pilots also utilized various flap extension strategies leading to vastly different approaches in terms of airspeed and corresponding flight envelope limits.

This freedom comes with several challenges for the analysis of the data. For a fair comparison, each flight is therefore cut into two sections based on the following criteria:

- Approach (APP) From the start of the flight until the triggering of the scenario, performed in every flight.
- Windshear (WS) From the onset of the windshear until the aircraft is stable at the missed approach altitude.
- Runway sidestep (RW) From sending the command to the text-to-speech generator till reaching 50ft AGL.
- Go-around (GA) From sending the command to the text-to-speech generator till stable at missed approach altitude.



Figure 4.16: Flight tracks of all flights combined, colored per pilot

Looking at all the other variables in the data, three more points should be raised. First, not all pilots managed to fly the approach speed of 140kt when the windshear was triggered. Notably pilots A1 and A5 had much higher velocities, generally this corresponds to a smaller loss of altitude. These pilots were consistently flying fast, irregardless of the display variant (Fig. 4.17). In both groups the airspeed is (slightly) lower in the second series of flights. Second, two flights stand out with a very high AoA of up to 29°. The pilots of both flights provided full back stick upon encountering the windshear. One of the pilots explained that he inadvertently thought he was flying in normal law. Third, flap extension time is different per pilot, leading to different performance during the initial approach phase.



Figure 4.17: Mean indicated airspeed per pilot at start of windshear ($V_{APP} = 140$ kt)

Figure 4.18: Mean windshear altitude loss per pilot

TYPICAL DATA

Fig. 4.19 shows data for all of the protected variables on a typical flight for the windshear scenario. The flap adjustments are clearly reflected in the maximum speed limits, as well as in the maximum permitted AoA. When turning onto the localizer, the pilot exceeded the 15° roll limit activating the haptic feedback on the roll axis. During the windshear, the pilot was in the AoA protection zone for circa five seconds, and exceeded the maximum AoA limit very briefly. Finally, a small airspeed violation can be seen on the climb out to the missed approach altitude when the pilot did not retract the flaps upon acceleration.

PERFORMANCE

Overall there seems to be little effect of the display on the performance scores, but there are some differences between the two airports. Especially in the windshear scenario at KATL, the PFD shows a much larger spread than the PFD+ (Fig. 4.18), which is not observed at EGLL. The other scenarios only showed marginal effects and thus their data is not visualised here for brevity. At EGLL the PFD+ leads to a slightly lower sidestep score, while at KATL the PFD+ has a higher score. Finally the go-around also shows a small difference, with more low scores at KATL than at EGLL. Wilcoxon signed-ranks tests show no significant differences for any of the performance scores. Windshear at EGLL (Z = -0.724, p = 0.469) and KATL (Z = -1.293, p = 0.196), sidestep at EGLL (Z = -1.028, p = 0.304) and KATL (Z = -0.159, p = 0.874) and finally the go-around at EGLL (Z = -0.035, p = 0.972) and KATL (Z = -0.175, p = 0.861).



Figure 4.19: Typical flight data: Pilot A2 flying a windshear scenario at EGLL with the PFD+

SECONDARY TASK

Combining the flights of all pilots, there were 734 ATC calls that required a reply. Just 22 of those were not or incorrectly answered. Further analysis shows that the vast majority of ATC requests that were missed occurred while the aural windshear or stall warnings were active, or when the pilot was already transmitting a message, and not the result of workload differences. The ratio of correct replies is therefore not a useful workload measure in this experiment.

TIME OUTSIDE THE SAFE FLIGHT ENVELOPE

The mean time spent outside the various limits of the safe flight envelope is shown in Fig. 4.20, where only flight phases are shown for which there was more than one excursion in the entire experiment. Roll protection limit excursions ($\phi > 15^{\circ}$) mostly occurred during the localizer interception and in the runway sidestep. Only during one windshear the roll protection was very briefly activated, while it was never activated in the go-around phase. Fig. 4.20a shows that in both approach and sidestep the excursions were slightly shorter with the new display. A Wilcoxon's signed-rank test shows that the change in approach is significant (Z = -3.206, p < 0.01) while in the sidestep it is not (Z = -1.034, p = 0.301). For the maximum roll limit ($\phi > 30^{\circ}$), there were too few violations to run a similar analysis.



Figure 4.20: Mean times in protection per pilot; flight phases in which one or zero excursions into the protection limits were registered are not shown

Speed excursions were primarily seen during windshear and approach (Fig. 4.20b). In approach these excursions were generally caused by a decreasing maximum speed upon flap extension. When climbing out of the windshear, flaps were often retracted too late while the airspeed increased rapidly. In windshear, pilots seem to spend less time in the high speed protection with the PFD+, but this decrease is not significant (Z = -1.619, p = 0.105). A similar, but significant, effect is seen during approach (Z = -2.521, p = 0.012). In the sidestep and go-around there were too few overspeed moments for any statistical analysis. The maximum speed was only exceeded once, during a windshear with the original PFD.

As expected, the angle of attack limits are almost only exceeded during the windshear. Fig. 4.20c shows the time spent above the protection limit. Only one pilot (A1) never exceeded the AoA protection limit. There was a small decrease in time with the PFD+ that is not significant (Z = -0.795, p = 0.427).

CONTROL ACTIVITY

The root mean square (RMS) control deflections of the side stick are given in Fig. 4.21 and 4.22 for, respectively, pitch and roll. Control activity is highest in the pitch axis during the windshear scenario. In the roll axis, most control activity is seen during the sidestep and to a lesser extent during the approach phase. There are no significant differences between the two displays, even though pitch control activity appears slightly higher in windshear with the PFD+ (Z = -0.879, p = 0.379), while roll control activity seems slightly lower in the sidestep (Z = -0.465, p = 0.642).

4.5.3. SUBJECTIVE RESULTS

Apart from objective data, subjective results were collected through a series of questionnaires. The results are discussed per questionnaire, starting with the questionnaire that was presented after each single flight. Followed by the questionnaire that wrapped up a series of six flights with a single display configuration and finally the questionnaire that was posed at the end of the experiment.





Figure 4.22: Mean RMS roll input per pilot

POST-RUN QUESTIONNAIRE

A short questionnaire after each single run allows to see how the display and haptics are experienced in the three scenarios. Fig. 4.23 shows the answers to the question '*Did you have the feeling you missed any essential information?*'. Wilcoxon signed-ranks tests indicate that the display had a significant effect on both the lack of information in the windshear scenario (Z = -2.691, p = 0.007) and sidestep scenario (Z = -2.121, p = 0.034). The go-around scenario showed no significant results (Z = -0.756, p = 0.450). In the windshear scenario, 11 of the 16 pilots indicated a lower score on the lack of essential information question when using the PFD+, while for three pilots the display version did not make any difference. Especially the angle of attack indication was said to be missed on the original PFD.



Figure 4.23: Post-run question: Did you have the feeling you missed any essential information?



Figure 4.24: Post-run question: Did you have the feeling you were in control?

No significant difference between displays is observed for any of the scenarios in the control metric regarding the question *'Did you have the feeling you were in control?'* (Fig. 4.24). During the windshear pilots feel slightly more in control with the new display, in correspondence with the indicated lack of information. Ten pilots indicated an improvement with the PFD+ in windshear, five pilots a decrease and one pilot was indifferent to the display variant. Overall most pilots had the feeling they were less in control in the windshear scenario than in the other scenarios.

In terms of subjective workload, the RSME scores, averaged over the two flights per scenario, show that the pilots perceived the highest effort in the windshear scenario (Fig. 4.25). The effort in the sidestep scenario is less and comparable to that in the go-around scenario. There are no statistically significant differences observed between the two displays.



Figure 4.25: Mean RSME scores per pilot

When asked about the usefulness of the haptic feedback on the pitch and roll axis of the stick, it can be seen that the pilots considered the haptic pitch cues most helpful in the windshear scenario (Fig. 4.26). Haptic feedback in pitch did not help in the sidestep scenario but provided some help in the go-around scenario. Roll feedback was scored as somewhat helpful during the sidestep, but much less so than the pitch cues in windshear. In the other scenarios roll cues were not so helpful. For both axes there is no significant change in subjective haptic usefulness between the two display variants.



Figure 4.26: Subjective usefulness ratings of the haptic axes in helping to stay inside the flight envelope limits

Results of a similar usefulness questionnaire regarding the various display indications are shown in Fig. 4.27. It reveals that pilots consider the airspeed indication useful in all scenarios, but especially in windshear. The AoA indication is even more useful in the

windshear scenario and for some pilots also in the go-around. The indication of bank is somewhat helpful during the sidestep scenario but not in the other scenarios. And finally the load factor indication is almost never useful according to the pilots, who often mentioned that they did not look at it at all.



Figure 4.27: Subjective usefulness ratings of display elements in helping to stay inside the flight envelope limits

POST BLOCK QUESTIONNAIRE

The Van der Laan ratings, that were collected after six consecutive flights with one of the display options, are shown in Fig. 4.28 after being averaged per category. [104] The ratings show a small insignificant positive effect of the PFD+ on usefulness (Fig. 4.28a). No such difference is observed in the acceptance of the system (Fig. 4.28b). Nevertheless the spread did reduce in both categories when the PFD+ was used. When splitting the two groups of pilots, the mean of the usefulness rating of the first batch of six flights appeared to be higher than that of the second batch, irrespective of the display order. Apparently the pilots considered the system less useful once they had practiced more with it. The mean of the acceptance rating did not change much between the first and second batch, but group B shows a greatly reduced spread with the PFD+, whereas group A does not. One pilot from group A gave the lowest rating of all pilots on both usefulness and acceptance when flying the PFD. His ratings were significantly higher with PFD+. As shown in Fig. 4.28c, only two pilots gave the system a negative usefulness rating, both when flying with the PFD. The PFD ratings show a strong correlation between usefulness and acceptance with a Pearson correlation coefficient $\rho = 0.877$, while the correlation is weaker with the PFD+ ($\rho = 0.757$).

To get a better understanding of what might have lowered their ratings, the pilots were asked what they would remove from the haptic system, if anything at all. No differences were observed between the two displays, with the exception of the neutral point shift at high speed. Two pilots would like to remove this cue with the PFD, but not with the PFD+. The neutral point shifts in general were not noticed by the pilots unless they explicitly paid attention. One pilot attributed this to his 'flying with my finger tips'. Furthermore, four pilots that would like to see the tick removed were annoyed by the strict limits in bank. They also considered the tick in pitch a nuisance when extending the flaps brought them above the 20 kts margin towards the maximum speed limit while still below the maximum flap extension speed. The tick itself was said to have the potential of a good attention grabber, as long as the limits are set to realistic values.



Figure 4.28: Van der Laan Ratings

The same question was asked regarding the display indications, assuming that the haptics would not change. The load factor is the only indication that should be removed according to a majority of 11 pilots (four from group A, seven from group B), with the other indications receiving at most three nominations in total for removal.

The Modified Cooper-Harper (MCH) ratings in Fig. 4.29 show little differences between the old and new PFD, except that the spread is less with the new display and there are less ratings of 4 and worse. When looking at the PFD+ ratings for each group separately, it can be observed that the rating is 3 on average for group A, while it is 2 on average for group B. To get from a 2 to a 3 or vice versa, one must answer differently on the question '*Does the system support efficient decision making?*' A MCH rating of 1 or 2 indicates that this question was answered with '*yes*', while a rating of 3 or more can only be chosen when the question is answered with '*no*'.



Figure 4.29: Modified Cooper-Harper ratings

Asking about the usefulness of the various haptic cues in preventing envelope excursions, all cues except for the stick shaker are considered more useful with the PFD+ (Fig. 4.30). The increasing stiffness and shifts of neutral point stand out with considerably higher ratings. The tick is slightly more useful with the PFD+, while the stick shaker is considered slightly less useful. The number of '*not at all*' ratings for the tick and neutral point shifts correspond to the similar number of pilots that indicated that these should be removed from the system.



Figure 4.30: Usefulness of haptic feedback cues

The same question was asked about the various elements of the display indications (Fig. 4.31). The indication of the protected limit (beyond which the safe flight envelope is exited) is considered just slightly more useful than the indication of the maximum flight envelope limit. Despite a slight inclination towards useful there is no clear consensus between the pilots on whether the thickening of the indication is a useful aid in preventing envelope excursions.



Figure 4.31: Usefulness of display cues



Figure 4.32: Preferred display

POST EXPERIMENT QUESTIONNAIRE

At the end of the experiments, the pilots had to fill in one final questionnaire. When asked which system had their preference, most pilots in both groups indicated that they would like to see the haptics combined with the PFD+ (Fig. 4.32). In group B the preference is less pronounced than in group A, but there is still a small majority in favor of the PFD+ over the original PFD.

Apart from this binary question, several statements were posed to get a better understanding of how the pilots experienced the system and the experiment itself. The results are shown in Fig. 4.33. From the figure a slight positive effect of the PFD+ on understanding the haptic cues can be observed. With the PFD, which lacked an indication for the overspeed protection at 20kts below V_{MO} , numerous pilots experienced ticks in the stick that they could not explain. Pilots also indicated to be able to return faster to the safe flight envelope upon exceeding the envelope when using the PFD+. Almost all pilots were of the opinion that their understanding of the haptics and display increased throughout the experiment; the so-called 'learning effect'. Nevertheless, a small majority of pilots thought the system does not require lots of training. The vast majority of pilots is of the opinion that the system would help prevent critical situations and if such situations do occur that the system would help solve them. In fact, almost all pilots thought implementation of such a system would have a positive effect on safety; only one pilot thought safety would be unchanged. Finally, there is no consensus on whether the display is too distracting. Pilots that said it was distracting, often attributed this to the strict bank angle limits leading to - when being accustomed to normal bank limits - premature warnings on the bank scale.

In terms of simulation fidelity, all aspects of the experiment are considered acceptable or better by the vast majority of pilots (Fig. 4.34). Two '*unrealistic*' ratings on flight dynamics and weather were given by pilot A7, who also gave the lowest rating of all pilots on the side stick and ND. The other '*unrealistic*' rating for weather was by pilot A8. There were considerable comments on the flight dynamics model, primarily about the thrust setting not matching that of a real Airbus and a too high sensitivity in pitch, which were also primary complaints in our earlier research. In terms of weather, some pilots thought the windshears were too strong compared to their usual training scenarios and



Figure 4.33: Subjective post experiment ratings

some attributed the effect of wind on the aircraft to the weather system. The projected environment (terrain, airport and sky) was rated acceptable or better by all pilots.



Figure 4.34: Subjective simulation ratings

When taking a closer look at the two – for this experiment – most important simulation elements, the side stick and the PFD, it is clear that both are sufficiently realistic. The nominal feeling of the side stick was considered at least acceptable by all but one pilot. Several pilots commented that the pitch and roll axes are more separated in the real stick, allowing for separate inputs in either axis. With the simulated stick it was said to be difficult to only apply pitch inputs without inadvertent roll input.

Pilots were in general also very positive about the realism of the PFD, saying it resembled the real instrument very well. Most criticism was about the nervousness of the speed trend vector and occasional disappearance of the flight path vector (FPV). The FPV only disappeared during the training sessions at KSEA when the aircraft was flying a heading of exactly 180°. This was only discovered on the third experiment day and was therefore left unfixed for the remainder of the experiment. The ND scored mostly acceptable or better, although some pilots missed the track indication from the real aircraft to help them line up with the runway.

4.6. DISCUSSION

Previous research has indicated that adding visualizations to haptic feedback improves user acceptance and possibly also performance and safety metrics. [92] The current experiment indeed shows a slight improvement in acceptance and safety with the newly designed display. It does not, however, show an increase in performance. The following discussion is split in parts that follow the hypotheses. It concludes with the experiment setup and an overall system evaluation.

4.6.1. WORKLOAD

There were only some small changes observed in control activity both in positive and negative direction, depending on the scenario. All changes lacked statistical significance. The secondary task, replying to ATC requests, actually turned out to be unusable for workload analysis due to the small number of ATC requests. A comparable result was seen in Chapter 3, therefore future research should consider using different secondary tasks to aid the measurement of workload. The subjective RSME rating, however, showed no change in workload either, nor did any of the pilots hint on a change in workload in the debriefing. Thus, it is reasonably safe to conclude that the PFD+ does not lead to a change in workload, rejecting hypothesis H1. The fact that there is no increase in workload makes the PFD+ an acceptable addition in terms of workload.

4.6.2. PERFORMANCE

Concerning the go-around scenario, several pilots indicated that it was 'unusual' to maintain 2000ft/min on go-around so they sometimes forgot to pay attention to the vertical speed. Another possible cause of the low scores for this scenario is the standard procedure to start reducing the rate of climb some 10% below the target altitude, while the score was based on the climb rate up to 100ft (ca. 5%) below the missed approach altitude.

While the performance measures in the sidestep and go-around scenarios were not expected to see significant improvements with the PFD+, there were strong expectations that the AoA indication would lead to better windshear performance. In theory it allows pilots to fly at the maximum performance of the aircraft, reducing the altitude lost during recovery. This is, however, not reflected in the results. A possible explanation is that the indication persuaded pilots to pitch up further than the standard 17.5° dictated by procedures. A larger pitch angle makes it harder to recover the aircraft once stalled. Limiting the indication to a fixed maximum – similar to Boeing's pitch limit indicator (PLI, [101]) – to prevent excessive pitch (pilot following symbol) may diminish this problem. Another potential source of poor performance was the ambiguity of the AoA indicator's reference. Aligning the 'whiskers' such that they touch the upper side of the fixed aircraft symbol when the angle of attack margin is zero would solve this ambiguity, while also making it easier to 'ride' on the limit. Concluding, the new display seems to neither significantly improve, nor deteriorate performance. Hypothesis H2 is thus also rejected.

4.6.3. SAFETY

During windshear, pilots flying the PFD+ spent slightly less time outside the safe flight envelope at high AoA and airspeeds. The decrease in time in overspeed protection during

the approach phase clearly shows that the stringent 20kts high speed margin, only visible to pilots with the PFD+, changed pilot behavior when clearly communicated. Similar behaviour was seen in roll. While the time spent in roll protection also significantly reduced, the artificially strict bank limits may have had a big impact on pilot behavior in roll. In order to ensure the pilots would enter the roll protection, the bank angle limits in the experiment were artificially reduced compared to the real aircraft. Many pilots indicated this was unrealistic and perceived the roll cues in the haptic system and display as a nuisance since they activated while flying at a bank angle perfectly acceptable in normal operation. A different scenario setup may allow for the standard bank angle limits to be used. Nevertheless pilots did respect the bank limits more when shown on the PFD+. The hypothesis H3 that the margin to the flight envelope boundaries would become larger with the PFD+ can, however, not be accepted due to a lack of statistically significant differences. There does seem to be a small effect of the display that warrants further research.

While the objective effect on safety was rather limited, a large share of the pilots does expect that the system would improve safety when implemented. The data do not provide an answer on whether that can be attributed to the haptics, the display or the combination of both. Chapter 3 does suggest that the haptic system by itself is already seen as a safety improvement so the effect of the display may be limited here.

4.6.4. PILOT APPRECIATION

Overall, most pilots preferred the PFD+ over the old PFD, suggesting an improved acceptance of the haptic system in combination with the new display. This is confirmed by the increased usefulness of the various haptic cues with the PFD+. Still, Van der Laan and MCH ratings did not indicate a significant change in appreciation of the system as a whole. A possible explanation for this is that the haptic feedback, which was always enabled, was a more prominently present novelty for the pilots and thus had a bigger impact on their system-wide ratings than the display. Testing a baseline condition, with no haptic feedback, would show the effect of just the haptic feedback. Previous evaluations of the haptic feedback did include such a condition, but did not use the Van der Laan and MCH rating scales. Based on the preceding, hypothesis H4 cannot be unequivocally accepted.

4.6.5. DISPLAY INDICATIONS

As hypothesized, the load factor indication was considered the least useful indication by the pilots. They often indicated that they did not look at it at all for mainly two reasons. Firstly, it is simply not needed, because whenever the load factor limits are reached there is always another limit crossed (in the conditions from the experiment that is indeed true). The other reason is that the indication is added to the left of the speed tape, where in the actual Airbus there is nothing. The new indication was therefore not included in the scanning pattern. It is worth noting that several pilots considered the addition of a load factor indication 'extremely useful' during the briefing at the start of the experiment, but then changed their opinion after flying with it. More training may improve this, but combining all results it is expected that a load factor indication can be removed

to reduce visual load and to make the display fit in the standard Airbus display size.

The AoA indication on the other hand was much more appreciated by the pilots. The only pilot that said to remove it, does like to see the AoA and load factor indications in certain critical situations, like windshear or terrain avoidance maneuvers. Although it did not bring the expected performance benefit, it gave pilots the feeling that they were better informed about the state of the aircraft. It is probably also the reason why the stick shaker is considered less useful with the PFD+, as stall information is now also communicated through the AoA indication. Hypothesis H5 is thus accepted.

4.6.6. EXPERIMENT

Looking back at the experiment itself, the use of two pilot groups with different display orders was a valid choice, as some dependent measures showed a stark contrast between the two groups. This can probably be primarily attributed to the learning effect. Haptic feedback was new for almost all pilots and those that did fly with it before did so over a year earlier. Even though the pilots received considerate training, they were clearly still getting more accustomed to both the simulator and researched systems as the experiment progressed. Subjective results may have also been affected by the fact that the pilots did not fly a baseline condition with haptic feedback disabled and the original PFD. This could have helped determine whether any changes are caused by the haptic feedback itself, or the display.

In the aim for realistic scenarios, pilots were given a lot of freedom which lead to challenges in the data analysis. It could help to limit this freedom in future experiments. For example by showing the route on the ND all the way from the start, instead of from a distant waypoint onward. Using the autopilot to bring the aircraft to a pre-determined state and hand-over control to the pilot on the onset of an event may also help and is an accepted method in flight training. [106] As with any simulator experiment, the simulator itself may also have influenced the results. To minimize the impact of differences between the real aircraft and the simulation, pilots were given considerable time to familiarize themselves. Together with this research's focus on the PFD and side stick, both rated as sufficiently realistic by the pilots, the differences with respect to the real aircraft are considered be with insignificant influence on the outcome.

Finally the lack of motion may have influenced pilot behavior. Especially in stall conditions, pilots are known to over-react when they do not feel the load factor. [107] Displaying the load factor was expected to make up for this lack of information. However, as discussed before, pilots did not pay much attention to the load factor indicator so this can not be assumed to be an adequate replacement. An experiment involving haptic feedback, PFD+ and motion cueing should be conducted to see whether motion has any effect.

4.6.7. OVERALL SYSTEM EVALUATION

Wrapping up, the PFD+ brings no big improvements nor any large deteriorations. Since pilots seem to like the display, albeit with a couple of modifications, the integration of input and output space seems to be a feasible solution. The display appears to fulfill its main design goal: increasing the understanding and appreciation of haptic feedback as a way to communicate flight envelope boundaries. Particularly in approach scenarios, in which a substantial number of LOC-I accidents in recent years occurred during cruise. Testing the haptics and display in a cruise situation where the pilots suddenly find themselves in alternate law could show the potential of the system in a wider range of flight phases.

4.7. CONCLUSION

This study looked into the effect of providing a visual display as a complement to haptic feedback in communicating flight envelope boundaries. The resulting display design is unique for displaying not only the limits of the flight envelope or the limits of the control inputs, but combining both in one display. In addition, the display shows the direction and force of the haptic feedback that is applied to the side stick. To accomplish this, the standard A320 PFD has been enhanced with new indications for angle of attack, airspeed, bank angle and load factor. The display was evaluated by inviting 16 professional Airbus pilots to TU Delft's SIMONA research simulator, where they flew two approaches in each of three different scenarios with both the original and modified PFD.

Unlike hypothesized, the design presented in this article did not yield significant differences in performance compared to the original PFD. Small but significant changes were observed in the time spent outside the safe flight envelope regarding roll angle and airspeed, hinting on a potential safety improvement. On the subjective front, the display proved to result in a small increase in pilot appreciation of haptic feedback. The display increased the pilots' understanding of what the haptics were trying to communicate and helped pilots stay within the limits of the safe flight envelope.

In conclusion, the proposed display can help increase pilot appreciation of haptic feedback. The combined system can lead to an improvement in aviation safety by reducing LOC-I accidents. Future research should focus on improving the display and experiment design. The unused load factor indication should be removed to reduce clutter. In contrast, especially the AoA indication appears to be useful, but also lead to a number of crashes when pilots followed it too closely. Further research is therefore suggested to improve this particular indication and reduce its ambiguity. It is also recommended to test the display with the actual bank limits, instead of the reduced ones used in this research.

5

JUST FEELING THE FORCE: JND FOR ASYMMETRIC VIBRATIONS

Previous research showed that haptic feedback, in the form of asymmetric vibrations, can be used to provide a cue and a direction to the operator in a laboratory setting. Nevertheless, it is unclear how these vibrations should be designed for pilots controlling their aircraft using a side stick, in terms of shape, amplitude, frequency, or background force. This chapter aims to determine the magnitude and shape for which vibrations can still be perceived as directional cues, for one fixed frequency based on literature. The threshold magnitude of two forcing function shapes (triangular and sawtooth) was determined for both pulling and pushing cues in a just-noticeable-difference experiment. Participants were asked to report the direction at varying input magnitudes while exerting different offset force levels on the stick at different stick positions. Results confirmed all hypotheses: they indicated a lower perception threshold for the asymmetric sawtooth shaped vibration compared to a triangular shaped; higher offset force decreased the threshold in the opposite direction; and stick position had no effect on the obtained thresholds. Based on the experiment we advise for this stick geometry to use sawtooth vibrations with an amplitude higher than 0.094 Nm.

This chapter is published as:

Title	Just Feeling the Force: Just Noticeable Difference for Asymmetric Vibrations
Conference	IEEE International Conference on Human-Machine Systems, Rome (Italy), 2020
Authors	D. Van Baelen, J. Ellerbroek, M. M. van Paassen, D. A. Abbink, and M. Mulder

5.1. INTRODUCTION

H^{UMAN-MACHINE} interface environments, such as the flight deck of an aircraft, provide an abundance of visual and auditory information. Haptic feedback, i.e., force feedback through the control device, presents a different and direct way of communicating with the operator, but is still little used. Within haptics, support ranges from simple 'attention'-demanding cues to haptic shared control mixing automation and human input. [108] Haptic shared control involves an automated system actively moving the control device, which could be unwanted when an operator is accurately controlling a vehicle near its limits. Vibrations, on the other hand, provide a cue to the operator without imposing a control input which makes them useful for accurate control.

Literature on vibrations shows two similar and parallel lines of research. In the first line, operators are only perceiving the signal and not actively controlling; the second line of research looks at applications involving active control. An example of the first line of research, Tappeiner et al. investigated an asymmetric vibrations applied to an operator holding a magnetic flotor. [109] The vibration was asymmetric *in time*: the 'rise time' differs from the 'fall time', see Fig. 5.1a. Their analysis showed that such a system can indeed be used to provide directional cues, yet requires more research when the operator is actively using the control device.



(a) Asymmetric-in-time and symmetric-in-amplitude vibration [109]



(b) Symmetric-in-time and asymmetric-in-amplitude vibration [71]

Figure 5.1: Asymmetric vibrations studied in this work

An example of the second line where operators are perceiving and also actively controlling, is a haptic lane departure systems when driving a car. Navarro et al. used pulse inputs when a lane departure was imminent and showed that participants were more inclined to follow these commands as they act as a 'motor priming' element. [36] Huang et al. investigated three variations: the forcing function shape (square, triangular pulses), amplitude (large/small) and frequency (20/10/3 Hz). The analysis of lane departures showed that a signal with small amplitude, a square shape and mid-frequency was the best compromise for practical applications. [71] These two examples show the use of vibrations which are asymmetric *in amplitude*: the upper and lower parts of the oscillation are not equal as shown on Fig. 5.1b. Note that next to these two groups of vibrations, asymmetric-in-amplitude and asymmetric-in-time, multiple other vibrations exist, yet this chapter limits itself to these two groups.

Although the last two examples show that providing a directional force cue to an actively-controlling operator is feasible, it can happen that the operator experiences the *asymmetric* amplitude as a *symmetric* vibration with a shift in mean force, losing the directional information. Hence both the asymmetric-in-time and asymmetric-in-amplitude vibrations required more investigation to transfer them from a laboratory setting to an application.

The work presented in this chapter is part of a project which applies haptic cues for increasing pilot awareness when controlling aircraft close to the flight envelope limits. [97, 110] First evaluations of our haptic flight envelope protection system indicate that pilots preferred rather simple cues which indicate that the aircraft is close to the flight envelope limits through a discrete 'tick on the stick'. Ideally, such a 'tick on the stick' would not interfere with pilot control actions (as he or she may be operating close to the flight envelope boundaries), so its magnitude should be small. But not too small, because then pilots may not perceive it at all. One of the downsides of the vibrations we used was that these did not provide advice about which direction to steer, unlike the haptic shared control forces, while this direction is valuable information when operating an aircraft near the limits.

When a pilot is flying, the side stick can be at different positions where the vibrations might have a different effect due the grip of the side stick or other bio-mechanical effects. Additionally, the side stick in consideration has a centering stiffness, which requires a force of the pilot at a certain deflection. This offset force might influence the effect of the vibrations following Weber's law: higher offset forces lower the perceivability. [111]

As we want to investigate the use of vibrations on a typical side stick manipulator to transmit both a triggering and directional cue, the goal of this research is to find the asymmetric vibration shape, its advised magnitude, and whether it depends on the side stick position or force exerted on the side stick. Therefore, we will determine the minimum amplitude which two shapes of asymmetric vibrations require such that pilots are able to distinguish a direction while actively exerting a force on the control device: the Just Noticeable Difference (JND) for an asymmetric vibration.

This chapter is structured as follows. Section 5.2 discusses the forcing functions design rationale. Section 5.3 describes an experiment set-up to obtain the JNDs in asymmetric vibrations. Results are shown in Section 5.4 and discussed in Section 5.5. The chapter ends with conclusions in Section 5.7.

5.2. DESIGN OF ASYMMETRIC VIBRATIONS

All forcing functions used in this chapter are intended to indicate a *direction*, which requires an *asymmetry*. This asymmetry can be either in time or in amplitude. An asymmetry in time shows itself as the difference in rise and fall time as can be seen on Fig. 5.1a. It results in a side stick which is accelerating more in one direction. Asymmetry in am-

plitude is a difference in the magnitude to one side as shown on Fig. 5.1b. Using such a function makes the side stick move mostly to one side.

Aside from the forcing function, the actions of the pilot need to be considered too. The closest control task found in literature is a lane keeping task for which the use of asymmetric-in-amplitude forcing functions was considered. [36, 71] Nevertheless, the literature available on asymmetric-in-time forcing functions shows that these can be of interest and hence a combination of asymmetry in time *and* in amplitude is used in this research and specified here.

Before the actual forcing functions can be discussed, one more component needs to be addressed, the link between the forcing function and the pilot: the side stick dynamics. Whereas some literature might not fully specify the dynamics involved, initial implementations of forcing functions found that the side stick dynamics can have a large impact on the perceivable forces and available functions. As such, this system is first discussed in Subsection 5.2.1, followed by the forcing function specification in Subsection 5.2.2.

5.2.1. SIDE STICK DYNAMICS

Stick dynamics are governed by a simple mass-spring damper system, representing a side stick used on the flight deck of commercial aircraft (with inertia, $m_{ss} = 0.2 \text{kgm}^2$, spring stiffness, $k_{ss} = 35.68 \text{Nm}/\text{ rad}$, damping $b_{ss} = 0.4 \text{Nm}/\text{ rads}$). Limb dynamics are modelled by a spring-mass-damper system, representing the inertia of the lumped neuro-muscular system and the damping/spring dynamics of the skin combined with limbs (with inertia, $m_l = 0.07 \text{kgm}^2$, spring stiffness, $k_l = 400 \text{Nm}/\text{ rad}$, and damping $b_l = 12 \text{Nm}/\text{ rads}$). [112] These are combined in a lumped system as shown in Fig. 5.2.



Figure 5.2: Schematic representation of the lumped limb and stick dynamics

The lumped system state-space matrices are shown in Equation 5.1 and Equation 5.2. It contains four states (side stick position and velocity x_{ss} and \dot{x}_{ss} , and the limb position and velocity x_l and \dot{x}_l), two inputs (neuro-muscular force F_{nms} , and force on the side stick F_{ss}), and three outputs (side stick position, limb position, and the contact force F_c which is the combination of the side stick and contact dynamics).
$$A = \begin{bmatrix} 0 & 1 & 0 & 0\\ \frac{-k_{ss} - k_l}{m_{ss}} & \frac{-b_{ss} - b_l}{m_{ss}} & \frac{k_l}{m_{ss}} & \frac{b_l}{m_{ss}} \\ 0 & 0 & 0 & 1\\ \frac{k_l}{m_l} & \frac{b_l}{m_l} & \frac{-k_l}{m_l} & \frac{-b_l}{m_l} \end{bmatrix} \text{ and } B = \begin{bmatrix} 0 & 0\\ \frac{1}{m_{ss}} & 0\\ 0 & 0\\ 0 & \frac{1}{m_l} \end{bmatrix}$$
(5.1)

$$C = \begin{bmatrix} 1 & 0 & 0 & 0 \\ 0 & 0 & 1 & 0 \\ k_l & b_l & -k_l & -b_l \end{bmatrix} \text{ and } D = \begin{bmatrix} 0 & 0 \\ 0 & 0 \\ 0 & 0 \end{bmatrix}$$
(5.2)

To investigate effects of the side stick dynamics, and the importance of specifying all parameters in scientific publications, an asymmetric sinusoid as described by Tappeiner et al. is applied (amplitude 0.25Nm, frequency 2Hz, offset 0.25Nm, asymmetry -1.5) to the system described before, and a system where the side stick stiffness is doubled ($k_{ss_2} = 2 \cdot k_{ss} = 71.36$ Nm/rad). The forcing functions for both systems are shown in Fig. 5.3a. From the resulting limb position, Fig. 5.3b, it can be seen that it is approximately *halved* when doubling the stick stiffness. The contact force, Fig. 5.3c, shows that equal magnitudes are obtained, yet frequency content differs.

A full analysis of the effects of side stick dynamics on the resulting observations to the pilot is beyond the scope of this thesis. Clearly, they play an important role in the design of haptic feedback systems. The side stick properties used in remainder of the chapter are as described with the first system.





(b) Side stick position (x_{ss})



Figure 5.3: System responses to an asymmetric input

⁽c) Contact force (F_c)

5.2.2. VIBRATION SPECIFICATION

With the side stick defined, we continue with the forcing function design. The effects of both asymmetric-in-amplitude and -in-time have shown to be beneficial for perceiving direction, and a combination of these two asymmetric dimensions could perhaps be used. An example is the input illustrated in Fig. 5.3a, which is asymmetric-in-time due to the difference in rise and fall time, *and* asymmetric-in-amplitude due to the positive offset resulting in only positive added forces.

The frequency of the signal, as well as the level of (a)symmetry, is heuristically determined by changing the settings in the simulator and striving for best noticeable direction by the experimenter. The final settings are a frequency of 2Hz and an asymmetry of -1.5, following the definitions used by Tappeiner et al. Next, the dynamics of the side stick are taken into account to further simplify the forcing function design: a forcing function with a sawtooth-shape was found to have a similar effect as the (slightly more complex) asymmetric sinusoid. This is illustrated in Fig. 5.3 (black and blue lines).

Fig. 5.3a shows that the asymmetric sinusoid is a smooth function, and the sawtooth contains a discrete step at each start of an 'oscillation'. Nevertheless, the dynamics of the side stick filter this input, as can be seen in the resulting limb position (Fig. 5.3b) and contact force (Fig. 5.3c) where the results with both forcing functions are very similar. Hence, in the remainder of this chapter, the sawtooth-shaped forcing function is used. To compare it with a signal that is only asymmetric in amplitude, we will compare it with a triangular pulse (see Fig. 5.1b) used in automotive applications.

5.3. METHOD

The experiment was approved by the Human Research Ethics Committee, and all participants signed an informed consent form. Eight participants took part in the experiment, with ages ranging from 23 to 51 years (mean 30 and standard deviation 8.9). None of the participants reported a medical condition that limited sense or use in the hand they used to hold the side stick. There were both left- and right-handed participants. Handedness should not be a determining factor, as pilots on the flight deck are controlling either as pilot-in-command (left seat, side stick at their left) or co-pilot (right seat, side stick at their right) position, regardless of their preferred hand.

5.3.1. APPARATUS

TU Delft's Human-Machine Laboratory was used as shown in Fig. 5.4, which features a custom-made, hydraulically driven side stick (of which the dynamic properties are easily changed), located at the right-hand side. To the left, a throttle quadrant is present, of which a toggle-button is used to let the participant input a direction after each trial. In front of the participant, a display as shown in Fig. 5.5 is placed to request an input on the side stick, a timer, an indication of the buttons, and a stop bar when a staircase is finished. The stick properties are tuned to match an Airbus-type stick ($m_{ss} = 0.2 \text{kgm}^2$, $k_{ss} = 35.68 \text{Nm/rad}$, $b_{ss} = 0.4 \text{Nm/rads}$); this ensures that results are transferable to our main application. [97]



Figure 5.4: Simulator inside view showing the throttle quadrant (left), screen (front), and side stick (right)



Figure 5.5: Display provided to the participants; the traffic light indicates the required input change by the participant and is shown in the 'start'-phase, the right arrows show which direction is selected during the 'wait for input'-phase, and the top bar indicates the time in one run

5.3.2. INDEPENDENT VARIABLES

The goal of this experiment is to obtain a threshold magnitude of the forcing function for which a pilot can indicate the direction while actively controlling the side stick. Therefore the participants are asked to exert a force on the side stick equal to the force required to deflect the Airbus side stick 0.1rad in pitch (3.57Nm). The deflection is chosen as pilots in our previous experiment were, even in emergency scenarios, controlling mostly within 0.1rad deflection. [110] As the forcing function should be felt when the pilot is pulling, pushing, or not actively using the side stick, the experiment is performed when the pilot is exerting negative (NEG, -3.57Nm), positive (POS, 3.57Nm), and no force (NO, 0Nm) on the side stick.

Previous research has shown that operators manipulating a side stick are more sensitive to differences in forces compared to positions, therefore the main analysis will be at one position (FORW, 0.1rad forward) and with the three force levels. [113, 114] To validate this assumption, one forcing function is additionally tested at zero neutral position (MID) with no force and with back force resulting in -0.1rad deflecting (AFT). If the assumption is indeed valid that the threshold is mainly determined by our sensitivity to force and far less by position, the comparison of the results should show similar thresholds. As discussed already in Subsection 5.2.2, two forcing functions are selected to determine their respective threshold: a sawtooth (SAW) and a triangle (TRIANGLE) shape with a frequency of 2Hz. Only the former is used to analyse the assumption on the position and force interaction.

This makes in total 8 conditions (2 forcing functions \times 3 force levels at 0.1rad, and the sawtooth function at the middle position with no force and -0.1rad).

5.3.3. PROCEDURE

Several 'trials' are performed where a single forcing function is presented. Trials consist of three phases, see Fig. 5.6. In the first phase the 'traffic light' display is shown, Fig. 5.5 helping participants apply the proper offset force. After a full second of proper offset force, the run phase is started, during which the vibration signal is applied. An end phase without vibration, of 0.5s closes off the trial. Participants can then, with their left hand on the toggle button, indicate in which direction the vibration input was felt.



Figure 5.6: Summary of the phases in one trial

A staircase procedure is performed: after the participant experiences a cue, he/she is asked whether a change was observed, i.e., one 'trial'. The magnitude of the cue is decreased on a correct answer, and increased on an incorrect one. [115] The starting magnitude is so large that all participants start with a correct answer; the decrease in magnitude converges to the limit where the participant is able to sense a change. To improve staircase accuracy, following the first incorrect answer the amplitude increases on each incorrect answer and decreases following two consecutive correct answers.

The event of switching from a correct to an incorrect answer, and vice versa, for a staircase is called a *reversal*. Step sizes are determined by how many reversals have passed, the decrease is 30% for the first three reversals, 15% afterwards; the increase is 60% for the first three reversals, 30% afterwards, where the percentages are calculated based on the current magnitude. A staircase is completed when seven reversals are encountered.

Our experiment differs with respect to a 'standard' staircase application: for every condition mentioned with the independent condition, *two* staircases are performed in parallel. One staircase looks for the threshold where a participant can correctly indicate a direction for *positive* forcing functions (representing a pushing cue), one staircase looks for the *negative* threshold (pulling cue). The trials presented to the participant are a mix of the two staircases. Whether the direction of the next cue is positive or negative, is randomly chosen from a binomial distribution. Random directions are selected until *both* staircases had seven or more reversals.

The entire experiment for one participant lasts about 1.5 hours. After an initial safety and experiment briefing, a training is performed in which the participant is given feedback on his/her answer. This training is at least four runs with the sawtooth function and four with the triangle shape, and is concluded when the participant feels confident with the procedure. Following this, the above-mentioned eight conditions are executed following a randomized Latin-square design, ensuring that all transitions of conditions are distributed over the participants, where one condition lasts about five minutes. Between each condition a small break is held, and after four conditions a larger break is added.

5.3.4. DEPENDENT MEASURES

Each condition results in two threshold values for the forcing function force amplitude. These are calculated by averaging the last four reversals of a single staircase. In the following, a pushing/positive threshold is coded with 'UP', a pulling/negative threshold 'DOWN'.

Statistical analysis is performed using the R-programming language, for which the 'PMCMR' and 'PMCMRplus' packages are loaded. The package manuals provide references to the theoretical background such as the method, and the definition of the *p*-value and test statistics (χ^2 and *V*).¹

5.3.5. HYPOTHESES

We want to determine the most effective asymmetric vibration shape, its advised magnitude, and whether it depends on the side stick position or force exerted on the side stick. Following this aim, the results of the JND experiment are expected to follow four hypotheses:

- The threshold force where a direction can be indicated of the sawtooth-shaped forcing function is lower compared to the triangle-shaped one. This is due to the sawtooth-shaped being asymmetric in both time and amplitude, whereas the triangle is only asymmetric in amplitude.
- 2. The threshold force where a direction can be indicated of any forcing function is lower in the opposite direction from the force exerted by the participant. This because opposite direction from the force exerted has a lower background force, hence following Weber's law.
- 3. The threshold force where a direction can be indicated is lowest, and equal for pulling and pushing cues, when exerting no force on the side stick. As background forces are equal, this follows again from Weber's law.
- 4. The threshold force where a direction can be indicated is primarily dependent on the force applied by the pilot, not by the position measured. Following the perception research where operations were found to be more sensitive to differences in forces compared to positions.

¹Available using respectively https://cran.r-project.org/web/packages/PMCMR/PMCMR.pdf and https://cran.r-project.org/web/packages/PMCMRplus/PMCMRplus.pdf.

5.4. RESULTS

An example of the result of a single staircase is shown in Fig. 5.7. The structure of the staircase is especially visible in the negative one: initially each correct answer decreases the amplitude. After the first incorrect answer, two consecutive correct answers are needed to decrease again. Note that after the fourth reversal, the step size is decreased.



Figure 5.7: Example staircase-procedure (Participant 3), when applying a backward force on the stick and using a triangular pulse; numbers next to the points indicate the trials' sequence, vertical lines indicate reversals

The resulting JND levels for all participants over all conditions are illustrated in Fig. 5.8. Note that the coding of the conditions is as given in Subsection 5.3.2. Some general trends can be observed: first, the spread in JND values for TRIANGLE are much higher as compared to those from SAW. Second, the JND values for TRIANGLE are higher. Third, differences in JND for pushing and pulling forces (UP versus DOWN) while varying the force applied by the participant (POS/NO/NEG) seem to be present for both forcing functions, yet are more visible for TRIANGLE. Fourth, the absolute maximum median JND value found for SAW is 0.094Nm.

Statistical analysis of all forward-positioned cues (FORW) using a Friedman Rank Sum Test showed that there are significant changes between conditions ($\chi^2(11) = 63.1$, p < 0.001). Further investigation in the differences was performed with pairwise comparisons using Conover's test, for which the Bonferroni corrected *p*-values are shown in Table 5.1.

Finally, a statistical analysis to investigate the position effect was performed: a paired Wilcoxon test was used to compared the JND values found for forward position with no force (SAW/FORW/NO) with middle position (SAW/MID/NO), separately for the pushing and pulling forces, as well as for forward position with backwards force (SAW/FORW/NEG) and backwards position (SAW/AFT/NEG). These tests show that there is no statistically significant change (for all p > 0.37). In more detail, comparing no force forward and



Figure 5.8: Boxplots of the JND values obtained for all conditions, circles indicate outliers, crosses indicate points for pushing cues, plus-signs for pulling cues

mid positions for pushing cues gives V = 23 and p = 0.55, for pulling V = 24 and p = 0.46; with the participant pulling on the side stick and comparing forward and aft positions for pushing cues V = 11 and p = 0.38, for pulling V = 18 and p = 1.

5.5. DISCUSSION

Considering the first hypothesis, a comparison needs to be made between the sawtooth and the triangle signals. Especially comparing the *p*-values indicated in Table 5.1 by subscripts 1, equal conditions (same force applied, and same forcing function direction) can be compared. These tests show that there is a clear difference between both forcing functions except for the NO/UP and POS/DOWN conditions. The differences can also be seen from Fig. 5.8: a decrease for the sawtooth-shape can be verified, together with a decrease in variation. This supports the hypothesis of decrease in threshold force when using the sawtooth-shape function instead of the triangle-shape.

For the second hypothesis, we consider the individual conditions (forcing function TRIANGLE or SAW, force level POS or NEG) and compare the UP and DOWN values. From Fig. 5.8, there seems to be a difference present for the triangular-shaped forcing function: a positive force by the participant (POS) results in a higher positive JND and visa versa; for the sawtooth-shaped functions this is less evident. This supports the second hypothesis: participants are more sensitive to forcing functions in the opposite direction of the force applied. Studying the *p*-values indicated by subscripts 2 in Table 5.1 this observation is not supported, however, with statistical significance. Because of the small sample size and the less powerful non-parametric statistical tests, the lack of statistical significance is not considered sufficient to reject the hypothesis either.

The third hypothesis requires no (statistically significant) change between the UP and DOWN thresholds when the participant is applying no force on the side stick (NO). Looking at the FORW/NO conditions for both forcing functions gives a visual confirmation that this is the case. The *p*-values indicated by subscripts 3 in Table 5.1 confirm this with clear statistical significance, hence supporting the third hypothesis.

The fourth and last hypothesis compares the results of the sawtooth-shaped forcing function for different positions. Visually comparing the results for the no load at forward and mid positions, and negative load at forward and aft positions, only small changes are observed. This is confirmed by the Wilcoxon tests, which showed no statistical significance. Hence all evidence supports this hypothesis: a participant is more sensitive to the amount of force applied compared to the position of the side stick.

Regarding the experiment, some other observations are worth sharing. First of all, we only had a rather small sample, with just eight participants affecting the statistical significance of our results. Second, whereas the side stick properties were designed to resemble Airbus sticks as close as possible – to allow for quick implementation in practice, our participants were not real pilots. However, there is no reason to believe that pilot JNDs will be different than the 'average' human, and we can assume that better averages can be found with a higher number of participants.

These results can be directly used on the side stick on the flight deck of an Airbus A320 and A330, and can be extrapolated to cars and other vehicles which are controlled using a side stick. Additionally, the methodology can be used to determine similar properties for a control column, used in some other aircraft types, as well as other control devices.

Note that our participants only focused on the force they had to exert on the side stick as well on the direction they perceived. In the intended application, pilots are actively operating the aircraft with a more specific task, for example flying an approach. Pilots are not fully focused on what he/she feels through the haptic feedback and the resulting threshold force is higher. When implementing a forcing function as researched in this experiment, a safety factor should be applied to make sure the pilot is feeling the force. A more time-consuming approach to circumvent this issue is a new experiment where the participants' focus is actively drawn away from the haptic feedback, yet a similar threshold task is performed in parallel.

An issue surfaced during this experiment, which is applying two parallel staircases: one participant's strategy was to input 'forward' whenever a pushing force was felt, and in *any* other case, whether nothing was felt, or the cue was not clear, a 'backward' input was given. This resulted in a good approximation of the threshold in the forward direction, whereas the backward direction approaches zero without any reversals. As the threshold values determined by the reversals assume a stochastic element in both staircases, results became invalid, the participant was removed and another one invited. To circumvent this problem, one can instruct participants to 'indicate direction, if not sure pick random input'.

TRIANGLE					8	SAW	V				ĺ			
100	PUG	NEG		F CO	DOC	Ĩ	ND	NEG						
UP	DOWN	UP	DOWN	UP	DOWN	UP	DOWN	UP	DOWN	UP				
0.00^{***}	1.00	1.00	1.00	1.00	0.00^{***}_{1}	1.00	1.00	0.26	0.07^{*}	1.00_{2}	DOWN	NI		
0.00^{***}	0.01^{**}	1.00	0.05^{**}	0.01_1^{***}	0.00^{***}	0.47	1.00	1.00	1.00		UP	G		
0.00^{***}	0.00^{***}	0.07^{*}	0.00_1^{***}	0.00^{***}	0.00^{***}	0.01^{***}	0.14	1.00_{3}			DOWN	N	SA	
0.00^{***}	0.00^{***}	0.26_{1}	0.00^{***}	0.00^{***}	0.00^{***}	0.04^{**}	0.47				UP	0	W	
0.00^{***}	1.00_{1}	1.00	1.00	1.00	0.00^{***}	1.00_{2}					DOWN	PC		
0.00_1^{***}	1.00	1.00	1.00	1.00	0.03^{**}						UP	SC		
1.00	0.84	0.00^{***}	0.26	1.00_{2}							DOWN	NEC		
0.35	1.00	1.00	1.00								UP	L,		
0.05^{**}	1.00	1.00_{3}					***	*	*	- 	DOWN	N	TRIANGI	
0.00^{***}	1.00						indicates	indicates	indicates		UP	0	Ξ,	
0.19_{2}							p < 0.01	p < 0.05	p < 0.1	- - - - -	DOWN	POS		

Table 5.1: p-values for post-hoc pairwise comparisons using Conover's test, Bonferroni correction applied, rounded to two digits; subscripts explained in text

5.6. INFLUENCE OF THE ASYMMETRIC VIBRATION

To see the effect of the asymmetric vibration, the off-line analysis performed at the start of this chapter is re-visited with the signal properties as identified in the experiment. As such, a sawtooth-shaped vibration with an amplitude of 0.094Nm and a frequency of 2Hz, shown in Fig. 5.9a, is applied to the lumped limb and stick dynamics.



Figure 5.9: System responses to the asymmetric input identified in the experiment

Fig. 5.9b shows that the resulting deflection remains below 0.004rad. The effect of this position is different for each controlled system. The side stick on the flight-deck, in our intended application, requests a load factor change of 1.5g when deflecting the side stick fully at 0.28rad (16°). [61] Therefore the deflection of the vibration results in a requested change in load factor of 0.022g, comparable to mild gusts. Additionally, this is comparable to the load factor experienced when performing a turn with a bank angle of 0.2rad (12°). Most airlines limit the bank during normal operations for passenger comfort to 0.52rad (30°), which is a turn with 1.15g, hence our vibrations stays well within the limits of passenger comfort. Nonetheless, since the vibrations are intended as a cue for the pilot, not a control action, more investigation is needed to reduce the effect of those vibrations on the vehicle dynamics.

5.7. CONCLUSIONS

We investigated the perceivability of asymmetric haptic force cues designed to indicate *direction*. Results show that a sawtooth-shaped signal has a lower threshold and is recommended for future implementation. When participants applied different levels of force on the stick, data trends indicate a lower threshold for cues *opposing* the applied force; a non-significant effect, however. In case no force is applied on the stick, the threshold is equal in for/aft directions. Our participants were more sensitive to forces as compared to positions. The findings allow us to design asymmetric-in-amplitude *and* -in-time vibrations, which provide pilots with a clear direction cue, even when actively controlling their aircraft. This will be used in future developments of a haptic flight envelope protection system.

6

EVALUATING ASYMMETRIC VIBRATIONS FOR FEEDBACK ON FLIGHT ENVELOPE PROTECTION

The evaluation of the first design in Chapter 3 showed that the pilots appreciated the discrete cues when the flight envelope protection systems becomes active, yet it lacks a clear direction. Therefore, Chapter 5 investigated what vibration shape and magnitude is required to communicate a clear direction. This chapter evaluates a haptic feedback design using those vibrations with real pilots in a fixed-base simulator. The evaluation involved 24 active PPL/LAPL pilots who flew a challenging vertical profile and encountered a windshear. We used a counterbalanced design to evaluate pilot behavior with and without the haptic feedback. It was expected that this haptic feedback design helped in performing a safer recovery. Results indicated no difference in the tested metrics, except for a slight increase in acceptance when providing haptic feedback. Subsequent analysis revealed a strong learning effect in the first four runs, despite the initial training. The 12 pilots that experienced haptic feedback first seemed to have a steeper learning rate than the other pilots. Additionally, when the haptic feedback was no longer provided, there was no indication of negative after effects. This suggests a potential beneficial effect of the haptic feedback on training, but further experiments are needed to elucidate such effects.

This chapter is published as:

Title	Using Asymmetric Vibrations for Feedback on Flight Envelope Protection
Conference	AIAA Modeling and Simulation Technologies Conference, Orlando (FL), 2020
Authors	D. Van Baelen, J. Ellerbroek, M. M. van Paassen, D. A. Abbink, and M. Mulder

6.1. INTRODUCTION

INTERNATIONAL aviation safety boards, such as the European Union Aviation Safety Agency EASA and the International Air Transport Association IATA, identify loss of control in flight as the key risk area, resulting in most fatalities within aviation. [2, 3] A safety issue contributing to such a loss of control is identified as the inadequate monitoring of the main flight parameters and automation modes. To ensure and improve current safety levels, loss of control events should be prevented at all times, especially looking at the expected growth of the aviation industry.

Improving the information presented to pilots is expected to help reducing the loss of control occurrences. The most straightforward measure would be to change the flight deck displays, through showing the flight-critical states on, e.g., the PFD. This display can be augmented with information on the limits of the aircraft, i.e., the flight envelope, which can improve safety by reducing the risk of violations of those limits. [51] Research on improving pilot understanding of the current status of their automation shows promising designs, yet a simple indication like a cross was not conclusively shown to be effective. [116, 117] On the other hand, once the limits are exceeded, for example in a stall, the information on the PFD can be augmented with recovery guidance which delivers performance improvements as shown in three simulator evaluations. [118]

Apart from the visual channel, pilots can also perceive information through the sense of touch. An example is the haptic interface, which provides force feedback through the control device. This form of information can have a significant positive effect when a pilot is guided along the approach path. [41, 67] Additionally it can be used to show a set of predicted controllability limits, which was shown to be used by pilots in an experiment. [52] Furthermore, our previous research showed that haptic feedback can be used to show the pilot information on the FEP system which can limit the input of the pilot to ensure that aircraft is flying within acceptable limits. [97] The evaluation of this feedback system showed a potential benefit, yet lacked conclusive data. It did indicate that not all haptic feedback cues used were equally effective. [110]

This chapter presents a further design of the haptic feedback system for FEP, which uses asymmetric vibrations to show the activation of the FEP *and* indicate a clear direction to move away from the limit. This design is simpler than our previous one, in that it lacks situation dependent changes in device stiffness, but providing pilots with direction specific 'ticks' on the stick is expected to be equally effective.

This chapter first discusses the new haptic design iteration in Section 6.2. To evaluate this new design, Section 6.3 presents an experiment where the pilots are required to operate the aircraft at the limits. In Section 6.4 and 6.5, results of the experiment are described and discussed. Finally, the conclusions are shown in Section 6.6.

6.2. HAPTIC DISPLAY

The haptic feedback design is based on the control structure similar to an Airbus A320. Full details are given in our earlier work on the design of a haptic feedback system, see Ref. [97], only the most important elements will be explained in this section. Using an Airbus-like control law structure, the working principles of the new haptic feedback design are elaborated.

6.2.1. AIRBUS A320 CONTROL STRUCTURE

Modern-day Airbus aircraft, like the A320 and the A330, all employ a FBW system. This means that there is no mechanical connection between the control surfaces and the control device. The latter acts as an interface for the pilot to provide inputs to the FCCs which then command the control surfaces with hydraulic actuators. This allows a Flight Envelope Protection (FEP) system to be used, which can check and, if necessary, limit pilot inputs, to ensure that no flight envelope limits are violated.

Longitudinal control in a FBW Airbus, when all sensors are functioning (under so called "normal law"), is provided using C^* -control, which is a combination of both pitch rate (*q*) and load factor (*n*). [10–13] On top of this control law, a hard envelope limit is employed which protects the pilot from exceeding limits on angle of attack (*a*), load factor (*n*), and maximum velocity (V_{MO}). This protection is depicted in Fig. 6.1, where the nominal flight envelope is the extreme limit which can not be exceeded, the safe flight envelope is the point where protections start acting. The envelope is constructed by the maximum (n_{max}) and minimum (n_{min}) load factor, their protection limits ($n_{max_{prot}}$ and $n_{min_{prot}}$, respectively), the maximum operation velocity (V_{MO} , and protection $V_{MO_{prot}}$), and minimum velocity ($V_{a_{max}}$, and protection $V_{a_{prot}}$).



Figure 6.1: Flight envelope, velocity (V) versus load factor (n)

When multiple FCCs fails, or when a sensor failure occurs, the control is reverted to a degraded control law. In this research, we will consider a control law close to the Airbus alternate law without reduced protections, where the same protections apply as before, only the angle of attack protection is lost. Hence, in alternate law the aircraft can be stalled, yet allowing the pilot more extreme control actions.

Lateral control in normal law is a bank (ϕ) rate command from -33° till $+33^{\circ}$ of bank. Beyond these limits, positive roll stability is achieved such that the aircraft rolls back to the protection value (ϕ_{prot}) of $\pm 33^{\circ}$. The maximum achievable bank, with full lateral side stick deflection is $\pm 66^{\circ}$ of bank. In alternate law, lateral control reduces to a pure rate command, irrespective of the actual bank angle. More details on the control laws and degraded control laws can be found in Ref. [97].

Given that for both longitudinal and lateral control, a degradation of the control law results in a different effect for a given control input, a clear indication of both the limits

and the active protections of the flight envelope is required. Nevertheless, accidents did occur where pilots were not aware of what control law was active, and what protections were engaged. [7] As such, a clear and intuitive way of presenting this information can be found in haptic feedback and a new design is proposed in the following.

6.2.2. HAPTIC FEEDBACK DESIGN

The design rationale of the haptic feedback presented in this chapter is based on the results of an evaluation with a previous design. We found that the majority of pilots appreciated a discrete cue when leaving the safe flight envelope, therefore additionally indicating when the FEP starts acting as indicated above. Furthermore, the stick shaker close to a stall, was considered to be very valuable. Although the other haptic cues used in the initial design – such as the increased stiffness of the side stick, the change of the position to which the stick returns when no force is applied – were appreciated, these were not received as well as the discrete indications. [110]

Looking at the flight deck implementation of a haptic feedback system with variable stiffness and changing neutral point, the verification, validation and certification procedure might present a huge hurdle to overcome. On the other hand, adding a 'simpler' discrete cue might be easier to implement as similar systems already exist, such as a stick shaker in Boeing aircraft, or an indication of angle of attack by a soft stop in the Gulf-stream G500/G600¹. [55] Additionally, the discrete cue might be incorporated in the grip of the side stick such that it can be retro-fitted to current passive side sticks. Combining the preference by the pilots, and the potential market, we considered it to be worthwhile to investigate whether a haptic feedback system using *only* discrete cues might already yield an improvement.

Although the discrete cues were the preferred way of communication, it was not always clear to pilots in our previous experiment what *direction* of control input was required. This is no surprise, since the discrete cues were not optimised to indicate direction, but it was a good incentive for us to investigate whether discrete cues *can* indicate direction. Therefore, an investigation into a number of forcing functions was performed in Chapter 5.

This analysis indicated that a sawtooth-shaped forcing function had best performance. The experiment showed the minimal required amplitude where the subjects could just indicate a direction, which was determined to be 0.094Nm. As the subjects only focused on the side stick and the perceived direction, the minimal amplitude for use on the flight deck is expected to be larger. For the present experiment the amplitude is multiplied with a 'safety factor' of three. The resulting forcing function has a sawtoothshape with amplitude (I_0) 0.282Nm, frequency 2Hz, lasts for one second, and is shown in Fig. 6.2. Although it has a discrete start, combining the forcing function with the actual side stick-dynamics results in a smooth experience on the stick as shown in Chapter 5.

Like the discrete cue evaluated in the previous experiment, this discrete cue is provided to the pilots when the aircraft leaves the safe flight envelope and the FEP becomes active, and the direction of the cue is the required direction to avoid the approaching limit. As such, a positive roll (i.e., to the right) cue is provided when $\phi < -\phi_{prot}$, a negative roll (left) cue is provided when $\phi > \phi_{prot}$. A positive pitch (push) cue is provide when

¹https://www.baesystems.com/en-us/feature/an-active-role, accessed June 6th 2019.



Figure 6.2: Sawtooth-shaped forcing function used

one or multiple of the following conditions are met: (i) $\alpha > \alpha_{prot}$ or (ii) $n > n_{max_{prot}}$. A negative pitch (pull) cue is provided when at least one condition is met of the following: (i) $V > V_{MO_{prot}}$, (ii) $n < n_{min_{prot}}$, or (iii) $\alpha < \alpha_{min_{prot}}$.

To inform the pilot that (s)he remains outside the safe flight envelope, a single tick is repeated every second as long as the safe flight envelope limits are exceeded. Additionally, the amplitude of the tick is used to transmit the proximity of the current aircraft state to the outer flight envelope. For a generic variable v, the intensity starts at the protection state (v_{prot}) with default value I₀, and increases to a multiple of the initial value determined by the gain K_I (2 in this setup) at the outer flight envelope (v_{nom}) using:

$$\mathbf{I} = \mathbf{I}_0 \cdot \left(1 + K_I \cdot \frac{\mathbf{v} - \mathbf{v}_{prot}}{\mathbf{v}_{nom} - \mathbf{v}_{prot}} \right)$$
(6.1)

To clarify the expected working principle of the haptic feedback system, Fig. 6.3 shows the time trace of an illustrative example: Fig. 6.3a, 6.3b and 6.3c show, respectively, the time traces of the velocity, angle of attack and resulting force supplied to the control device, i.e., the haptic feedback.

At the start of the example, the aircraft accelerates to a value just below its maximum velocity. When the protection velocity is exceeded at Frame 1, as shown on Fig. 6.3a, the two initial ticks are supplied to the side stick visible on Fig. 6.3c. These initial discrete cues have a set intensity of I_0 and are positive/backwards: indicating a pulling action is required. While remaining in the upper velocity protection, the ticks are repeated every second, and the intensity is adjusted to reflect the relative distance from the protection limit to the ultimate flight envelope limit. Upon decelerating the flight state leaves the upper velocity protection in Frame 2, the ticks are stopped and the nominal side stick feeling is resumed.

When the angle of attack increases above the protection value at Frame 3 shown on Fig. 6.3b, two more ticks are provided on the side stick as illustrated on Fig. 6.3c. Note that the direction of the tick is opposite from the previous: these ticks at intensity I_0 are negative/forward, indicating that the current problem can be resolved by pushing. By decelerating more, the velocity drops below the green dash-dotted line on Fig. 6.3a, i.e., halfway between the lower protection and lower flight envelope velocity, at Frame 4. At that point, the stick shaker is activated as shown by the added oscillations on the force to the side stick on Fig. 6.3c. The intensity of the discrete cues is again adjusted to reflect the relative distance to the ultimate flight envelope. Increasing the velocity/lowering



(c) Haptic feedback: force supplied to the side stick

the angle of attack, stops the stick shaker at Frame 5 and the discrete ticks at Frame 6, resuming nominal side stick feeling again.

Using the haptic feedback cues presented here, it is expected that the pilot is better informed on the flight envelope limits and protection zones, yet this needs to be proven by a system evaluation as proposed in the following.

6.3. METHOD

To evaluate the haptic interface design as proposed in the previous section, an experiment was performed.

6.3.1. PARTICIPANTS & INSTRUCTIONS TO PARTICIPANTS

For this experiment, 24 pilots (1 female, 23 male) with a current Private Pilot License (PPL) or Light Aircraft Pilot License (LAPL) license were invited. As these pilots are not necessarily active Airbus pilots, they were reminded that the aircraft model used has a mass of 64,000kg and has to be handled with more care than a general aviation aircraft. Additionally, they were instructed to always stay within the nominal limits of the flight

6

Figure 6.3: Illustrative example of haptic feedback triggering on changing states

envelope (black line on Fig. 6.1) which was shown on the PFD using the indications proposed in Ref. [119]. It was mentioned that each run would stop at 50ft above ground level irrespective of any other event/performance due to limitations of the simulation.

Participant	Age	Flight hours	License
1	34	116	PPL
2	36	106	PPL
3	25	80	LAPL
4	38	150	PPL
5	49	205	PPL / E-IR
6	56	130	PPL
7	40	630	PPL
8	26	152	PPL
9	48	350	PPL
10	47	160	PPL
11	24	500	PPL
12	69	800	PPL
13	66	900	PPL
14	45	120	PPL
15	46	250	LAPL
16	55	190	PPL
17	48	500	PPL
18	47	552	PPL / IR
19	50	400	PPL
20	60	170	PPL
21	33	240	PPL
22	41	80	PPL
23	53	600	PPL / E-IR
24	42	330	PPL
Mean	44.9	321.3	-
Std.Dev.	11.9	237.1	-
IR Instrum	ent Rating		

Table 6.1: Participants in the experiment

Instrument Rating

E-IR Enroute-Instrument Rating

6.3.2. EXPERIMENTAL SETUP

The experiment was performed in the Human Machine Interaction (HMI) research simulator of Delft University of Technology. It is a fixed-base, near 180° outside field-of-view, used in the first officer position of which an inside-view is shown in Fig. 6.4. Since the pilot was sitting in the first officer position, the display to his front-left was the ND showing a top-down overview of the situation, shown in Fig. 6.5a, combined with a basic engine N1-indication and slats/flaps indication. The display right in front of the pilot was the PFD showing the critical flight states, shown in Fig. 6.5b, which included display indications used to show why and when the haptic feedback was active. [119] Next to the visual information, auditory warnings were presented when the angle of attack was above the maximum value, and when the velocity was above the maximum velocity.



Figure 6.4: Inside view of the HMI flight deck



Figure 6.5: Flight deck display setup used in the experiment

A custom-made, hydraulically driven side stick with programmable dynamic properties is located at the right-hand side and was configured to Airbus side stick properties. [61] To the left, a throttle quadrant is present which could be used to control the throttle and high lift device settings. Centrally placed, a Boeing 737 Mode Control Panel (Airbus terminology: Flight Control Unit (FCU)) enabled the interface with the heading, velocity and altitude references on the displays. Outside visuals were generated using FlightGear¹ and showed the airport infrastructure, terrain and important buildings at the airport. A proprietary A320-like flight dynamics model with control laws, from the German Aerospace Center (DLR), was used as the simulated aircraft. [120]

The nominal control device settings for this experiment, including mass (m), spring coefficient (k), damping coefficient (b) and maximum deflection (δ_{max}) , for both longitudinal and lateral side stick axes are given in Table 6.2, no breakout was used as it was not present in Chapter 5.² Forcing functions used in the experiment were: (i) a sawtooth shape of intensity 0.282Nm, duration 1s and frequency 2Hz when exiting the safe flight envelope, (ii) a sawtooth shape with varying intensity proportional to the relative distance of the protection and flight envelope limit, duration 0.5s and frequency 2Hz when remaining outside the safe flight envelope and (iii) a stick shaker for low velocities as a sinusoid with frequency of 20Hz and magnitude 0.426Nm.

¹Open source flight simulator available at http://flightgear.org

²Our conference publication incorrectly provided breakout properties.

Property	Value
m	$0.2 \text{kg} \text{m}^2$
$k_{lon_{nom}}$	36.3 Nm/rad
b_{lon}	0.4 Nm s/rad
$\delta_{lon_{max}}$	0.279 rad
$k_{lat_{nom}}$	21.8 Nm/rad
b_{lat}	0.4 Nm s/rad
$\delta_{lat_{max}}$	0.314 rad

Table 6.2: Control device in the experiment

6.3.3. EXPERIMENT SCENARIOS

The haptic feedback system was designed to communicate the 'proximity of the flight envelope limits' to the pilot and therefore requires an evaluation at these limits. In previous work, this was achieved by presenting the pilots with a scenario as realistic as possible, by initializing the simulation right before the pilot would intercept the glide slope and localizer. An event was triggered after following the glide slope/localizer for several minutes. The analysis showed that, due to variation in performing this initial part of the simulation, the aircraft states (velocity/altitude/heading) presented to each pilot when the events were triggered, were not equal, which resulted in a large spread in the data. These results indicated that for the experiment it would be beneficial to better control the conditions under which the system is tested. [110, 119] Therefore, in the present experiment the required flight trajectory was more stringently prescribed, as discussed below. This ensured more uniform conditions when the emergency scenario was encountered.

FLIGHT PATH

Each run was started when the aircraft was flying 140kts (72.0mps) at 2500ft (762m) with flaps setting 3, overhead the threshold of runway 23 of Zoersel (Belgium) and aligned with the respective runway. This location was chosen as it has no special terrain features closeby, and the runway was not visible from the starting position as illustrated on Fig. 6.7a. Additionally, the auto-throttle was set to 140kts and activated, reducing the variability of the initial aircraft state when the event was triggered, which should provide more consistent results. From this position, pilots were presented with visual markers (squares of 60m by 60m) on the outside display to fly a flight profile consisting out of six 'hills', for which an example path is presented in Fig. 6.6 and is visualized as illustrated by Fig. 6.7.

One hill was 2.27NM (4200m) long and followed a saw-shaped trajectory with one of three possible amplitudes: 150ft (45.72m), 300ft (91.44m) or 500ft (152.4m). The flight path started with a horizontal segment of 0.41N*M* (750m) and one hill of the smallest amplitude as run-in. This was followed by a randomized order of hills such that each hill amplitude occurred twice in the flight path. Each flight ended with a horizontal segment of 0.54NM (1000m) as run-out. This setup of hills was chosen as it was expected that it allowed the results to be evaluated for each hill separately. Eight different realizations of the randomization were obtained to present pilots with variability in the scenarios. The resulting trajectories are all shown in Section C.1.



Figure 6.6: Flight path side-view, solid black vertical lines indicate "fly-through gates" shown on the outside visual; the thick red line indicates the trigger point of the windshear (not shown on the outside visual); the dotted blue lines lines indicate the windshear section used in our evaluation



(a) View at start of run

(b) Perspective view on flight path (viewing angle is for illustrative purpose only, never encountered during flight)

Figure 6.7: Example of the outside visual flight path visualization

EMERGENCY SCENARIO

As the pilots of the experiments mentioned before did express the potential added value of the haptic feedback system in a windshear event, this event was re-used for this research. A wind shear is a meteorological phenomenon where wind velocities are locally rapidly changing, and can be caused by a large cylinder of air suddenly "droping" towards the earth. During such a wind shear event, downdrafts can push the aircraft dangerously close to the ground. [76] To recover from this event, the pilot has to fly as close to the stall limit as possible to prevent further height loss and maximise aircraft performance.

The windshear was *always* started when the aircraft moves through the visual marker of the large amplitude hill at an altitude of 2900ft (883.92m). Each flight path contained two large hills, only one of them was selected at random to contain the windshear trigger point. The windshear itself was modeled by both a head-on and top-down component as shown in Fig. 6.8. [76] Once the windshear was initiated, the visual and aural warning triggered, and the pilot had to apply the windshear recovery procedure as stipulated in Section B.6, which was based on the Airbus Flight Crew Operating Manual. [13]

When providing only windshear as the emergency scenario, pilots might anticipate this event, even in the first run. To prevent this, two more checklists for an emergency were presented to the pilots: the actions required for a single engine stall (Section B.7), and for a sudden center of gravity shift (Section B.8). Note that the checklists presented in Fig. B.6 and B.7 are heavily modified from the FCOM, and the checklist for the sudden center of gravity shift is non-existing in the FCOM.





(a) Headwind component

(b) Downwind component

Figure 6.8: Windshear component distribution

6.3.4. EXPERIMENT DESIGN

In order to provide pilots with sufficient familiarity with the simulator and the haptics, a familiarization phase was performed, followed by measurement runs.

FAMILIARIZATION

After a briefing on the simulator safety procedures, controls and displays, the pilots *felt* the design rationale behind the haptic feedback design. This was done by presenting the flight envelope (an image similar to Fig. 6.1), the haptic feedback (a time trace of the forcing function on the side stick), and the PFD (Fig. 6.5b) to the pilot. In this setup, no aircraft model was used, yet the flight envelope state was changed directly (hence changing the velocity and load factor) and all visual, auditory and haptic cues were presented and experienced by the participant.

Next the model was introduced to the pilot by flying a traffic pattern twice to a final approach at Schiphol (EHAM) as shown on Fig. 6.9 without the haptic feedback, hence focusing on familiarization with the model. Pilots were instructed to follow the instructions as indicated. Some pilots did encounter a stall and/or an overspeed condition during these first runs. If the pilots did not hit one or both limits by their own exploration, they were asked to deliberately explore those boundaries to ensure that all pilots encountered these before the measurement runs.

MEASUREMENTS

The measurement phase contained eight realizations of the flight path presented above. They were flown in a randomized fashion, distributed over all participants using a Latinsquare distribution. The participants were divided in two groups: one group performed four runs <u>with</u> haptic feedback, followed by a break, and four runs <u>without</u> haptic feedback, the other group had reversed order of the haptic feedback: first off, then on.

After each run, pilots were asked to indicate their workload using a RSME rating ([85]), complete a post-run situation awareness questionnaire, and indicate how helpful the display and haptic (if supplied) elements were. They also provided a misery scale rating tracking effect of motion sickness. [121] Once this was completed, pilots were informed on how much time they spent inside the flight envelope, which they had to maximise.



Figure 6.9: Traffic pattern flown to runway 36L at Schiphol (Schiphol layout from AIP [84])

Table 6.3: Experimental design

Block	1				I	2			
Run	1	2	3	4	5	6	7	8	
HFG		Н	F			Ν	NH		
HSG		N	Η			Н	IF		

After one block of four runs, the pilots were asked to complete a questionnaire with a modified Cooper-Harper rating scale ([122]), and a Van der Laan-rating scale. [104] After the experiment was completed, pilots were asked to complete a post-experiment questionnaire, which contained a number of questions with Likert-scales on how they experienced the haptic feedback system.

6.3.5. INDEPENDENT VARIABLES

The experiment had two independent variables. First, the haptic feedback was withinparticipant either present (Haptic Feedback, HF), or not (No Haptics, NH). Second, participants were divided in two between-participants groups: the Haptics First Group (HFG) receives haptic feedback for four runs, followed by no haptic feedback, the Haptics Second Group (HSG) vice versa as shown in Table 6.3. Participants with an odd number in Table 6.1 were placed in the HFG, all even-numbered participants were part of the HSG.



Figure 6.10: Time trace of velocity with safety metrics indicated

6.3.6. DEPENDENT **M**EASURES

The dependent measures of the experiment are split into objective and subjective.

OBJECTIVE MEASURES

The objective measures were retrieved from the windshear recovery procedure, and focus on performance and safety. To show why metrics were chosen, a time excerpt of a windshear recovery is shown in Fig. 6.10. Another example is further elaborated in the results section, for now it is sufficient to understand that this shows a participant hunting for the best performance of the aircraft close to the maximum angle of attack.

Looking at the example, one can argue that a safe flight is performed when the aircraft state is within the flight envelope limits, indicated with the solid black line representing α_{max} . Although participants were instructed to stay within the limits at all times, at certain moments in time the pilot could control the airplane beyond these limits. A first performance metric therefore was considered to be the *time spent outside the flight envelope limits*.

Time by itself only informs about the length of the limit violations, it does not take into account the severity/safety: two different limit violations might be of equal time, yet one just slightly over the limit while another one is in a deep stall. As such, a safety metric combining both the time and the magnitude of the violation was the *integral of the variable over the flight envelope limit*.

The example also shows that a participant can operate the aircraft within the limits, yet they can either stay well clear of the limits, or push the system by flying very close to the limits. A straightforward metric to determine this safety definition was the *closest obtained distance of the state relative to the flight envelope limit*: it indicates how close to the limits the participant dared to control the airplane.

Next to this performance, and two safety metrics, one more performance metric on the overall windshear recovery procedure was available: the *total amount of altitude lost during the recovery*. Although not communicated to the participants, the maximum

altitude lost from the windshear initiation to the end of the windshear recovery is an indication of how much of the available aircraft performance was utilized by the participants.

Previous research showed that the perceived level of risk was mostly kept constant with increasing support, i.e., risk homeostasis, as exampled by a haptic feedback system in an automotive study. [86, 87] For the current experiment, risk homeostasis was defined by improved performance combined with a degradation of objective safety metrics, as pilots obtained better awareness of the risk involved when supplied with haptic feedback.

SUBJECTIVE MEASURES

Subjective measures were obtained by asking the pilot for an opinion, or experience. The categories and measures are:

- Workload: after each run, the pilot was asked to provide a RSME [85]
- Situation awareness questions: after each run, the pilot was asked to answer two questions on a linear scale (0–100) ranging 'Never' left (0), and 'Always' right (100):
 - 1. Did you have the feeling you were in control of the situation?
 - 2. Did you have the feeling you missed critical information?
- Usefulness:
 - 1. Pilots were asked after each run to rate the usefulness of all display and haptic elements
 - 2. After a block of 4 runs, pilots were asked to provide a modified Cooper-Harper rating
 - 3. After a block of 4 runs, pilots were asked to fill a Van Der Laan-questionnaire
- Pilot experience: after the experiment, the pilot was asked to fill in a questionnaire regarding the experience with the haptic feedback system. It used a five point Likert-scale where the all points are labeled.

6.3.7. HYPOTHESES

From the experiment, we expected the following when the pilot was provided <u>with</u> haptic feedback:

- 1. Risk homeostasis is present during the windshear recovery, therefore:
 - Performance of the pilots improve.
 - Objective safety metrics decrease.
- 2. Subjective workload ratings decrease.
- 3. Pilots will have an increased (subjective) situation awareness:
 - Have an improved feeling of being in control.
 - Have a reduced feeling of missing information.
- 4. Modified Cooper-Harper ratings improve.
- 5. Van Der Laan-rating scales improve.

6.4. RESULTS

To see how a pilot can use the haptic feedback system, Subsection 6.4.1 shows a time trace of a windshear recovery. Next, the objective and subjective measures are discussed in, respectively, Subsection 6.4.2 and 6.4.3. Answers to the debriefing questionnaires are shown in Subsection 6.4.4. All flown trajectories for all flight paths used are included in Section C.1, together with data from one additional participant: a very experienced commercial airline pilot, whose data was included to check the performance of the 24 less experienced pilots.

When presenting data using box plots, medians are indicated using a horizontal thick line, outliers are indicated using plus-signs; all individual data points are presented next to the boxes using crosses. Furthermore, statistical analysis is performed in R ([123]) and a significance level of 0.05 is used. When comparing the means of the haptic versus no haptic feedback conditions (HF vs NH), or within one group on different runs, a Friedman test is performed. Within single runs and between groups, results are compared using the Wilcoxon Rank Sum test.

6.4.1. TIME TRACE

Fig. 6.11 illustrates all windshear recoveries from Participant 23, and shows that they all follow a similar pattern in terms of altitude. The black line indicates a situation where the pilot made use of the haptic feedback, as shown on Fig. 6.12. During the windshear recovery, the goal is to stay as close to the limit as possible, while not surpassing it during the highly dynamic recovery phase.



Figure 6.11: Windshear recoveries flown by Participant 23, Run 3 indicated in black, vertical lines represent gates on the outside visual

During the recovery performed in this example, shown on Fig. 6.12, at Frame 1 the haptic feedback (Fig. 6.12b) informs the pilot of the approaching limit (Fig. 6.12a), the pilot reacts by reducing the input (Fig. 6.12), even pushing the side stick. When clear of the limit, a more negative input is given again, resulting in a quick encounter of the stick shaker at Frame 2. The pilot reacts by pushing hard on the side stick and succeeds

in increasing the aircraft velocity. In the dynamic situation, the wind is changing and the state of the aircraft is moving closer to the limit even though a positive/push input is provided. In Frame 3, the pilot is informed of the protection zone, yet decides not to act on it yet, only at the subsequent stick shaker, the positive input is increased and the state moves out of the protection zone. After Frame 4, two more occurrences of the initial tick occur on which the pilot reduces input slightly.



(c) Control device deflection, i.e., input from the side stick to the FCC

Figure 6.12: Example of Participant 23 using the haptic feedback during the windshear recovery in Run 3

This case exemplifies that pilots *can* make use of the haptic feedback, yet the metrics presented below should indicate *how* pilots use them. It will be further explored what the possible consequences are on the results in the discussion.

6.4.2. OBJECTIVE MEASURES

Objective measures are retrieved from the simulation states and discussed next.

ALTITUDE LOST DURING RECOVERY

For the windshear recovery, the difference in altitude between the windshear trigger point and the lowest point encountered during the windshear is calculated and averaged over the four runs in one block (following the block/run structure in Table 6.3). The results in Fig. 6.13a show that there is no difference in performance when comparing the HF and NH conditions, which is confirmed by the results of a Friedman test that indicated no statistical significance. This result is found despite the fact that the final runs with haptic feedback appeared better compared to the runs without haptic feedback during execution of the experiment.



Figure 6.13: Altitude lost during windshear recovery

To find out why no effect of the haptic feedback was found, the windshear performance is plotted for each individual run in Fig. 6.13b. This shows that, irrespective of whether haptic feedback is supplied or not, a strong learning effect is present during the first runs, resulting in unchanged means, as shown in Fig. 6.13a. Additionally, it shows that the median of the haptic first group (HFG) reduces faster compared to the haptic second group (HSG). To investigate this learning effect present, statistical analysis is performed to compare Runs 1 and 4, indicating a statistical difference for the HFG ($\chi^2 = 8.33$, p < 0.01) and near statistical significance for the HSG ($\chi^2 = 3$, p = 0.08). Using a Wilcoxon Rank Sum, no statistical significance at Run 4 is found between both groups (W = 42, p = 0.09), although the plot show a small difference in median. Friedman analysis on Runs 4 and 8 did not show a significant difference in means for both groups. In other words, there seems to be no further improvement in performance over the final four runs.

TIME ABOVE MAXIMUM ANGLE OF ATTACK

Pilots were instructed to stay within the flight envelope limits at all times. Although time spent outside the flight envelope should ideally be zero, the means for both blocks shown on Fig. 6.14a indicates that this was not always the case. Similar to the performance in terms of altitude lost during the recovery, this performance metric does not have a visual or statistical difference when providing haptic feedback.



Figure 6.14: Time with angle of attack above maximum value during windshear recovery

As also seen with that previous metric, this metric shows a quicker improvement of the performance metric in terms of each run as shown on Fig. 6.14b. There again is a visual and statistical difference between Runs 1 and 4 for both groups (HFG: $\chi^2 = 12$, p < 0.001, HSG: $\chi^2 = 8.33$, p < 0.005). Although visually the HSG appears to spend more time above the maximum angle of attack, no statistical significance using the Wilcoxon Rank Sum was found between the results of Run 4 for both groups, possibly due to the fact that numerous data-points are at zero. Nevertheless, the median of the HFG approaches zero from Run 2, whereas the median of the HSG has a median clearly above zero even at Run 4. Statistical analysis showed no difference between Runs 4 and 8 for the HFG, yet the HSG has a 'near' statistically significant difference ($\chi^2 = 3.57$, p = 0.059).

INTEGRAL ABOVE MAXIMUM ANGLE OF ATTACK

Safety during the windshear recovery is further evaluated using the integral of the angle of attack above the maximum value allowed by the flight envelope, with the per block averaged results in Fig. 6.15a. As before, no difference is observed (visually or statistically), yet Fig. 6.15b shows a difference between Runs 1 and 4 (for both groups $\chi^2 = 12$, p < 0.001). Although no statistical difference is found between both groups at Run 4, a statistical significant difference is found between Runs 4 and 8 for the HFG ($\chi^2 = 5$, p < 0.05) as all participants have near zero metric, and near statistical difference for the HSG ($\chi^2 = 3.57$, p = 0.059) although the visual change appears to be quite large.



Figure 6.15: Integral of angle of attack above maximum value during windshear recovery

HIGHEST ANGLE OF ATTACK

The second safety metric is the closest point to the maximum angle of attack, for which the per-block means are shown on Fig. 6.16a. These again show no difference. Focusing on the individual runs in Fig. 6.16b, Friedman tests show a significant difference between Runs 1 and 4 for both groups (HFG: $\chi^2 = 8.33$, p < 0.01, HSG: $\chi^2 = 12$, p < 0.001).



Figure 6.16: Highest angle of attack obtained during the windshear recovery, relative to the maximum angle of attack (positive values result in a stall warning)

While a Wilcoxon test comparing the results of both groups at Run 4 did not show statistical significance, one has to observe one critical difference: the mean of the HSG (hence without haptic feedback) is above zero, meaning that at least half of the participants reached angles of attack above the maximum. For the HFG, the median is below

zero, indicating that at least half of the participants in the HFG did not exceed the maximum angle of attack. Looking at the evolution of the metrics after the break, the HFG seems to slightly improve yet this is not supported by statistical significance. On the other hand, the HSG does have a 'near' statistically significant difference from Runs 4 to 8 ($\chi^2 = 3$, p = 0.083). In addition, it is clear that the spread of all data reduces and the median approaches zero when providing haptic feedback.

6.4.3. SUBJECTIVE MEASURES

The subjective measures are obtained after each run, those are the Rating Scale Mental Effort (RSME) and a situation awareness scale, and metrics obtained after a block of four runs, which are a Van Der Laan and MCH rating.

RSME

After each run, participants were asked to fill a RSME on a calibrated scale ranging from zero to 150, indicating how much mental load was required for the task. Results per block are shown in Fig. 6.17a, and similarly to the objective measures, have no difference (both visually and statistically). Results per run in Fig. 6.17b clearly show a learning effect: an improvement is present from the initial towards the final run. Both groups show a similar trend, and no statistical differences between groups are observed within one run.



Figure 6.17: Rating Scale Mental Effort (subjective workload)

SUBJECTIVE SITUATION AWARENESS

Situation awareness is subjectively measured by asking the pilot whether (s)he has the feeling of being in control (Fig. 6.18) and the feeling of missing information (Fig. 6.19). Ideally, a pilot always has the feeling of being in control and is never missing information. Looking at the plot for block in Fig. 6.18a and Fig. 6.19a, no differences between conditions can be seen, and no statistical difference was found after testing.

When looking at the individual runs (Fig. 6.18b and Fig. 6.19b), no clear visual differences can be seen between groups in terms of trend or final value. Statistical analysis showed a significant difference for the HFG with the feeling of being in control of Run 4 compared to Run 8 ($\chi^2 = 2.78$, p < 0.1).

An additional interesting point can be the transition from Run 4 to Run 5: going from receiving haptic feedback to no feedback (HFG) increases the feeling of missing information, and when enabling the haptic feedback reduces the feeling of being in control (HSG). Nevertheless, this is not supported by statistical significance using a Friedman test for both sub-scales.



Figure 6.18: Did the pilot have the feeling of being in control?



Figure 6.19: Did the pilot have the feeling (s)he was missing information?

MCH RATINGS

After a set of four runs with or without haptic feedback, a questionnaire queries the pilots for more high level feedback on the way of presenting the flight envelope limits. Two typical scales are used and shown here. First of all, a Modified Cooper-Harper rating scale is presented for which the results are shown in Fig. 6.20. Keep in mind that the scale ranges from ten to one, where one is the best score. Additionally, the horizontal lines on the figure indicate the tipping point for the questions on the decision tree. Both the figure and statistical tests did not show a difference between groups for either haptic feedback or no haptic feedback. Equally so, the figure and Friedman test did not show an in-between participant difference for the haptic feedback.



Figure 6.20: Modified Cooper-Harper rating

VAN DER LAAN RATINGS

Secondly, a Van Der Laan-questionnaire asks nine questions to score the system on usefulness and satisfaction, keep in mind that due to the definition of the questionnaire, a perfectly useful and satisfying system would score minus two on both scales. Fig. 6.21 shows the results for the analysis, Fig. 6.21a shows the raw scores for both groups and indicates that both with and without haptic feedback, the system is well received in terms of usefulness and satisfaction.

Fig. 6.21b shows the differences of the system with haptic feedback, relative to the no haptic feedback case. Looking at the mean of all participants, only a small improvement in usefulness is obtained. More intriguing is the mean of each group separately indicated with the bold lines. This indicates that the participants who started without haptic feedback experience the system with haptic feedback to be more useful and more satisfying. On the contrary, participants who started with haptic feedback experienced the conditions without haptic feedback as more useful and satisfying. As such, an order effect of presenting the haptic feedback is present.



Figure 6.21: Van Der Laan-ratings; black bold indicates population mean, coloured bold indicate group means; squares represent the score after a block without haptic feedback, crosses with haptic feedback

6.4.4. DEBRIEFING QUESTIONNAIRES

In order to have a structured debriefing session when all runs are completed, a questionnaire with 19 questions was presented to the pilots. The first and foremost question presented pilots a simple choice: do you prefer to fly with or without haptic feedback? Results to this first question are shown in Fig. 6.22, which follows the order effect stated before with the Van Der Laan-questionnaires: the vast majority of the HSG prefers the haptic feedback, the participants in the other group are divided: half of them prefer the haptic feedback. Most other debriefing questions used a five point Likert-scale for which questions and results are shown in Fig. 6.23.



Figure 6.22: Do you prefer to fly with the haptic feedback system?

Pilots indicate that the haptic and visual display are not distracting (respectively Fig. 6.23a and 6.23b), yet are indecisive on whether a lot of training is needed (Fig. 6.23d and 6.23e). Shown on Fig. 6.23c, they do indicate that the combination of visual and haptic feedback did not give conflicting signals (as they were designed for in Ref. [119]).



(a) The haptic interface is distracting.



(d) The haptic interface requires a lot of training.



(g) The haptic feedback system affected my workload. $^{1} \label{eq:general}$



(j) The haptic interface helps in preventing critical situations.²



(b) The visual interface is distracting.



(e) The visual interface requires a lot of training.



(h) During the experiment, my understanding of the haptic interface increased after each flight.²



(k) If a critical situation occurs, the haptics helps in resolving it.²



(c) The visuals and haptics gave conflicting signals.



(f) I was fighting the haptic interface.



(i) Using the haptic system, my knowledge on the edges of the aircraft performance changed.³



(l) When implementing this system on an aircraft, what would be the effect on safety?⁴

Figure 6.23: Debriefing Likert-scale questions. Possible answers (unless specified otherwise) were:

1) Disagree 2) Slightly disagree 3) Disagree nor agree 4) Slightly agree 5) Agree

¹ Possible answers: 1) Decreased	2) Marginal decrease	e 3) Did not change	4) Marginal increase	5) Increased
² Possible answers: 1) Agree	Slightly agree	3) Disagree nor agree	e 4) Slightly disagree	5) Disagree
³ Possible answers: 1) Increased	2) Marginal increase	 Did not change 	4) Marginal decrease	5) Decreased
⁴ Possible answers: 1) Much safer	2) Safer	3) Safer nor unsafer	4) Unsafer	5) Much unsafer
Focusing on the haptic feedback, most pilots felt that they were not fighting the haptic feedback (Fig. 6.23f), nevertheless they are indecisive on whether the workload was changed (Fig. 6.23g). For this latter question, note the order effect: the HFG mostly indicates a marginal increase in workload, whereas the HSG indicates a decrease in workload. The majority of the pilots indicate that there is a learning effect present over the run (Fig. 6.23h), and they agree that their knowledge on the edges of the flight envelope did increase using the haptic feedback (Fig. 6.23i).

As mentioned in the Introduction section, upset prevention is one of the major fields to improve in aviation, and our participating pilots agree that this haptic feedback system can help to achieve this (Fig. 6.23j). When in an upset condition, the majority of our participants think the haptic feedback system might help, yet their answer is not as convincing (Fig. 6.23k). The final Likert-type question asked the pilot for the effect when implementing this haptic feedback on an aircraft, Fig. 6.23l shows that a clear majority of the pilots believe that implementing this system can improve safety.

The final questions, asked for further textual elaboration on any of the previous questions, asked for comments on the reality of the simulation and possible final comments. These comments include:

- I was not looking at instruments anymore with the haptic feedback, instead I reacted naturally to the cues by looking outside and monitor to the behavior of the airplane.
- I used the haptic feedback to double check whether my input was correct. If it was correct, I increased the input.
- The initial tick made me look for the problem.
- With the haptic feedback, I made more subtle movements and anticipated more.
- The haptic feedback indicates the problem, the visual display gives the space left (criticality).
- Too much information for a pilot in combination with visible/audible.
- When using all warning system extra training is recommended to process everything correctly.
- I think my performance improved solely due to experience with the simulator and the scenarios.
- Audible warnings are best for VFR pilots together with minimal visual information, and a stick shaker for stall.

6.5. DISCUSSION

Five hypotheses were formulated, all related to the comparison between the conditions with (HF) and without (NH) haptic feedback. First, risk homeostasis was expected to be present in the form of improving performance metrics while objective safety metrics decreased. Results showed no difference between the haptic enabled/disabled conditions for both metrics, hence this hypothesis is not supported by the data.

Second, workload was expected to decrease which did not show in both the RSME or the debriefing questionnaire, again not supporting the hypothesis.

Third, enabling haptic feedback should improve situation awareness which is not supported by the two situation awareness-related questions proposed to the pilots after every run. Pilots did indicate in the debriefing, however, that the haptic feedback increased their knowledge of the flight envelope, and that it helps in preventing and resolving critical situations. So even though this hypothesis is not supported by the data, there are subjective indicators that haptic feedback improved pilot situation awareness.

Fourth, although handling qualities as observed with a Modified Cooper-Harper rating were expected to improve, the results obtained do not support this hypothesis.

Finally, Van Der Laan-questionnaires indicate an overall positive acceptance of the system both with and without the haptic feedback. When enabling haptic feedback, the acceptance rating when averaged over all participants improves as expected in the hypothesis, hence supporting this last hypothesis. In summary, all hypotheses except the last one are not supported by the data.

Discussions with the participants showed that the majority believes that a haptic feedback system in the current form could be a useful tool in combination with a fly-by-wire flight deck. This is supported by the answers to the debriefing questionnaire which indicate that the majority of the participating pilots prefers the system with haptic feedback, irrespective of whether haptic feedback was introduced in block one or two. Furthermore, pilots indicated that the system is not distracting, matches other (visual/aural) interfaces, and improves knowledge and prevention of critical situations. Summarizing, pilots expect aviation to be safer when using this haptic feedback system.

A reason for the non-effect on the hypotheses can be found in the design of the experiment: participants are divided in two groups where one group received haptic feedback for the first block of four runs (HFG) and the other groups received haptic feedback in the second block (HSG). The intended goal of this division was to perform a balanced comparison where the order effect due to unfamiliarity and training is eliminated. Nevertheless, the results of the individual runs already show that although all participants received a basic explanation of the interfaces (visual/haptic/aural) and a basic training with the model, the subsequent task and procedure used in the measurement runs was of such difficulty that all participants showed a major learning effect.

This learning effect is also confirmed by some of the comments made by the participants after the experiment: they indicate that it was a lot to process from the start. As a result, a large order effect is present which cannot be compensated by the balanced experiment design, hence making a comparison of the per block averaged metrics useless. Additionally, the flight path presented on the outside visual is a challenging profile and is not a realistic profile which can be flow in the real aircraft with passengers (within six minutes the load factor varied between 0.45g and 1.6g). The goal of this profile is to ensure that the initial condition for the windshear is always equal, which was achieved, yet probably induced more artefacts because of a training effect.

Despite this unexpected learning effect among all participants, the resulting metrics per run do show an interesting behaviour: when supplying the participants with haptic feedback from the start, the learning effect appears to be stronger. This is visible in all performance and safety metrics, as well as the subjective situation awareness (specifically the feeling of being in control), indicating that the haptic feedback might help in more quickly developing a proper strategy to handle the windshear recovery procedure. In addition, the first run without haptic feedback for the HFG does not show any regression, which indicates that there is no reliance on the haptic feedback for the developed strategy. Participants in the HSG, hence those who do not receive haptic feedback from the start, improve their performance over the first runs. Nevertheless, the spread of the metrics only matches the results of the HFG after one run with haptic feedback. This suggests that the haptic feedback can help to further improve an already developed strategy.

The final remarks are based on the analysis performed and observations made in this chapter. Further analysis should be performed to investigate the learning effect by fitting learning curves through the data of each participant and comparing their characteristic properties. [124] Furthermore, a re-design of the experiment setup is required to match a transfer-of-training experiment which involves a training phase (comparable to the first block in this experiment), followed by a test phase where the (haptic feedback) support is removed (comparable to the first run without haptic feedback for HFG), and a generalization run where the participants are performing a different, but comparable task, to investigate whether either a skill was learned or participants relied on a pure feedforward technique. [125]

In summary, the experiment did not show the expected change in metrics when enabling the haptic feedback. Nevertheless, an interesting effect of the haptic feedback on learning during training is found, which should be further investigated.

6.6. CONCLUSION

This chapter presents a further design of a haptic feedback system, which uses force feedback through the control device to inform pilots on the flight envelope limits and protections of modern fly-by-wire aircraft. The new design uses forcing functions which are *asymmetric in both time and amplitude*, to inform pilots of an approaching limit *as well* as of the direction of the corrective action.

The system is evaluated by 24 PPL/LAPL pilots who flew a flight trajectory shown on the outside visual and encountered a windshear during each run. To counter-balance the learning effect, the participants are divided in two groups: one group performed a block of four flights with haptic feedback, followed by a block of four flights without, the other groups vice versa. It was expected that the data would show that enabling haptic feedback during this windshear recovery allowed the pilots to have an increased performance while having a reduced (objective) safety margin, i.e., risk homeostasis, and that pilot acceptance and handling ratings of the system haptic feedback would increase when providing haptic feedback.

Results did not show those expected results: most metrics were unchanged when switching the haptic feedback, only the acceptance scale (Van Der Laan-questionnaire) improved slightly when enabling the haptic feedback. When looking in more detail to the individual runs, the cause of this can be found in the order effect: irrespective of which group, i.e., of whether haptic feedback is provided, the first four runs showed such a strong learning effect that this order effect renders the comparison invalid.

The results do show, however, that enabling the haptic feedback appears to *improve the learning rate* over the first runs, and that no after effects are present when removing the feedback. As such, besides the fact that most pilots were positive about the system and indicated that they expect it to improve safety, our experiment suggests a potential training benefit when using haptic feedback. This deserves to be further explored, for instance using a transfer-of-training experiment.

EVALUATING STIFFNESS AND POSITION GUIDANCE FOR FEEDBACK ON FLIGHT ENVELOPE PROTECTION

The previous chapter used asymmetric vibrations to cue the pilot on the flight envelope. The evaluation showed no improvement in metrics at the first emergency encounter, yet did show a potential training benefit. Therefore, a new haptic feedback concept was designed with the specific aim to guide the pilot when approaching a limit and provide support from the first time use. This chapter evaluates these haptic feedback designs with 36 active PPL/LAPL pilots who flew a challenging vertical profile and encountered a windshear in a fixed-base simulator. The pilots were divided in three groups who received either cueing, guidance, or no haptic feedback. It was expected that: (i) cueing haptic feedback results in best performance from the first run yet worse metrics when no feedback is provided. Comparing the results of the cueing and no-haptic feedback groups confirmed the previous results.Results showed that the guidance haptic feedback resulted in improved metrics at the first run, and the worsening of metrics when no longer provided.

This chapter is accepted for publication as:

Title	Evaluating Stiffness and Position Guidance for Feedback on Flight Envelope Protection
Conference	AIAA Modeling and Simulation Technologies Conference, Nashville (TN), 2021
Authors	D. Van Baelen, M. M. van Paassen, J. Ellerbroek, D. A. Abbink, and M. Mulder

7.1. INTRODUCTION

B ^{OTH} international aviation safety boards, such as the European Union Aviation Safety Agency EASA, and airline associations, for example the International Air Transport Association IATA, identify loss of control in flight as one of the key risk areas resulting in most fatalities within aviation. [1, 3] A safety issue contributing to such a loss of control is identified as the inadequate monitoring of the main flight parameters and automation modes. To ensure and improve current safety levels, these loss of control events should be prevented.

Improving the information presented to pilots is expected to help reducing the loss of control occurrences. This can be achieved by augmenting the visual displays on the flight deck with information on the limits of the aircraft, i.e., the flight envelope. Research showed that this can improve safety by reducing the risk of violations of those limits. [51] Once the limits are exceeded, for example in a stall, the information on the PFD can be augmented with recovery guidance which delivers recovery performance improvements as shown in three simulator evaluations. [118]

Apart from the visual channel, pilots can also perceive information through the sense of touch. An example is the haptic interface, which provides force feedback through the control device. This form of information can have a significant positive effect when a pilot is guided along the approach path. [41, 67] Additionally it can be used to show a set of predicted controllability limits, which was shown to be used by pilots in an experiment. [52] Research indicates also that haptic feedback can be used to show pilots information on the FEP. [126]

The latter experiment had two groups where the first group started with haptic feedback which was 'cueing' the pilot on the flight envelope limits, and the second group had no haptic assistance. After a break the groups switched: only the second group received haptic feedback. The initial hypothesis for this experiment was that haptic feedback would support performance, and that performance would reduce after reverting to a condition without haptic support. Contrary to this, however, it was found that haptic feedback mainly contributed to pilot learning, and performance persisted after haptic support was removed. In addition, haptic support did not improve performance during the first run, which indicates that when implemented on an aircraft, it might not provide pilots with support the very first time they encounter a new situation. As the haptic feedback system aimed to support pilots also in new, unforeseen circumstances, a new iteration of the haptic feedback is required.

Actively supporting the pilot has been found to help at the first encounter, yet is subject to reversion to base performance when the support is removed. In a skill acquisition task where a slider had to be moved left and right, four groups of participants received feedback on their performance in a training phase at different times: after each run, or an average score after every five, ten or fifteen runs. [127] Their results showed that increasing the amount of feedback increases performance. Immediately after the training phase, another set of measurements was performed where no feedback was provided. There, the group with the most amount of feedback in the training performed worst, although not significantly different from the other groups. Another measurement was performed two days after the initial training, which showed again a tendency for decreasing performance with increasing feedback during training. This phenomenon is called the "guidance hypothesis": a dependency on the feedback develops while learning the task; disabling this feedback then results in worse performance, due to required re-adaption. This phenomenon was also reported in a similar, vertical task. [128]

Within the field of haptic feedback, different applications have been recently designed to support the human operator in a task, and to provide support from the first encounter. Examples of this are a support for an abstract control task ([33]), a lane keeping assist in the automotive domain ([129, 130]), and an obstacle-avoidance system for UAV tele-operation. [32] These examples used active haptic feedback, for example an increased stiffness or actively moving control device, to guide the operator to complete the task. Transferring these active haptic feedback principles to the aircraft flight envelope protection system might provide a feedback system which supports pilots from the first run and solve the issue with our previous 'cueing' system. [126] Nevertheless, such implementations of haptic support have been found to be also hindered by to the guidance hypothesis described before, and it should be investigated whether this is also true in our particular application.

The aim of this chapter is to present a new haptic feedback for FEP design which is more actively 'guiding' the pilot, and to compare the results of this guidance haptic feedback system, as well as the existing 'cueing' haptic feedback system, to the results of a group of pilots who did not receive any haptic feedback at all. It is hypothesised that the group without haptic support required more time to learn the task when compared to the results of the 'cueing' group, and that the guidance haptic feedback design is able to support pilots from the very first run, however, with possible reversion in performance when the haptic assistance is removed.

This chapter first discusses the different haptic designs used in Section 7.2. Section 7.3 presents the experiment where the participants were required to operate an aircraft at the limits. In Section 7.4 and 7.5, results of the experiment are described and discussed. Finally, the conclusions are shown in Section 7.6.

7.2. FEEDBACK DESIGN

The haptic feedback design is based on a control structure similar to an Airbus A320. Full details are given in our earlier work, see Ref. [131], only the relevant elements for understanding the current experiment will be explained in this section. Two designs are elaborated which use haptic feedback to communicate the flight envelope protection limits by changing the feel on the control device.

Note that the designs shown here do not include a breakout force, i.e., a minimal force required to move the side stick, which is present on an actual A320 aircraft. The two haptic feedback designs to communicate the flight envelope limits are discussed below, respectively a cueing and guidance haptic support system. But first some basic knowledge on the A320 control structure is presented.

7.2.1. AIRBUS A320 CONTROL STRUCTURE

Modern-day Airbus aircraft, like the A320 and the A330, all employ a FBW system. This means that there is no mechanical connection between the control surfaces and the control device. The latter acts as an interface for the pilot to provide inputs to the FCCs

which then command the control surfaces with hydraulic actuators. This allows a Flight Envelope Protection (FEP) system to be used, which can check and, if necessary, limit pilot inputs, to ensure that no flight envelope limits are violated.

Longitudinal control in a FBW Airbus, with all sensors functional (a mode designated as the normal law control law), is provided using C^* -control, which is a combination of both pitch rate (*q*) and load factor (*n*). [10–13] On top of this control law, a hard envelope limit is employed which protects the pilot from exceeding limits on angle of attack (*a*), load factor (*n*), and maximum velocity (V_{MO}). This protection is depicted in Fig. 7.1, where the nominal flight envelope is the extreme limit which can not be exceeded, the safe flight envelope is the point where protections start acting. The envelope is constructed by the maximum (n_{max}) and minimum (n_{min}) load factor, their protection limits ($n_{max_{prot}}$ and $n_{min_{prot}}$, respectively), the maximum operation velocity (V_{AO} , and protection $V_{AO_{prot}}$), and minimum velocity ($V_{a_{max}}$, and protection $V_{a_{prot}}$).



Figure 7.1: Flight envelope, velocity (V) versus load factor (n)

When multiple FCCs fail, or when a sensor failure occurs, the control is reverted to a degraded control law. In this research, we will consider a control law close to the Airbus alternate law without reduced protections, where the same protections apply as before, only the angle of attack protection is lost. Hence, in alternate law the aircraft can be stalled, and it allows the pilot to give more extreme control actions.

Lateral control in normal law is a bank (ϕ) rate command from -33° till $+33^{\circ}$ of bank. Beyond these limits, positive roll stability is achieved such that the aircraft rolls back to the protection value (ϕ_{prot}) of $\pm 33^{\circ}$. The maximum achievable bank, with full lateral side stick deflection is $\pm 66^{\circ}$ of bank. In alternate law, lateral control reduces to a pure rate command, irrespective of the actual bank angle. More details on the control laws and degraded control laws can be found in Ref. [131].

Given that for both longitudinal and lateral control, a degradation of the control law results in a different effect for a given control input, a clear indication of both the limits and the active protections of the flight envelope is required. Nevertheless, accidents have occurred where pilots were not aware of what control law was active, and what protections were still active. [7] As such, a clear and intuitive way of presenting this information can be found in haptic feedback and a new design is proposed in the following.

7.2.2. CUEING HAPTIC FEEDBACK DESIGN

In the cueing haptic feedback design, the pilot is cued about the flight envelope limits using forcing functions (forces on the control stick) which are asymmetric in both time *and* amplitude. To visualize the feel, the amount of force required to displace the side stick to a certain deflection is combined in the haptic profile as given in Fig. 7.2. This figure shows the nominal feel on an Airbus side stick with a neutral point, the point at which no force on the side stick is required, and a linearly increasing force with a certain spring coefficient with an increased stiffness at 6°. [61] Such a haptic profile provides the pilot with information on the input magnitude: larger inputs require larger forces.

Previous research showed that such an asymmetric vibration can be used to both cue the pilot about an imminent limit, as well as indicate a required control action to move away from that limit. [132] Such a forcing function is vertically shifting the default haptic profile (Fig. 7.2). It is assumed that the forcing functions are short in time and/or amplitude such that the input to the aircraft is minimal. The feedback design uses three cues to communicate the flight envelope limits to the pilot:

- 1. When the aircraft state leaves the safe flight envelope, i.e., crosses the red line on Fig. 7.1: a sawtooth-shaped forcing function of 1s with an amplitude of 0.282Nm and frequency of 2Hz is activated.
- 2. As long at the aircraft state remains outside the safe flight envelope: one sawtoothshaped 'tick' is provided every second, where the intensity of the tick is linearly increasing with the magnitude of the safe flight envelope excursion, up to a maximum of twice the default magnitude.
- 3. When the velocity drops below $(V_{\alpha_{max}} + V_{\alpha_{prot}})/2$, i.e., left of the green line on Fig. 7.1, a stick shaker signal defined by a sinusoid with amplitude of 0.426Nm and frequency of 20Hz is activated.



Figure 7.2: Default haptic profile in the cueing design, i.e., force required on the side stick for a given deflection

The *direction* of the sawtooth-shaped forcing functions is used to suggest a control direction to move away from the limit. As such, the cue is forward/push for high angles of attack and high load factors, the direction is opposite for other conditions. More details and an example can be found in Chapter 6.

7.2.3. GUIDANCE HAPTIC FEEDBACK DESIGN

The guidance haptic feedback design informs the pilot on the limits of the flight envelope using two changes to the haptic profile: i) a change in spring coefficient and ii) a displacement of the neutral point position. To guarantee that the pilot has the final authority of the side stick, the maximum amount of force required to displace the stick to the maximum position is limited to 11.6Nm which results in a default haptic profile for the guidance design as shown on Fig. 7.3a. This maximum value is chosen based on the forces exerted by pilots on the stick in the experiment discussed in the previous chapter. A summary of all tuning parameters can be found in Table 7.1. The two cues introduced above are elaborated next.

Table 7.1: Summary of all haptic feedback tuning parameters

Property	Value
$\delta_{ m max}$	16°
$\tau_{\rm overspeed}$	5s
Δt_{α}	3s
F _{max}	11.6Nm



(a) Default feeling in guidance design



F [Nm]

20

10

0

·20 _____

-10



(c) Guidance design with increased

stiffness (severity 0.5)



0 10 20

 δ [deg]



0 10 20

 δ [deg]



STIFFNESS FEEDBACK

Increased manipulator stiffness has been investigated in previous research for indicating an undesired control deflection when a pilot-induced oscillation is imminent ([46, 47]), signaling a lagging adaptive controller ([75]), or indicating a limit on the main rotor setting of a helicopter. [74] In our scenario, an undesired control deflection is defined as an input which brings the aircraft closer to the limits of the flight envelope, which can be e.g., a control deflection in one specific direction. As such, our haptic feedback system will increase the spring coefficient in the direction of the unwanted deflection, leaving the other direction unchanged as shown on Fig. 7.3c.

The amount of stiffness change is determined by the magnitude of the safe flight envelope excursion, similarly to the amount of stiffness change in previous research to indicate a criticality. [75] Starting at the edge of the safe flight envelope until the flight envelope limit (respectively, the red-dashed line and black line on Fig. 7.1), the stiffness is gradually increased. Using a generic symbol v for the different limits of the flight envelope (maximum velocity, max/minimum load factor, maximum angle of attack), the default stiffness of the unwanted direction is multiplied with a factor K_k , determined by the gain K_v and the severity of the violation:

$$K_{k} = \begin{cases} 1 & \text{if } v < v_{prot} \\ 1 + K_{v} & \text{if } v > v_{nom} \\ 1 + K_{v} \frac{v - v_{prot}}{v_{nom} - v_{prot}} & \text{else} \end{cases}$$
(7.1)

The severity is defined as the ratio of the violation of the safe flight envelope, $v - v_{prot}$, where v_{prot} is the value at the edge of the safe flight envelope, and the distance between the safe and nominal flight envelope, $v_{nom} - v_{prot}$, where v_{nom} is the value at the edge of the nominal flight envelope. To illustrate this, the haptic profile with a stiffness change for a severity of 0.5 is shown on Fig. 7.3c. Increasing the severity to 1 results in a haptic profile shown on Fig. 7.3d which requires even more force for a backwards stick deflection. In this experiment, K_v is set to 2 for all limits.

NEUTRAL POINT FEEDBACK

A shift in the neutral point can be used to indicate a required deflection to follow a certain flight path ([41]) or, in automotive applications, to follow the road ahead. [133] If a positive/push deflection is required, this would result in a haptic profile as shown Fig. 7.3b. In our scenario, the aircraft is nearing its limit and the required deflection to return to the safe flight envelope can be indicated through the side stick. Since the aircraft dynamics at the different edges of the flight envelope are not equal (i.e., high velocity, angle of attack, and load factor), for each of these limits a required side stick deflection is determined as follows:

Velocity protection ($V > V_{prot}$) When an overspeed occurs, the speed has to be reduced actively by the pilot by either reducing the throttle, or by pitching up such that kinetic energy is rapidly exchanged for potential energy. The Airbus control law will implement a forced nose-up command (see Subsection 7.2.1), which could be translated to a change in neutral point. Nevertheless, the actual implementation of this signal is not known for

this research and is approximated as described below. The main reason for this cue is to inform the pilot that maintaining the stick at zero deflection does *not* solve the flight envelope violation, and action needs to be taken. Note that here our research deviates from the A320 FEP: the nose-up command is not activated when crossing V_{MO} , it is already activated when crossing $V_{MO_{prot}}$.

For this research, the nose-up command, and therefore the magnitude of the neutral point shift, is governed by the change in load factor required to bring the positive acceleration to zero. It is determined by starting from the longitudinal equations of motion ([69]), where we assume engine thrust to be parallel to the aircraft longitudinal body axis:

$$T\cos(\alpha) - D - W\sin(\gamma) = m\frac{dV}{dt}$$
(7.2)

The pilot can manipulate the aircraft flight path (γ), through moving the stick. Here, the neutral point is shifted to obtain a flight path angle such that there is no positive acceleration, $\frac{dV}{dt} = 0$. If the aircraft is accelerating before the activation of the neutral point shift, the left part of Equation 7.2 is not zero and can be rewritten to obtain a steady flight path:

$$\gamma_{\text{steady}} = \arcsin\left(\frac{T\cos\left(\alpha\right) - D}{W}\right)$$
(7.3)

Thrust and drag cannot be measured directly, their effects can be measured through accelerometers, mounted on the aircraft body, which therefore must first be rotated to the velocity reference frame:

$$T\cos(\alpha) - D = ma_{x_a} + W\sin(\gamma)$$

= $m(a_{x_b}\cos(\beta)\cos(\alpha) + a_{y_b}\sin(\beta) + a_{z_b}\cos(\beta)\sin(\alpha)) + W\sin(\gamma)$ (7.4)

Combining Equation 7.3 with Equation 7.4 then yields the required change in flight path angle for zero acceleration ($\gamma_{\text{steady}} - \gamma$), all expressed in measurable quantities.

As discussed above, the side stick gives load factor commands for high velocities and therefore also a relation between the change in flight path angle and load factor is required. By assuming that the steady state pitch rate is predominantly determined by a change in flight path, the required load factor can be expressed by the required flight path angle and a tuning factor ($\tau_{overspeed}$) which is an indication of the recovery speed: [73]¹

$$n_{\text{req}} = \frac{V}{g} \cdot q + 1 \approx \frac{V}{g} \cdot \dot{\gamma} + 1 = \frac{V}{g} \cdot \frac{\gamma_{\text{steady}} - \gamma}{\tau_{\text{overspeed}}} + 1$$
(7.5)

Angle of attack protection ($\alpha > \alpha_{prot}$) When the angle of attack is above the maximum value, the required change to bring it back to the protection value should be translated to the side stick. The required change in load factor can be obtained by starting from the effect of pitch rate on load factor:

$$n = V \cdot q \tag{7.6}$$

¹The conference publication contained a mistake in the equation derivation and is corrected for this thesis.

Furthermore, the required pitch rate can be approximated by a required change in angle of attack over a certain time, assuming that for short periods of time the change in pitch is dominated by a change in angle of attack. As a desired angle of attack is available (α_{prot}), and by choosing a time, the required change in load factor is determined by:

$$n = V \cdot q \approx V \cdot \frac{\alpha - \alpha_{\text{prot}}}{\Delta t_{\alpha}}$$
(7.7)

This results in one tuning parameter (Δt_{α}) which can be used to indicate how responsive the side stick will move for a given required change in angle of attack. In the current setup, this tuning parameter is set to 3s.

Load factor protection ($n > n_{\text{prot}, \text{ pos}}$ or $n < n_{\text{prot}, \text{ neg}}$) When a load factor outside the safe flight envelope occurs, a required change in control inputs can readily be obtained since side stick inputs are proportional to a change in load factor. The required load factor in case of positive load factors is $n_{\text{prot}, \text{ pos}} = 2.0$ g, in case of negative load factors $n_{\text{prot}, \text{ neg}} = -0.5$ g, resulting in a required stick deflection:

$$\delta_{n} = \begin{cases} \left(n - n_{\text{prot, pos}}\right) \cdot \frac{n_{max, pos}}{\delta_{\max}} & \text{if } n > n_{\text{prot, pos}} \\ \left(n_{\text{prot, neg}} - n\right) \cdot \frac{n_{max, neg}}{\delta_{\max}} & \text{if } n < n_{\text{prot, neg}} \\ 0 & \text{else} \end{cases}$$
(7.8)

When this haptic feedback system is implemented, it presents the pilot with continuous feedback which uses the stiffness to indicate an undesired deflection, and a shift in neutral point to show the required deflection to return to the safe flight envelope. The stiffness change and neutral point shift can occur simultaneously, for example in Fig. 7.3e where a positive neutral shift is combined with an increased stiffness for backwards deflections. The combination of these two cues might result in unacceptable high forces required to move the side stick. This is prevented with the implementation of the maximum force, resulting in a flat slope on the haptic profile. The remainder of this chapter discusses the results of an experiment to evaluate both cueing methods.

7.3. METHOD

To evaluate the haptic interface designs, an experiment was performed which uses the same setup as used in a previous experiment which investigated the 'cueing haptic feedback', see Chapter 6.

7.3.1. INDEPENDENT VARIABLES

The experiment had a between-participants design, with one independent variable. The participants were divided in three groups: the cueing group, the guidance group, and the manual (no-haptics) group. Each group (12 participants per group) performed two blocks of four runs each, elaborated below, and summarized in Table 7.2.

In the first block, participants were presented with one of the three haptic support conditions. Literature found that an increasing amount of feedback in this initial stage, results in worse performance when that feedback is removed: the "guidance hypothesis". [134] To investigate the consequences of removing the feedback in our application, all participants performed a second block in the manual, no-haptics condition.

The results of the cueing group (12 participants) were obtained from our previous experiment in Chapter 6 which had the exact same experimental setup. Twenty four new participants were invited and numbered in sequence of experiment participation. Evennumbered participants were placed in the guidance group group, all odd-numbered participants are part of the manual group. A total of 36 participants results from combining the previous and present experiment groups.

Table 7.2: Experimental design



7.3.2. PARTICIPANTS AND INSTRUCTIONS

For this experiment, data from 36 pilots (1 female, 35 male) with a current PPL or LAPL license were used. As these pilots are not Airbus pilots, they were reminded that the aircraft model used has a mass of 64,000kg and had to be handled with more care than a general aviation aircraft. The experience of the three different groups can be found in Tables 7.3, 7.4 and 7.5. A visual comparison of the flight hours per group is shown Fig. 7.4. A Kruskal-Wallis rank sum test did not show statistical significant differences in experience between groups ($\chi^2 = 3.17$, p > 0.2).



Figure 7.4: Flight hours per group

Participant	Age	Flight hours	License	
M1	65	400	PPL	
M2	52	1,500	CPL / IR / FI	
M3	66	1,860	PPL / IR	
M4	20	150	PPL	
M5	62	430	PPL	
M6	57	180	LAPL	
M7	49	420	CPL	
M8	20	82	PPL	
M9	62	175	PPL	
M10	50	200	PPL	
M11	20	55	PPL	
M12	23	65	PPL	
Mean	45.5	459.8	-	
Std.Dev.	19.1	590.4	-	
FI Fli	FI Flight Instructor			

Table 7.3: Participants in the manual group

IR Instrument Rating

Table 7.4: Participants in the cueing group

Participant	Age	Flight hours	License
T1	34	116	PPL
T2	25	80	LAPL
T3	49	205	PPL / E-IR
T4	40	630	PPL
T5	48	350	PPL
T6	24	500	PPL
Τ7	66	900	PPL
T8	46	250	LAPL
Т9	48	500	PPL
T10	50	400	PPL
T11	33	240	PPL
T12	53	600	PPL / E-IR
Mean	43	397.6	-
Std.Dev.	12.3	240.2	-

E-IR Enroute-Instrument Rating

Participants were instructed to always remain within the nominal limits of the flight envelope (black line on Fig. 7.1) which were shown on the PFD using the red indications proposed in Ref. [119]. Additionally, it was mentioned that a simulation run would stop when the aicraft reached an altitude of 50ft above ground level, irrespective of any other event/performance.

Table 7.5: Participants in the guidance group

Participant	Age	Flight hours	License
G1	67	475	PPL
G2	57	300	PPL / IR
G3	26	100	PPL
G4	30	78	PPL
G5	44	170	PPL
G6	50	80	PPL
G7	43	150	PPL
G8	47	500	CPL
G9	71	300	PPL
G10	52	250	PPL
G11	50	200	PPL
G12	60	200	PPL
Mean	49.8	233.6	-
Std.Dev.	13.3	140.5	-
IR Instrument Rating			

7.3.3. EXPERIMENTAL SETUP

The experiment was performed in the HMI research simulator of Delft University of Technology. It is a fixed-base simulator, with a near 180° outside field-of-view, used in the first officer position of which an inside-view is shown in Fig. 7.5. Since the pilot was sitting in the first officer position, the display to his front-left was the ND showing a top-down overview of the situation, shown in Fig. 7.6a, combined with a basic engine N1-indication and slats/flaps indication. The display right in front of the pilot was the PFD showing the critical flight states, shown in Fig. 7.6b, which included display indications used to show *why* and *when* the haptic feedback is active. [119] Next to the visual information, auditory warnings were presented when the aircraft angle of attack was above the maximum value, and when the velocity was above the maximum velocity.

A custom-made, hydraulically driven side stick with programmable dynamic properties is located at the right-hand side and was configured to Airbus side stick properties. [61] To the left, a throttle quadrant is present which was used to control the throttle and high lift device settings. Centrally placed, a Boeing 737 Mode Control Panel (Airbus terminology: FCU) enabled the interface with the heading, velocity and altitude references on the displays. Outside visuals were generated using FlightGear¹ and showed the airport infrastructure, terrain and important buildings at the airport. A proprietary A320like flight dynamics model, including control laws from the German Aerospace Center (DLR), was used as the simulated aircraft. [120]

The nominal, no-haptics control device settings for this experiment, including mass (m), spring coefficient (k), damping coefficient (b) and maximum deflection (δ_{max}) , for both longitudinal and lateral side stick axes are given in Table 7.6.

¹Open source flight simulator available at http://flightgear.org



Figure 7.5: Inside view of the HMI flight deck



(a) ND

(b) PFD

The 'cueing haptic feedback' used: (i) a sawtooth shape of intensity 0.282Nm, duration 1s and frequency 2Hz when exiting the safe flight envelope, (ii) a sawtooth shape with varying intensity proportional to the relative distance of the protection and flight envelope limit, duration 0.5s and frequency 2Hz when remaining outside the safe flight envelope and (iii) a stick shaker for low velocities as a sinusoid with frequency of 20Hz and magnitude 0.426Nm. Further details can be found in Chapter 6.

'Guidance haptic feedback' used an increase in spring coefficient to maximal twice the nominal stiffness, and neutral point shifts to provide recommended side stick deflections as discussed in the design section.

Table 7.6: Control device in the experiment

Property	Value
m	$0.2 \mathrm{kg}\mathrm{m}^2$
$k_{lon_{nom}}$	36.3 Nm/rad
b_{lon}	0.4 Nm s/rad
$\delta_{lon_{max}}$	0.279 rad
$k_{lat_{nom}}$	21.8 Nm/rad
b_{lat}	0.4 Nm s/rad
$\delta_{lat_{max}}$	0.314 rad

Figure 7.6: Flight deck display setup used in the experiment

7.3.4. EXPERIMENT SCENARIOS

The haptic feedback system was designed to communicate the 'proximity of the flight envelope limits' to the pilot and therefore required an evaluation at these limits. In analogy to our previous experiment, the scenarios presented a stringent flight path, as discussed below, followed by the emergency scenario encountered during each flight.

FLIGHT PATH

Each run was started when the aircraft was flying 140kts (72.0mps) at 2,500ft (762m) with slats and flaps set for approach (Airbus setting 3), overhead the threshold of runway 23 of Zoersel (Belgium) and aligned with the respective runway. This location was chosen as it has no special terrain features close-by, and the runway was not visible from the starting position as illustrated on Fig. 7.8a. Additionally, the auto-throttle was set to 140kts and activated, reducing the variability of the initial aircraft state when the event was triggered and should provide more consistent results. From this position, pilots were presented with visual markers (squares of 60m by 60m) on the outside visual display to help them fly a flight profile consisting of six 'hills', for which an example path is presented in Fig. 7.7 and visualized on the outside visual as shown in Fig. 7.8.



Figure 7.7: Flight path side-view, solid black vertical lines indicate "fly-through gates" shown on the outside visual; the thick red line indicates the trigger point of the windshear (not shown on the outside visual); the dotted blue lines indicate the windshear section used in our evaluation





(a) View at start of run

(b) Perspective view on flight path (viewing angle is for illustrative purpose only, never encountered during flight)

Figure 7.8: Example of the outside visual flight path visualization

A hill was 2.27NM (4,200m) long and had one of three possible amplitudes: 150ft (45.72m), 300ft (91.44m) or 500ft (152.4m). Combining six hills yielded a saw-tooth trajectory. The flight path started with a horizontal segment of 0.41NM (750m) and one hill of the smallest amplitude as run-in. This was followed by a randomized order of hills such that each amplitude of hill occurred twice in the flight path. Each flight ended with a horizontal segment of 0.54NM (1,000m) as run-out. This setup of hills was chosen as it was expected that it allowed the results to be evaluated for each hill separately. Eight different realizations of the randomization were obtained to present pilots with variability in the scenarios. The resulting trajectories are all shown in Section C.2.

EMERGENCY SCENARIO

As the pilots of our initial experiments did express the potential added value of the haptic feedback system in a windshear event, this event is re-used for this research. [110] A wind shear is a meteorological phenomenon where wind velocities are locally rapidly changing, and can be caused by a large cylinder of air suddenly "droping" towards the earth. During such a wind shear event, downdrafts can push the aircraft dangerously close to the ground. [76] To recover from this event, the pilot has to move as close to the stall limit as possible to prevent further height loss and maximise aircraft performance. [13]

The windshear in each run was *always* started when the aircraft moved through the visual marker of the windshear hill at an altitude of 2900ft (883.92m). Each flight path contained two hills with the largest amplitude, only one of them was selected at random to contain the windshear trigger point. The windshear itself was modeled by both a head-on and top-down component as shown in Fig. 7.9. [76] Once the windshear was initiated, the visual and aural warning trigger, and the pilot had to apply the windshear recovery procedure as stipulated in Section B.6, which was based on the Airbus Flight Crew Operating Manual. [13]



(a) Headwind component

(b) Downwind component

Figure 7.9: Windshear component distribution

When providing only windshear as the emergency scenario during each run, pilots might anticipate this event, even in the first run. To prevent this, two more checklists for an emergency were presented to the pilots beforehand: the actions required for a single engine stall (Section B.7), and for a sudden center of gravity shift (Section B.8). Therefore, pilots were expecting one of these three emergency scenarios, but were unaware of what scenario was actually triggered. Note that the checklists presented in Fig. B.6 and B.7 are heavily modified from the FCOM, and the checklist for the sudden center of gravity shift is non-existing in the FCOM.



Figure 7.10: Traffic pattern flown to runway 36L at Schiphol (Schiphol layout from AIP [84])

7.3.5. EXPERIMENT DESIGN

To allow pilots to become sufficiently familiar with the simulator and the haptics (if applicable), a familiarization phase was performed, followed by measurement runs.

FAMILIARIZATION

After a briefing on the simulator safety procedures, all pilots were explained the controls and displays by presenting the flight envelope (an image similar to Fig. 7.1), and the PFD (Fig. 7.6b) to the pilot. In this setup, no aircraft model was used, yet the flight envelope state was changed directly (hence changing the velocity and load factor) and all visual and auditory cues were elaborated. After that, pilots in the cueing or guidance groups *felt* the design rationale behind the haptic feedback design using an image similar to Fig. 7.3.

Then the aircraft model was introduced to the pilot by flying a traffic pattern twice to a final approach at Schiphol (EHAM) as shown on Fig. 7.10, without the haptic feedback, hence focusing on familiarization with the model. Pilots were instructed to follow the instructions as indicated. Some pilots encountered a stall and/or an overspeed condition during these first runs. If the pilots did not hit one or both limits, they were asked to deliberately explore those boundaries to ensure that all pilots encountered them before the measurement runs.

MEASUREMENTS

The measurement phase contained eight realizations of the flight path presented above. They were flown in a randomized fashion, distributed over all participants using a Latinsquare distribution. Each group performed two blocks of four runs, with a break in between, with haptic feedback as shown in Table 7.2.



Figure 7.11: Time trace of velocity with safety metrics indicated

After each run, pilots were asked to indicate their workload using a RSME rating [85], and complete a post-run situation awareness questionnaire, to indicate how helpful the visual, auditory and haptic (if supplied) elements are. They also provided a misery scale rating to measure and account for possible effects of motion sickness. [121] Once this was completed, pilots were informed on how much time they spent inside the flight envelope, which they had to maximise.

After each block of four runs, pilots were asked to complete a questionnaire with a modified Cooper-Harper rating scale ([122]), and a Van der Laan-rating scale. [104] After the experiment was completed, pilots were asked to complete a post-experiment questionnaire, which contained a number of questions with Likert-scales on how they experienced the haptic feedback system.

7.3.6. DEPENDENT MEASURES

The dependent measures are split into objective and subjective measures.

OBJECTIVE MEASURES

The objective measures are retrieved from the windshear recovery procedure, and focus on performance and safety. To illustrate why these metrics were chosen, a time excerpt of a windshear recovery is shown in Fig. 7.11. Another example is further elaborated in the results section, for now it is sufficient to understand that this shows a participant aiming for the best performance of the aircraft, flying close to the maximum angle of attack.

Looking at the example, one can argue that a safe flight is performed when the aircraft state is within the flight envelope limits, indicated with the solid black line representing α_{max} . Although participants are instructed to stay within the limits at all times, at certain moments in time the pilot could control the airplane beyond these limits. A first performance metric was therefore the *time spent outside the angle of attack limits*.

Participants can push the aircraft by flying very close to its limits, even above the limits, or they can choose to remain well away from the limits. A straightforward metric to determine this safety definition was the *maximum angle of attack obtained relative to the flight envelope limit*: it can indicate how close to the limits the participant dares to control the airplane.

Time by itself only informs about the length of the limit violations, it does not take into account the closest distance: two different limit violations might be of equal time, yet one just slightly over the limit while another one is in a deep stall. As such, a safety metric combining both the time and the magnitude of the violation was the *integral of the angle of attack over the flight envelope limit*.

One additional performance metric on the overall windshear recovery procedure was used: the *total amount of altitude lost during the recovery*. Although not communicated to the participants, the maximum altitude lost from the windshear initiation to the end of the windshear recovery is considered here as an indication of how much of the available aircraft performance is utilized by the participants. Best performance is achieved when this amount of altitude lost is minimum.

Previous research showed that the level of risk humans experience is mostly kept the same when support increases, i.e., risk homeostasis. [86] This was found in an automotive studied where supplying haptic feedback resulted in participants driving at higher velocities. [87] For the current experiment, risk homeostasis was also expected, and can be defined by improved performance, combined with objective safety metrics closer to the maximum value, as pilots obtain a better awareness of the risk involved when supplied with haptic feedback.

SUBJECTIVE MEASURES

Subjective measures are obtained by asking the pilot for an opinion, or experience. The categories and measures are:

- Workload: after each run, the pilot was asked to provide a RSME rating [85]
- Situation awareness questions: after each run, the pilot was asked to answer two questions on a linear scale (0–100) ranging 'Never' left (0), and 'Always' right (100):
 - 1. Did you have the feeling you were in control of the situation?
 - 2. Did you have the feeling you missed critical information?
- Usefulness:
 - 1. Pilots were asked after each run to rate the usefulness of all display and haptic elements on a Likert scale
 - 2. After each block, pilots were asked to provide a modified Cooper-Harper rating
 - 3. After each block, pilots were asked to fill a Van Der Laan-questionnaire
- Pilot experience: after the experiment, the pilot was asked to fill in a questionnaire regarding the experience with the haptic feedback system.

The questionnaire presented to the pilots after the experiment used five point Likertscales where all points are labeled. A different set of questions was presented to the participants in the manual, no-haptics group because they did not experience any haptic feedback at all. Both sets of questions can be found in Section A.8.

7.3.7. Hypotheses

In the evaluation of the experiment, the manual group served as a baseline to compare pilot behaviour during the windshear recovery. The expected behaviour of the other two groups is visually summarized in Fig. 7.12 and explained in the following.



Figure 7.12: Schematic representation of the expected results

We expected the *cueing group* to perform initially at the same performance level, yet have a faster learning rate over the first four runs, have an improved performance level at Run 4, and keep performance equal when no haptic feedback is provided in the final four runs, i.e., no after-effects. In terms of dependent measures, this means no change in performance/safety between the manual and cueing group at Run 1. At Run 4, the cueing group has an improved performance between-groups compared to the manual group, and within-group compared to Run 1. Comparing the metrics of Runs 4 and 8 within the cueing group, should give no differences to indicate no after-effects.

The *guidance group* was expected to have an improved performance from the first run as long as haptic feedback is provided, but when this haptic guidance is not provided, (Run 5), we expected the performance to suddenly worsen following the "guidance hypothesis". [134] In terms of dependent measures, this would translate to improved performance and safety margins at Run 1 when between-groups comparing the manual and guidance groups. At Run 4, the guidance group has an improved performance betweengroups compared to the manual group. After-effects were expected to show up when comparing performance and safety margins of Runs 4 and 5 within the guidance group.

We expected pilots to perceive the *cueing haptic feedback* as a useful source of information, yet the information still needs to be interpreted. Therefore, the subjective workload ratings at Run 1 of the cueing group were expected to not differ from the manual group, yet indicate this group to have an improved situation awareness. At Run 4, the workload of the cueing group is expected to be lower due to familiarization.

The *guidance haptic feedback* was expected to be supporting pilots from the first run, yet it can be less clear in the reason why it provides a cue. It was expected that the subjective workload ratings at Run 1 of the guidance group is lower compared to the manual group, yet deteriorate when no haptic feedback is supplied anymore (at Run 5). Subjective situation awareness ratings of the manual and guidance groups were expected to be similar.

For both haptic designs, we expected the Modified Cooper-Harper ratings and Van Der Laan-ratings to improve. The remainder of this chapter looks into the results of this experiment and discusses whether our hypotheses can be supported.

7.4. RESULTS

Before the metrics are discussed, Subsection 7.4.1 shows one example case where a pilot used the guidance haptic feedback system and two other noteworthy events which happened during the experiment. Next, the objective and subjective measures are presented in, respectively, Subsection 7.4.2 and 7.4.3. Answers to the debriefing questionnaires are presented in Subsection 7.4.4. For reference, all flown trajectories included in the analysis are shown in Section C.2.

When presenting data using box plots, medians are indicated using a horizontal thick line, outliers are indicated using plus-signs; all individual data points are presented next to the boxes using crosses. Furthermore, statistical analyses are performed in R ([123]) and results are only reported if p-values of 0.05 or lower are found. Results are compared both within- and between-groups: within-group the differences of Runs 1, 4 and 8 are examined, the between-group comparison investigates the difference between groups at Runs 1, 4 or 8. Tests are performed using a Kruskal-Wallis rank sum test which indicates whether there is a statistically significant difference. If a difference is found, a post-hoc test is performed using a pairwise Wilcox test where p-values are adjusted using the method proposed by Benjamini and Hochberg. [135]

7.4.1. TIME TRACE EXAMPLES

This section discusses three time traces from the experiment: the first shows a participant using the guidance haptic feedback, the second shows the only crash which occurred, and finally a design flaw of the guidance haptic feedback is illustrated.

EXAMPLE USE OF GUIDANCE HAPTIC FEEDBACK

This is an example where Guidance Participant 2 during his first run made use of the guidance haptic feedback system proposed in the design section of this chapter. Time traces for pitch angle, angle of attack and control device deflections can be found in, respectively, Fig. 7.13a, 7.13b, and 7.13c. Three frames are indicated for which the haptic profiles are given in Fig. 7.14. The haptic feedback can be seen in the haptic profiles, and on the control device deflection plot: the neutral point (δ_{np}) is the stick shift by the haptic feedback, the actual control device deflection (δ_{cd}) is the sum of haptic feedback and the human operator.

The windshear recovery procedure requires the pilot to use all of the available performance of the aircraft, which can be achieved by operating the aircraft near the maximum angle of attack. Initially, the pilot has to obtain a pitch angle of 17.5°, which is achieved at Frame 1 as can be seen on Fig. 7.13a. Here, the current state is still within the safe flight envelope and the corresponding haptic profile shows the nominal stick feeling on Fig. 7.14a. Next, Frame 2 shows the participant exerting back pressure on the stick to maintain pitch and to avoid the aircraft from descending, despite the haptic feedback indicating that a pitch down input is required to return to the safe flight envelope (shifted neutral point, Fig. 7.13c, and increased stiffness, Fig. 7.14b). Subsequently, the participant notices that a sustained back pressure brings the aircraft too close to the stall and starts following the haptic feedback cues to operate the aircraft near its limits. This can be seen by the matching of the neutral point and the actual control device position on Fig. 7.13c, for which one haptic profile is given in Fig. 7.14c.



(c) Control device deflection, i.e., input from the side stick to the FCCFigure 7.13: Time traces of Guidance Participant 2 using the haptic feedback during the recovery in Run 1



Figure 7.14: Haptic profiles for frames indicated in Fig. 7.13, cross indicates the current state

This example shows that this pilot used the haptic feedback, even in the first run where participants were expected not to be fully familiar with the aircraft model, task, or emergencies procedures. The metrics presented next can be used to further investigate whether this is a one-off example, or participants can indeed use the haptic feedback effectively from the first encounter.

ONLY CRASH OF THE EXPERIMENT

This example shows the first run of Partipant 7 in the manual group, where Fig. 7.16 shows the haptic profiles for the time frames shown on the time traces in Fig. 7.15. As this

-20

(a) Frame 1

20-10 0 10 20

 δ [deg]



187

(b) Frame 2 Figure 7.16: Haptic profiles for frames indicated in Fig. 7.15, cross indicates the current state

-20

20-10 0 10 20

 δ [deg]

-20

(c) Frame 3

20-10 0 10 20

 δ [deg]

was the first run, this was also the first time the participant encountered the flight path and windshear. After the windshear warning, the participant aimed for a pitch angle of 17.5°, as can be seen on Fig. 7.15a before Frame 1. At Frame 1, the participant notices that the aircraft is not climbing anymore (Fig. 7.15d), and as indicated in the checklist, increases nose-up input (Frame 1 at Fig. 7.15c, Fig. 7.16a).

After this, the windfield suddenly pushes the angle of attack above the maximum, as shown on Fig. 7.15b, and on the aural stall warning, the particpant starts applying *more* back pressure on the side stick. The input, shown on Fig. 7.15c, one snapshot at Frame 2 in Fig. 7.16b, shows a negative (pull) input of more then 50s during which sustained visual and aural stall warnings are provided, yet no haptic feedback as the participant is part of the manual group. About 60s after the windshear trigger, the participant retracts flaps which reduces in a reduced maximum angle of attack enlarging the problem.

Near the end of the flight, at Frame 3, the participant starts using a positive input and starts to solve the angle of attack excursion. Nevertheless, the action is too late and not sufficient altitude is left for the recovery. The flight is, as indicated in the briefing, stopped 50ft above ground level.

After this run, the participant filled the required questionnaires and he was told that he had a sustained stall, reducing his time inside the flight envelope. Additionally, he was reminded that one of the windshear recovery items indicates *not to change configuration*, i.e., do not change flaps. During the next run, the participant was able to recover from the windshear and complete the next flights. Although this crash is a one-off example, the only crash throughout the entire experiment campaign did occur when no haptic feedback of any form was present.

GUIDANCE HAPTIC FEEDBACK DESIGN ISSUE

The third example shown is the first run of Guidance Participant 5 with time traces in Fig. 7.17 and corresponding haptic profiles in Fig. 7.18. It is a show-case of a flaw in the current design of the guidance haptic feedback: a possible haptic 'lock-in' where eventually all haptic feedback is lost. The origin of the flaw can be traced back to the method used to guarantee that the pilot always has final authority over the automation on the side stick: with a changing neutral point and an increasing stiffness, a certain deflection of the side stick might require a force level which is not reasonable anymore to be achieved. Therefore, as mentioned in the design section, the maximum amount of force required to move the side stick is limited to 11.6Nm.

Looking at the angle of attack before the windshear trigger, indicated as negative times on Fig. 7.17a, at Frame 1 the state of the aircraft was already near the limits and the participant was informed as such using a shift in neutral point and increased stiffness for negative inputs as shown on Fig. 7.18a. At that moment, the participant was trying to follow the tunnel-in-the-sky presented on the outside visual and needed all available performance to do so as he was slightly below the tunnel. On the windshear trigger, the condition worsened and the haptic feedback provided a full stick forward input which is maintained throughout the time trace as can be seen on Fig. 7.17c.

Starting from Frame 2, the participant reached a backwards pressure of the limiting 11.6Nm, resulting in a flat haptic profile as in Fig. 7.18b. The participant maintained the backwards input, nevertheless, one level of force on the side stick is required for all negative deflections. As a result, the participant was inputting significant pitch-up



(c) Control device deflection, i.e., input from the side stick to the FCC

Figure 7.17: Time traces of the design flaw which occurred during Run 1 of Guidance Participant 5



Figure 7.18: Haptic profiles for frames indicated in Fig. 7.17, cross indicates the current state

commands for a significant time, with one snapshot in Frame 3 (Fig. 7.18c), possibly without noticing the magnitude of the input due to the lack of a force gradient: the participant was 'locked-in'. Additionally, in this situation, whatever the neutral point shift or stiffness changes, the participant is not able to perceive this feedback.

One possible solution to avoid such a haptic 'lock-in' is to not use a flat haptic profile, yet implement a very shallow slope. By using a minimal slope, the participant is able to distinguish between different magnitudes of input, and the neutral point shift by the haptic feedback can still be observed. In the runs after this, the participant was able to apply the recovery procedure without entering a haptical lock-in.

7.4.2. OBJECTIVE MEASURES

Objective measures are directly retrieved from the simulation data and are discussed in the following.

ALTITUDE LOST DURING RECOVERY

The amount of altitude lost during the windshear recovery is shown in Fig. 7.19.For the manual and cueing groups, a learning process is present: from Run 1 onward, performance increases and less altitude is lost during the recovery. Comparing Runs 1, 4 and 8 within one group shows a statistical significant difference for the manual group (p < 0.05, $\chi^2 = 6.87$) where the post-hoc indicates that only Run 1 and Run 8 contain a difference (p < 0.05). This indicates that the group improved performance from the start till end, yet at Run 4, the participants were not fully learned yet. For the cueing group, differences are present (p < 0.005, $\chi^2 = 12.87$) between Run 1 and Run 4 (p < 0.005), and between Run 1 and Run 8 (p < 0.005). This confirms that after Run 1, the participants quickly learned how to handle the windshear recovery, yet were not able to reach the final performance at the first run. Within the guidance group, no large differences are observed, which is confirmed by no significant differences between Runs 1, 4 and 5.



Figure 7.19: Altitude lost during windshear recovery

At first glance, the guidance group seems to have a lower median and lower spread compared to the other groups. At Run 4, the differences do not seem large. Nonetheless, no statistical significant results are found by the Kruskal-Wallis test. When no haptic feedback is supplied, performance of the guidance group seems to have a larger spread for worse performance. For Run 8, there is a 'near statistical significant' result of the Kruskal-Wallis test (p = 0.06, $\chi^2 = 5.64$), and the post-hoc test points to a difference between the guidance and cueing group (p = 0.09), as well as between the guidance and manual group (p = 0.09), indicating that the observation can be right, yet not supported by clear statistical significance.

TIME ABOVE MAXIMUM ANGLE OF ATTACK

The duration of the flight spent above the maximum angle of attack, Fig. 7.20 – a metric that was also communicated to the participants received after each run which they had to optimize – clearly shows the learning effect: starting from Run 1, this metric reduces, and thus participants also spent less time with the stall warning active. Within each group, this learning effect is confirmed with statistical tests: the manual group (p < 0.005, $\chi^2 = 12.60$) has significant differences between all runs (Run 1 and 4 p < 0.05, Run 1 and 8 p < 0.05, Run 4 and 8 p < 0.05) indicating that they improved over the course of the entire experiment. The cueing group has significant differences (p < 0.001, $\chi^2 = 23.50$) between Run 1 and 4 (p < 0.001), as well as between Run 1 and 8 (p < 0.001), indicating that they improved performance from Run 1 to 4, yet kept there performance level afterwards. The guidance group has only statistical differences (p < 0.05, $\chi^2 = 7.11$) between Runs 1 and 4 (p < 0.05) indicating an improvement over the first block of four runs, yet the lack of significant differences between Runs 1 and 5 might indicate the slight deterioration of the metric which is also visible on Fig. 7.20.



Figure 7.20: Time with angle of attack above limit during windshear (Manual Participant 7, 61.7s, not shown)

The time above the maximum angle of attack shows that at Run 1 the guidance group has a better performance compared to the two other groups. This is confirmed by a significant result of the Kruskal-Wallis test (p < 0.01, $\chi^2 = 9.92$), and post-hoc analysis indicates differences between the guidance and cueing group (p < 0.005), and 'near significant' difference between the guidance and manual group (p = 0.10). At Run 4, the boxplot of the manual group shows more participants still encounter a stall warning, yet this is not supported by statistical evidence. At Run 8, the guidance group appears to have three participants encountering a stall warning, compared to one in each of the other groups, but without statistical significance.

Another between-groups observations can be made regarding the spread of the data on the first block of four runs: each run of the manual group has the largest spread, and each of the guidance group has consistently the lowest spread. This might indicate that the guidance feedback is more stringent in its communication of the flight envelope limits, compared to the cueing feedback, and especially compared to the manual group.

HIGHEST ANGLE OF ATTACK

Fig. 7.21 shows how the largest angle of attack obtained throughout the recovery relates to the maximum angle of attack, i.e., it indicates how close pilots operate the aircraft near the limits, and if a maximum angle of attack violation is made, its magnitude. Again, a learning effect is present: the metric improves over the runs. Within the manual group, the plot shows that the learning effect seems to be rather slow, and this is confirmed by the statistical analysis: there are statistically significant difference (p < 0.01, $\chi^2 = 9.28$), more specifically, the post-hoc test reveals significant differences between Runs 1 and 8 (p < 0.05), and Runs 4 and 8 (p < 0.05), not between Runs 1 and 4.

Combining the lack of statistical difference between Run 1 and 4, and the large spread of the data indicates that participants in the manual group need more time to learn how to properly control the angle of attack. The values obtained for the cueing group show the worst results on the plot but improves over the four runs: the results have significant differences between runs (p < 0.001, $\chi^2 = 19.18$) and post-hoc indicates differences between Runs 1 and 4 (p < 0.001) and Runs 1 and 8 (p < 0.001), indicating that indeed participants made a significant difference in the first four runs and maintained this control strategy during the subsequent runs. For the guidance group, the Kruskal-Wallis test indicates 'near statistical' differences between Runs 1 and 4 (p = 0.076, $\chi^2 = 5.15$) between runs, yet the posthoc test only indicates 'near statistical' differences between Runs 1 and 4 (p = 0.06) and Runs 1 and 8 (p = 0.07). The plot does show a decrease in median during the first four runs, and an increase in spread when no haptic feedback is provided anymore.



Figure 7.21: Highest angle of attack obtained during the windshear, relative to the maximum angle of attack (positive values result in a stall warning)

At Run 1, the guidance group seems to have the lowest, yet still positive, safety margin compared to both other groups. This is confirmed partly as significant differences between groups (p < 0.01, $\chi^2 = 9.61$) are present, yet only between the cueing and guidance groups (p < 0.005) indicating that the participants in the guidance group have consistently lower maximum angle of attack violations compared to the cueing group. No significant difference is found between the manual and guidance groups probably due to the spread of the data. One key difference between groups which is not captured by the

statistical test is where the median is located with respect to zero, in other words, whether a median angle of attack above the maximum is achieved. At Run 1, the guidance group clearly has the median closest to zero. Furthermore, it takes the manual group until Run 6 to achieve a median below the maximum value, whereas both the cueing and guidance group achieve this at Run 3.

INTEGRAL ABOVE MAXIMUM ANGLE OF ATTACK

Combining time and distance above the maximum angle of attack results in the integral as shown in Fig. 7.22. The figure shows that from Run 1, the guidance group has the least amount of angle of attack above the maximum value. Additionally, the learning effect of both other groups is clearly visible, and the cueing group appears to have a smaller spread compare to the manual group. All groups though, seem to be able to reduce the amount of angle of attack above the maximum to an reduced level by Run 4.

Within the manual group, the learning effect is confirmed by statistical test: a significant difference is found (p < 0.005, $\chi^2 = 12.14$) and the post-hoc test indicates statistical significant differences between all runs (Runs 1 and 4 p < 0.5, Runs 1 and 8 p < 0.01, Runs 4 and 8 p < 0.05). The cueing group has differences (p < 0.001, $\chi^2 = 22.86$) between Runs 1 and 4 (p < 0.001), and between Runs 1 and 8 (p < 0.001), again indicating the initial learning effect and no after-effects in this metric. Statistical difference are present in the guidance group group (p < 0.05, $\chi^2 = 6.26$), with the post-hoc pointing to a difference between Runs 1 and 4 (p < 0.05), no difference between Run 4 and 5. This again indicates that participants perform better over the first four runs, but the next run has no statistical evidence. The figure does show that in the second block of four runs, without haptic feedback, the spread of the guidance group is slightly higher.



Figure 7.22: Integral above angle of attack limit during windshear (Manual Participant 7, 8.45rad s, not shown)

At the first run, we saw that the guidance group has the lowest median. The statistical test indeed confirms differences (p < 0.01, $\chi^2 = 9.76$), yet the post-hoc only confirms differences between cueing and guidance groups (p < 0.005). Differences between manual and guidance groups are almost significant (p = 0.10). At Run 4 and 8, no statistical significant differences are found.

7.4.3. SUBJECTIVE MEASURES

The first subjective measures are obtained after each run: a measure for situation awareness by asking the pilot whether (s)he was in control and whether (s)he was missing information, and a measure of workload. After each block of four runs, another questionnaire asked the participants to provide a Van Der Laan and MCH rating.

WAS THE PILOT IN CONTROL?

The results of the first measure can be found in Figs. 7.23 and 7.24, where no clear differences between groups can be observed. All groups seem to show an improving trend from Run 1 to Run 8.



Figure 7.23: Was the pilot feeling in control?

Statistical tests show significant differences between the runs for the manual group (p < 0.05, $\chi^2 = 8.86$), post-hoc indicates a significant difference between Runs 1 and 8 (p < 0.05), and 'near significance' between Runs 4 and 8 (p = 0.06). This confirms the observation of improving over all runs and indicates a gradual change of the runs. The differences within the cueing group are also significant (p < 0.001, $\chi^2 = 14.34$), and post-hoc indicates differences between Runs 1 and 4 (p < 0.05), between Runs 1 and 8 (p < 0.005), and not between Runs 4 and 8. As with the objective metrics, this can indicate that the metric improved over the first block, yet did not significantly improve after that.

WAS THE PILOT MISSING INFORMATION?

Results for the amount of information missed by the pilots did not result in any statistical significant results. Fig. 7.24 shows that the cueing group has an increased spread from Run 4 to Run 5. At the latter run, no haptic feedback was provided, possibly pointing out that the haptic feedback was providing the pilot with extra information. Additionally, the spread of the guidance group seems to be larger over most runs compared to the other groups, especially for the last three runs, where the spread is larger compared to the other groups.



Figure 7.24: Was the pilot missing information?

RSME

The mental load required for each of the runs was measured using the RSME and is shown in Fig. 7.25. Within groups, a general decreasing trend, can be observed. Statistical analysis for the cueing group shows differences (p < 0.05, $\chi^2 = 7.48$) and the post-hoc indicates significant differences between Run 1 and 8 (p < 0.05), confirming the general decreasing trend. Two between-group observations can be made: the median of the manual group for each run, except Run 7, is the highest, and the median of the cueing group, except Run 1, is the lowest. These last observations are not statistically significant.



Figure 7.25: Rating Scale Mental Effort (RSME)

VAN DER LAAN RATINGS

The Van Der Laan uses nine questions to score the system on usefulness and satisfaction, a perfectly useful and satisfying system would score minus two on both scales. As suggested by Van der Laan, both the absolute values and the difference from Block 1 to Block 2 are shown in respectively Fig. 7.26a and Fig. 7.26b. The absolute values show that the systems are in general well received and that the averaged-per-group results, indicated by thick indications, are situated close together, i.e., there is little effect of the haptic feedback support on the initial rating. No statistical significant difference is found between blocks, or between groups within one block.



Figure 7.26: Van Der Laan-ratings; crosses represent the score after Block 1, squares after Block 2, bold indicate group means

Looking at the change in rating after Block 2, the manual group seems to have an improvement in both usefulness *and* satisfaction, both other groups change mainly in satisfaction, to a lesser extent in usefulness. This corresponds to the verbal comments pilots gave after the second block: *"The visuals need time to learn to understand and use them."*, implying that the system becomes more useful after more training. The groups provided with haptic feedback made such comments less frequently. Nevertheless, this difference is again not statistically significant.

MCH RATINGS

Results for the MCH rating, shown in Fig. 7.27, are obtained using a decision-tree format and the main decision points are indicated with horizontal dashed lines. Note that lower scores are better, and going from score three to two requires that 'The information facilitates efficient decision making.' Two ratings do stand out: one participant rated the system twice as ten, where he indicated that all the information was too cluttered (especially the visual display), and warnings should only be provided when close to the limits. Other ratings provided by the participants in the guidance group did not change between blocks. Ratings for the two other groups have reduced spread and shift towards an improved rating: more participants rated the system one or two after Block 2. This indicates that the systems can provide efficient decision making, yet after training. Nonetheless, no statistical differences are found between blocks, nor between groups within a block.



Figure 7.27: Modified Cooper-Harper rating

7.4.4. DEBRIEFING QUESTIONNAIRE

A final questionnaire is used to structure the debriefing session and involved 19 questions when participants were in either the cueing or guidance group, 14 when in the manual group, as shown in Section A.8. The first question simply queried the pilots of the cueing and guidance groups whether they prefer flying the aircraft with, or without haptic feedback. The results in Fig. 7.28 show that in the guidance group all but one pilot prefered to fly with the haptic feedback, in the cueing group opinions are divided evenly. The difference between groups is confirmed with statistical significance (p < 0.05, $\chi^2 = 42$).





Unless mentioned otherwise, debriefing questions presented in this chapter used a five-point Likert scale, the results are shown in Fig. 7.29. Fig. 7.29a shows that about half of the participants in the cueing group agrees that the supplied haptic feedback is distracting, whereas the guidance group was more inclined to disagree with this statement.

Concerning the visual system, Fig. 7.29b, pilots in general disagree, i.e., the visual system is not distracting. For both the haptic and visual interface, all groups are unde-


(a) The haptic interface is distracting.



(d) The haptic interface requires a lot of training.



(g) The interface affected my workload. $^{\rm l}$



(j) The interface helps in preventing critical situations.²



(b) The visual interface is distracting.



(e) The visual interface requires a lot of training.



(h) During the experiment, my understanding of the interface increased after each flight.²



(k) If a critical situation occurs, the interface helps in resolving it.²



(c) The visuals and haptics gave conflicting signals.



(f) I was fighting the haptic interface.



(i) Using the interface, my knowledge on the edges of the aircraft performance changed. 3



(I) When implementing this system on an aircraft, what would be the effect on safety?⁴

Figure 7.29: Debriefing Likert-scale questions. Possible answers (unless specified otherwise) were: 1) Disagree 2) Slightly disagree 3) Disagree nor agree 4) Slightly agree 5) Agree

¹ Possible answers: 1) Decreased	2) Marginal decrease	3) Did not change	4) Marginal increase	5) Increased
² Possible answers: 1) Agree	Slightly agree	3) Disagree nor agree	4) Slightly disagree	5) Disagree
³ Possible answers: 1) Increased	2) Marginal increase	3) Did not change	4) Marginal decrease	5) Decreased
⁴ Possible answers: 1) Much safe	2) Safer	3) Safer nor unsafer	4) Unsafer	5) Much unsafe

cided on whether much training is required (Fig. 7.29d and 7.29e). The majority of the pilots who received haptic feedback did agree that the visual and haptic feedback did not give conflicting signals, attesting to the design work reported in Ref. [119]. Nevertheless, a difference in groups might be present on whether the pilots were fighting the haptic system: Fig. 7.29f shows that for the majority of the cueing group this was not an issue, whereas the majority of the guidance group reported to be 'fighting' the haptic system.

Looking at the subjective effect on workload, Fig. 7.29g, the manual group indicates a decrease, the cueing group indicates an increase, and the guidance group is undecided. This is the second question where a statistical difference is found between groups (p < 0.05, $\chi^2 = 6.15$), and the post-hoc test indicates that there is a significant difference between manual and cueing groups, confirming the difference between these two groups.

Participants are almost unanimous on the possible learning effect: Fig. 7.29h indicates that their understanding of the interface increased throughout the experiment. Finally, when one of the systems (visual, cueing haptics, of guidance haptics) is implemented, participants expect the knowledge on the aircraft performance boundaries to increase (Fig. 7.29i), it prevents critical situations to occur (Fig. 7.29j), and if a critical situation occurs, it can help in resolving it (Fig. 7.29k). In conclusion, pilots expect the visual and haptic systems to increase safety (Fig. 7.29l).

The questions at the end of the debriefing questionnaire allowed participants to elaborate on any of the above questions, comment on the reality of the simulator, and possible other comments. The manual group indicated that the display increased awareness, yet might could lead to pilots focussing on the instruments, not looking outside. Participants in the cueing group used the haptics as an alerting cue, and the visual display as a measure of criticality. The guidance haptic feedback made flying at the limits harder, yet provided intuitive feedback.

Both groups who received haptic feedback indicated that more training might be required. All groups indicated that their performance might have improved solely due to familiarity with the simulator. Verbatim comments can be found in Section C.3.

7.5. DISCUSSION

With the results of the experiment presented before, two designs of haptic feedback were evaluated for their effect on training and their general acceptance by pilots. First of all, the debriefing questionnaires showed that pilots appreciated the haptic feedback and expect the system to support in knowledge of the flight envelope limits, prevent critical situations, and even help in resolving them. However, in each group they do voice a concern for over-reliance on the haptic and/or visual system.

The design principle with haptic feedback is to keep the pilot in the loop, thereby making the pilot more aware of the situation and possibly less likely to be surprised to an automation failure compared to supervising an autopilot. Nonetheless, a next step to investigate this can be the inclusion of haptic feedback failures in the experiment design. In the following paragraphs, both the cueing and guidance haptic feedback design is discussed separately to evaluate the hypotheses presented before.

7.5.1. CUEING HAPTIC FEEDBACK

Using the results of the manual group as a baseline, the cueing group was expected to obtain similar performance levels and safety margins at the first run. All metrics showed the expected result at Run 1, both as observed from the graph, and confirmed by lack of significant differences between the results of the manual and cueing groups.

Next, the cueing group was expected to learn faster, which would result in differences between both groups at Run 4. Although not directly visible in the altitude lost or integral of angle of attack over the maximum, looking at the time where the maximum angle of attack is exceeded, the cueing group is indeed closer to zero compared to the manual group (although not statistically significant). Additionally, the median of the highest angle of attack achieved during the recovery for the cueing group is below the maximum, whereas this is not the case for the manual group.

Furthermore, the cueing group shows a quicker improvement of the safety margins over Runs 2, 3, and 4. When removing the haptic feedback from the cueing group, no after-effects are expected, i.e., no changes in metrics from Run 4 to Run 8. This latter is present in all objective metrics used, hence the pilots used the haptic feedback to learn quicker how to stay within the angle of attack limits, and they are still able to do so when no haptic feedback is provided anymore.

These observations are in line with the results found in our previous experiment, see Chapter 6. That experiment contained a group which did not receive haptic feedback in Block 1, followed by the cueing haptic feedback in Block 2. Since the second block contained haptic feedback, the data from that group can not be used as a comparison for training effects. The first block did, however, show also that participants who did not receive haptic feedback had a much harder time keeping the aircraft within its limits. Participants in the earlier experiment had even more difficulty compared to the current manual group in this experiment.

The cueing haptic feedback was expected to provide pilots with an increased situation awareness at the cost of an increased workload and without the benefit of immediate improvements at the first run. Both the information missing as the feeling of being in control for the manual and cueing group did not differ, providing no evidence that an increased situation awareness is present at Run 1.

The RSME, however, did show a (non statistical significant) difference: the median of the cueing group is lower compared to the manual group, actually indicating less work-load in contradiction to what was expected. The workload for cueing group was expected to decrease to Run 4 due to familiarization, which is partly confirmed by the RSME data: the statistical analysis only showed a difference between Runs 1 and 8, nonetheless, a decreasing trend in median RSME is visible supporting the expected decrease in workload due to familiarization.

The subjective situation awareness scales did not provide any evidence that a difference between groups is present at Run 1, 4, or 8. It did show an improving trend over the runs for all groups. This is to be expected due to familiarization with the task and model. Interestingly, going from Run 4 to Run 5, i.e., disabling the cueing haptic feedback, increased the spread for information missing. This can indicate that the haptic feedback is providing pilots information which they are consciously integrating in their control loop. Modified Cooper-Harper ratings did not indicate any differences compared to the manual group. The change in Van Der Laan ratings between blocks showed that the cueing group did not have an improvement in usefulness, whereas the manual group did. This might indicate that the haptic feedback made the pilots understand the visuals more easily, yet differences are very small. For these two system evaluation ratings, differences are considered too small to draw any conclusions from them.

In contrast to the observations on workload before, the debriefing questionnaire showed that participants who received the cueing feedback indicated a marginal increase in workload. This might be attributed to the fact that the haptic feedback is slightly distracting, as the pilots indicated as well. Because of the contradicting answers in the RSME and the debriefing no clear conclusions can be draw for a change in workload when using the cueing haptic feedback.

In conclusion, the cueing haptic feedback system is able to provide the pilots with support in the learning process without after-effects, and without having large effects on workload or situation awareness. It does lack the ability to support the pilot at the very first encounter with a certain limit. Therefore, it might be well-suited to provide such feedback to pilots in simulators to support the learning process, without the disadvantage of dependency on such a system when transferring to the real aircraft.

7.5.2. GUIDANCE HAPTIC FEEDBACK

The guidance group was expected to have an improved performance and safety margin compared to the other groups already at Run 1. For all metrics, there is a significant improvement in relation to the cueing group, and only 'near statistical significance' differences compared to the manual group. Nevertheless, the plots of all metrics show that the guidance group obtained a lower median *and* a lower spread. This means that the participants in the guidance group did not only do better, they were also more consistent across all participants. In other words, when using this haptic feedback system, a safer operation from the first time can be expected.

At Run 4, performance and safety margins are not statistically different from the manual group, yet the time of angle of attack above the maximum only shows two outliers above zero, and the highest angle of attack is clearly below the maximum value. This indicates that better metrics are achieved compared to the manual group. They are, however, of similar value as the cueing group. An improvement for the guidance group in time and integral of angle of attack above the maximum is found from Run 1 to Run 4, indicating that a minor learning process is still present.

When the haptic feedback is disabled, we expected the performance and safety margins to deteriorate. This did not largely happen in the amount of altitude lost, yet in the other safety margin metrics an increase in spread of the data can be seen. In other words, when no haptic feedback is supplied, three participants (or 25%) did exceed the angle of attack limits, giving some evidence for the "guidance hypothesis". This improved spread also contributes to the lack of significant changes between Run 1 and Run 5 for the guidance group.

The guidance haptic feedback was expected to actively support the pilot from the first run and thereby reducing the workload, at a cost of decreased situation awareness. The lack of changes in information missing and feeling of being in control at Run 1 gives no evidence that a decrease in situation awareness is present. The workload ratings at Run 1 show a (non-statistical significant) lower median rating compared to the manual group, indicating a reduced workload. This is a similar observation as with the cueing group which can indicate that for this setup/model/task, a multi-modal system, including haptic and visual feedback, is preferred over visual only.

Looking at all runs, the workload rating of the guidance group is consistently between the ratings of the cueing and the manual group, indicating that the guidance haptic feedback provides a reduced workload compared to no haptic. Nevertheless, in the debriefing questionnaire, not all pilots were convinced that a lower workload is achieved, and it shows that a number of them reported fighting the haptic feedback system at times. As the RSME and questionnaire are again contradicting, no clear conclusions can be draw with respect to workload.

Although the subjective situation awareness did not differ at Run 1, the larger spread on the amount of information missing for the guidance group can indicate that the haptic feedback is supporting the pilots, yet it is not transparent in why it is doing so. Additionally, during the final four runs, the spread of the data is much higher compare to the manual group, indicating that the pilots who do not receive guidance feedback anymore have difficulty adjusting to the new situation.

As before, modified Cooper-Harper ratings did not change, and no change in usefulness was present for the Van Der Laan ratings of the guidance group. The latter might indicate that the visuals were better understood because of the presence of haptics, i.e., multi-modal information. Nonetheless, differences for both ratings are considered too small to have any meaningful indication.

Compared to no haptic feedback, the guidance haptic feedback system is able to provide pilots with support during operation of the aircraft, without having large effects on workload or situation awareness. It is subjected to after-effect, as the spread of both the objective metrics and the amount of information missing increases when haptic feedback is no longer supplied. As such, it might be well-suited to provide such feedback to pilots during continuous operation of the aircraft. The after-effects have to be mitigated, for which more research is needed. Possibly a visual/aural indication can be added such that pilots are aware when the system is disabled and do not rely on it.

The design of the guidance haptic feedback was setup to make sure the pilot is in final control and can always over-ride the haptic inputs. To guarantee this with an increasing stiffness, a maximum force required to move the side stick was implemented after which a zero-slope force was present. The last example presented in the results showed one design flaw: when the participant exerts a force on the side stick of this maximum value, the feeling can be 'lock-in' on the flat slope. In this situation, the stiffness and neutral point can still change, yet the participant will not notice this as (s)he is already pulling at the maximum force level. Future designs are recommended to avoid such a condition by, for example, implementing a shallow force gradient beyond the maximum force such that at least the effect of the neutral point shift can be felt.

7.6. CONCLUSION

This chapter compared two designs of a haptic feedback system, using force feedback on the control device, to a situation without haptic feedback. These systems are expected to inform pilots on the flight envelope limits and protections in a modern fly-by-wire aircraft. One design used forcing functions which are asymmetric in time and amplitude, and warn the pilots of an approaching limit, as well as of the direction of the corrective action. In other words, this design is haptically *cueing* the pilot on the limits.

The second design actively changed the side stick mechanical properties to indicate a required direction to remain clear of the limits, as well as increased the stiffness for an input which brings the aircraft closer its limits. Therefore, this second design is haptically *guiding* the pilot near the limits.

Three groups – one with the cueing, one with the guiding, and one without haptic feedback – of each 12 PPL/LAPL pilots were used to evaluate the systems by flying 8 runs. Each run contained a trajectory shown on the outside visual and during the run a windshear was encountered. The first 4 runs were flown with haptic feedback for two out of three groups, the last four runs were flown without haptic feedback.

The results of the pilots flying with the cueing haptic feedback, compare to those who flew without haptic feedback, showed that the cueing feedback is able to provide support in the learning process. It can do so without suffering after-effects, i.e., when transitioning to a condition without haptic assistance. Additionally, it does not have large effects on workload or situation awareness. This system does not support the pilot at the first encounter with a certain limit.

Results showed that the guidance haptic feedback system is able to provide the pilots with support during operation of the aircraft, also without having large effects on workload or situation awareness. It is sensitive to an after-effect, and creates a reliance on the system, as the spread of both the objective metrics and the amount of information missing increases when no haptic feedback is supplied.

DISCUSSION, RECOMMENDATIONS, CONCLUSION

Each chapter in this thesis looks at an individual aspect or iteration of the haptic feedback system for flight envelope protection. This last chapter revisits the conclusions presented in those previous chapters and indicates how these provide answers to the main research questions. This information is used to form recommendations for future haptic feedback implementations and evaluations. Finally, the conclusion of this thesis is presented.

Parts of this chapter are currently under review to be published as:

TitleFlying by Feeling: Communicating Flight Envelope Protection through Haptic FeedbackJournalInternational Journal of Human-Computer Interaction

Authors D. Van Baelen, M. M. van Paassen, J. Ellerbroek, D. A. Abbink, and M. Mulder

8.1. DISCUSSION

W^{TTH} the advent of fly-by-wire, the mechanical connection between the pilot and the control surfaces disappeared. This resulted in the loss of haptic information, i.e., through the sense of touch on the controls, on the aircraft near the limits of the flight envelope. For example, a change in velocity did not result in a different force required on the controls, nor did the controls show the stall buffet. Missing this information might contribute to a reduced situation awareness near the limits of the aircraft. Additionally, the fly-by-wire systems enabled designers to include a flight envelope protection system: when approaching a limit of the aircraft flight envelope, the computer over-rides the inputs of the pilot to make sure no limit excursion will occur. Nevertheless, the state of such a system is not always clearly communicated to the pilots.

New control devices have the possibility to re-introduce haptic feedback on the flight deck, and are not limited to the feel of flight controls physically connected to the control surfaces. This presents a new design opportunity: using the haptic feedback to provide the pilot with information on the flight envelope protection system. In order to see how such feedback system can be designed, this thesis investigated several concepts.

In the following sections, each of the research questions presented in Chapter 1 are revisited and discussed. Those answers are used in the recommendations which follow in the next section. Finally, the conclusion of this thesis is presented.

8.1.1. FIRST DESIGN ITERATION

Previous exploratory research showed that a haptic feedback system could be useful, yet pilots noted that they missed a clear cue on the flight envelope protection activation. This thesis started with an investigation into the addition of such a cue to the haptic feedback system and results in the first research question:

Research Question 1: Combining vibrations and guidance design

Does a haptic feedback design combining stiffness changes, neutral point shifts, stick shaker, and discrete cues improve pilot situation awareness?

Chapter 2 described a haptic feedback system involving all those cues to inform pilots on the flight envelope protection system. To evaluate this design, an experiment involving eleven Airbus pilots was conducted as presented in Chapter 3. Pilots were asked to fly approaches during which they encountered either a windshear, or ice building up on the wings. The first scenario required the pilots to move close to the aircraft limit, while in the second scenario the limits approach the current aircraft state.

Results showed no significant changes in performance and safety margins for any of the conditions or control laws used. Nevertheless, the haptic feedback did not hinder pilots in performing their tasks either. The debriefing indicated that some pilots experienced an increased situation awareness, and most pilots see a potential benefit of implementing the haptic feedback system on a modern fly-by-wire flight deck. In conclusion, although most trends were positive, the experiment did not provide conclusive evidence of improved pilot situation awareness. Research Question 1 could not be answered due to the lack of evidence. This conclusion corroborates with the systems evaluated in literature: multiple haptic concepts show great potential, yet lack clear conclusive evidence. This was found for the evaluation of system which changes the reference point of the control input, and limits the deflection based on the remaining control space ([52]), as well as the starting point of this thesis. [53, 54] The evaluation of an increased resistance near the flight envelope edges did show more conclusive results, although it used several non-pilots in the evaluation which renders the results less convincing. [50]

A probable reason for the lack of conclusive evidence for the results presented in Chapter 3, and possibly for the evaluation in literature, is the low number of experiment participants, the natural tendency of pilots to stay away from the limits, and the difference in behavior of multiple pilots.

8.1.2. COMPLEMENTING VISUALS

Pilots in the first design evaluation indicated that the trigger of a haptic cue was not always clear. A solution was expected to be found in providing the information using multiple modalities, i.e., multiple senses. Therefore, a visual addition was proposed in the next research question:

Research Question 2: Complementing visuals

What kind of visual display can be used to complement the haptic information?

Chapter 4 looked into the design and evaluation of such a visual addition to the primary flight display to complement the haptic feedback in the communication of the flight envelope protections. The primary goal was to visualize the direction and force of the haptic feedback so as to make the cause of the haptic feedback cues more transparent with multi-modal signals.

The combination of haptic and visual feedback was evaluated by 16 professional Airbus pilots. These pilots flew twelve approaches for the measurement phase in which one of three scenarios occurred. Either a windshear occurred, a sidestep to an adjacent runway was required, or ATC required the pilot to discontinue the approach and goaround. The windshear scenario was, as in the previous evaluation, intended to force pilots to fly at the longitudinal aircraft limits, the sidestep scenario was intended to force pilots to the lateral aircraft limits.

Results did not show significant differences in performance, or safety metrics. The debriefing of this experiment revealed that pilot appreciation of the haptic feedback marginally increased, and that their understanding of the haptic feedback was increased. It is therefore recommended to show the flight envelope protection information using multiple modalities, in other words, we advice to use the visual display proposed in Chapter 4 when providing haptic feedback, answering Research Question 2.

Literature corroborates this conclusion since multi-modal information is generally assumed to improve acceptance. [58] Such improved acceptance with visual and haptic information was found in a remote-pilot task ([92]), and in an automotive context. [136] As such, independent of the application, any haptic feedback should be complemented with a visual indication.

8.1.3. VIBRATIONS DESIGN

The debriefing of the first design iteration showed that the vibro-tactile cue, the 'tickon-the-stick', was particularly well received by the participating pilots. Nevertheless, the direction of that cue was not always clear. The next research question focused on further investigation of those vibro-tactile cues, or vibrations, to inform the pilot of an active flight envelope protection:

Research Question 3: Vibrations design

Does a haptic feedback design using only vibrations improve pilot situation awareness?

First, an experiment was performed in the perceivability of direction and activation of such vibrations in Chapter 5. Results showed that a sawtooth-shaped vibration had a lower threshold compared to a triangular shape. At the threshold, participants are just able to tell that a cue was provided *and* indicate the correct direction of the cue. Therefore, the sawtooth-shaped vibration was recommended for implementation as a cue to alert pilots.

This sawtooth-shaped vibration was included in the haptic feedback system in Chapter 6. It is used to inform pilots on an approaching flight envelope limit as well as on the direction of the corrective action. It was evaluated by 24 PPL/LAPL pilots who flew two sets of four runs. One set included haptic feedback, one did not. Each run consisted of a flight trajectory shown on the outside visual and included a windshear encounter.

Results did not show the expected changes: most metrics were unchanged when switching the haptic feedback, only the acceptance scale (Van Der Laan-questionnaire) improved slightly when enabling the haptic feedback. This could be contributed to an order effect: the first four runs caused such a strong learning effect that this order effect rendered the comparison invalid.

The results also indicated, however, that enabling the haptic feedback appeared to improve the learning rate after the first run, and that no after-effects were present when the feedback is removed. As such, besides the fact that most pilots were positive about the system and indicated that they expect it to improve safety, the results suggest a potential training benefit. Such an effect could only be possible if some useful information is provided by the haptic feedback system, and therefore it should improve situation awareness, answering Research Question 3.

The training benefit makes the cueing design a possible addition to a flight simulator training device in order to enhance learning. It could, for example, be used to improve the cues during the training of the flare manoeuvre to indicate off-nominal states. [125]

Vibro-tactile cues in the form of asymmetric vibrations are used in domains other then aerospace, for example, in automotive to indicate an imminent lane departure. [36, 71] The sawtooth-shaped asymmetric vibration used in our application proved to be more powerful in communicating the direction of the cue. It is therefore recommended to use this cue in other applications, like automotive, to improve haptic communication.

8.1.4. GUIDANCE DESIGN

The haptic design based on vibrations alone did not help on the first encounter, and a different haptic feedback design approach was required. Literature showed that a design involving force feedback is often better able to guide pilots near the limits, which is the topic of the final research question:

Research Question 4: Guidance design

Does a haptic feedback design combining stiffness changes and neutral point shifts improve pilot situation awareness?

Chapter 7 presented a design that actively changed the side stick mechanical properties to indicate a required direction to remain clear of the limits, as well as increased the stiffness of the stick for inputs which would bring the aircraft closer its limits. Therefore, it is haptically guiding the pilot near the limits.

It was evaluated in a similar setup as the previous design using runs where pilots followed a trajectory shown on the outside visual during which a windshear was encountered. In the first set of four runs, one group (of twelve pilots) received the guidance haptic feedback, one group did not receive any haptic feedback at all. Next, a second set of four runs was flown where all groups performed the task without haptic feedback.

Results show that the guidance haptic feedback system is able to provide the pilots with support during operation of the aircraft, without having large effects on workload or subjective situation awareness. Since it supports from the very first use, it *has* to transfer information to the pilot and therefore, implicitly, improve pilot situation awareness, answering Research Question 4. The concept is, however, sensitive to an after-effect: it creates a reliance of the pilot on the system as indicated by the spread of both the objective metrics, and pilots commenting that information is missing when the haptic feedback is no longer supplied. This indicates that this last design can be useful to be implemented on the control device of an aircraft, as long as the support is always provided.

This conclusion is corroborated by literature on concepts which use force feedback to provide support from the first use, and is typically denoted as the "guidance hypothesis". [127] It was found in, for example, an abstract control task ([33]), and automotive examples. [38] Therefore, any application using force feedback should always provide the feedback, and should consider the implications when such feedback can no longer be supplied.

8.2. RECOMMENDATIONS

Based on the findings of the concept evaluations a number of general design recommendations can be given. In addition, some avenues for future research in haptic feedback support for envelope protection systems are stated as well.

8.2.1. DESIGN GUIDELINES

Chapter 1 introduced the two main types of haptic feedback discussed in this thesis: vibro-tactile and force feedback. The first concept, presented in Chapter 2, mixes both type of feedback: it uses both a 'tick-on-the-stick' to warn the pilot of the approaching limit, i.e., vibro-tactile feedback, and adjusted the stiffness to indicate the proximity, i.e., force feedback. The second design, presented in Chapter 6, involved only vibro-tactile cues in the form of asymmetric vibrations, and the last design, in Chapter 7, used pure force feedback by changing the stiffness and changing the neutral point.

Combining the division of the concepts based on the type of haptic feedback, together with the discussion of the research question presented above, results in the three main design guidelines from this thesis:

1. A haptic feedback system for flight envelope protection should be complemented with visual indications.

It shows pilots why a haptic cue is provided and improves pilot acceptance.

2. Vibro-tactile cues should be used during training to inform the pilot of the flight envelope protection system.

It increases learning rate, and it appears that such systems do not shown after-effects when support is removed.

3. Force feedback should be provided throughout the operation of the aircraft to inform the pilot of the flight envelope protection system.

It supports the pilot in staying within the flight envelope limits, yet is susceptible to performance degradation when removed.

8.2.2. FUTURE RESEARCH

To further investigate the application of haptic feedback on the modern flight deck, more research is required on several aspects.

Flight envelope protection system One common denominator across all concepts and evaluations is the data used to trigger the haptic feedback: each concept used an Airbus A320 model and the associated flight envelope protection system. Other aircraft manufacturers can have different approaches on how to provide flight envelope protection, for example, where Airbus typically has a 'hard' flight envelope protection in normal law, i.e., the aircraft can not be controlled outside the nominal flight envelope limits, Boeing aircraft have a 'soft' flight envelope protection, i.e., the flight control computers allow states outside the nominal flight envelope when the pilot exerts extreme control inputs. In the latter, pilots are informed about the flight envelope limits by aural, visual

cues, and an increased stiffness, which resembles the stiffness used on our haptic shared designs. [137]

Although the concepts have been evaluated on a single flight envelope protection system, the concepts have not only been tested on a hard flight envelope protection system. The first evaluation was performed with both Airbus normal and alternate control law, respectively a hard and a soft flight envelope protection system. [13] The other haptic feedback concepts were only evaluated in alternate law, to provide the participants with a more challenging environment.

It was argued in Chapter 2 that the haptic feedback concepts are intended to provide the same haptic cues in both normal and alternate law, i.e., in both hard and soft flight envelope protection systems. Since pilots should always be able to over-rule the haptic feedback, it provides a layer of soft flight envelope protection in all situations. As such, the concepts presented in this thesis can be used in conjunction with either a hard or soft flight envelope protection system.

A possible difference in the flight envelope protection system between aircraft manufacturer can be the variables which define the flight envelope. [138] For all concepts in this thesis, the flight envelope was defined by angle of attack, load factor, velocity and bank limits, which are commonly used throughout the aviation domain. We expect that the haptic cues used in this thesis can be extrapolated to other flight envelope definitions, yet this needs to be further investigated.

Startle by haptic feedback As mentioned in the previous paragraphs, the haptic feedback was designed to be used in both normal and alternate control law and provide intuitive support. This strategy results in pilots being more accustomed to the haptic feedback. Familiarity with the system is required as the haptic feedback system for flight envelope protection is activated only when the aircraft state approaches the flight envelope limits, which occurs very rarely. [98] If a pilot is not accustomed to the haptic feedback system, there is a risk for pilot startle. [139]

In the evaluations performed in this thesis, the haptic feedback system was explained in detail to each participant, and practised with, prior to starting the measurement phase. This procedure made sure that all participants were familiar with the simulator and the interfaces, including the haptic feedback, to reduce startle to a minimum. Next, the aircraft in each measurement run was operated at the edge of the flight envelope, making the participant accustomed to the warnings.

Although startle should not have been present in the evaluations, warnings are known to be a possible source of startle. [140, 141] Future research should therefore investigate whether the haptic feedback concepts in this thesis are prone to startle. Such experiment could, for example, simulate a departure and cruise phase, followed by an emergency situation, comparable to what happened to Air France 447.

Implementing asymmetric vibrations Asymmetric vibrations were found to provide an alerting and directional cue in Chapter 5, and were evaluated as a flight envelope protection cue in Chapter 6. The effects of those cues on the aircraft dynamics were investigated in Section 5.6, and considered to be minor. Similar vibrations were implemented in an automotive study and found that most of the motion is dampened by the driver holding the steering wheel. [36] Chapter 6, however, assumed that these cues could be implemented such that they have no effect on the aircraft dynamics.

Such an implementation was not designed and should be done in a next step. Possible solution could be at hardware level, for example, by providing the vibration internally in the side stick, or at software level by subtracting the vibration force from the measured force on the side stick.

After-effects The analysis of the haptic feedback systems presented in Chapter 6 and Chapter 7, looked into a possible effect of the haptic feedback system on the learning rate. It included a block of runs in which no haptic feedback was provided anymore to investigate any after-effects, i.e., a possible change in metrics when the feedback is not provided. Those experiment were confined to one scenario, windshear, which was repeated several times. Although two other checklists for other scenarios were presented, some participants indicated that they started to expect the windshear.

Using one scenario for training and evaluation is an approach similar to those found in evaluations of other haptic support systems with force feedback. [33, 127, 128] This approach was useful to compare results found in this thesis to those found in literature. Nevertheless, pilots are expected to encounter a large variety in scenarios during operation.

To further investigate the effect of the repeated scenario on the results and possible after-effects, an experiment which presents different scenarios to the pilots should be performed. If the repeated scenario of the experiments in Chapter 6 and Chapter 7 was not a confound, this new experiment should provide similar results.

Additionally, the guidance design presented in Chapter 7 was showing evidence of the guidance hypothesis: when the haptic feedback is no longer provided, a degradation in pilot situation awareness was present. The impact of this degradation should be further investigated and has to be mitigated. Such mitigation could involve a visual and/or aural indication to inform the pilot on the loss of haptic feedback.

Transfer-of-training Another element in training is the transfer-of-training which is an combination of how much knowledge obtained in the training is transferred to the targeted task, and whether that knowledge can be generalized. [142] The former was evaluated, for the experiments in Chapter 6 and Chapter 7, when the haptic feedback was no longer provided in the second phase of the experiment. The latter, whether the knowledge is generalized, is not evaluated in this thesis.

To evaluate whether the skill obtained during training can be used in different scenarios, a new generalization scenario should be presented to the participants. For example, in a flare training, an different approach was presented to the participants. [125] The generalization scenario in the context of this thesis should provide a new situation to the participants in which the flight envelope protection system is activated, but due to a different emergency. Evaluating the possible changes in metrics can indicate whether the participants can apply the skill to different applications, i.e., whether they can generalize the knowledge obtained during training. **Single-pilot operations** Economic advantages of having fewer pilots are transitioning future flight decks of commercial passenger aircraft from multi-pilot to single-pilot operations. [143] Although this transition is already present for smaller aircraft, such as business jets like the Cessna CJ series, larger aircraft did not make the transition yet. A major hurdle for this transition is maintaining situation awareness of the pilot on all systems, and the increase in workload when operating large aircraft in the highly congested airspace with fast communication and without room for deviations. [144]

In order to cope with the challenges of those single-pilot operation flight decks, disruptive changes are required to current flight decks of which several are investigated with grants from the National Aerospace Technology Exploitation Program, for example, see Ref. [145] and in the 'Disruptive Cockpit for Large Passenger Aircraft'-project of the Clean Sky program. [146] Their publications indicate that those technologies focus on improving automation, for example using vision-augmented approaches and autonomous instrument monitoring, and interfaces for pilots, for example using touch screens. However, the latter does not seem to include haptic feedback through the control device.

Haptic feedback has been shown in this thesis to have a potential to improve pilot awareness of the flight envelope protection system. Therefore, it should be used in those single-pilot flight decks to support the pilot. Additionally, haptic feedback has been used to support pilots in a path-following task ([41]), which could be used to inform the pilot on the nominal autopilot control action.

To mitigate high workload during an emergency, a remote pilot can assist the onboard pilot. [143] Proper crew resource management is a vital component for a successful cooperation. Haptic feedback can be used to communicate control actions between pilots, for example, by linking the movements of the control devices such that each pilot can feel the control actions of the other.

This thesis has investigated the use of haptic feedback for flight envelope protection system awareness. A haptic feedback design for the other uses might include similar cues, resulting in possible mixed messages. Following the research on the visual display complementing haptic feedback, any design should be supported by a matching visual indication. Nevertheless, more research is required to evaluate whether some or all types of haptic feedback can be mixed to support pilots in future single-pilot flight decks.

8.3. CONCLUSION

This thesis started with a concept which involved both types of haptic feedback considered: vibro-tactile *and* force guidance. It was important to make the distinction between both types as the evaluation of the combined system showed no conclusive results. To further investigate the two types of haptic feedback, both were separately developed further and tested. The evaluations of these concepts did not result in large differences in metrics, but each concept was found to have its strengths: vibro-tactile feedback proved to be valuable during training, force guidance valuable during continuous operation. Results indicated that pilots did not perform any worse when provided with haptic feedback. Overall, pilots agree with the potential added benefit of the haptic feedback. With this information, we can look back at the main research goal of this thesis:

Research goal

Within the current fly-by-wire flight deck, improve pilot situation awareness of the aircraft flight envelope protection system using haptic feedback.

The concepts considered in this thesis used haptic feedback to communicate the flight envelope limits, and therefore the flight envelope protection activation. Each concept was shown to provide this information for a specific use-case.

Therefore, this thesis proposes that pilot situation awareness of the aircraft flight envelope protection system can be improved using haptic feedback and should be implemented on the fly-by-wire flight deck.

APPENDICES

EXPERIMENT QUESTIONNAIRES

T^{HIS} appendix shows the documents used during an experiment, specifically the experiment performed to investigate the asymmetric vibrations for flight envelope protection. The briefing provided to the pilots approximately two days before the experiment is shown in Section A.1. Before the actual experiment starts, all pilots are asked to provide information concerning their experience, Section A.3, as well as sign an 'informed consent form' shown in Section A.2.

During the measurement phase of the experiment, participants are asked to fill in a short questionnaire after each run which is slightly different when the run was provided with or without haptic feedback, as exemplified in respectively Section A.4 and Section A.5. After a block, i.e. after four runs, the participants are asked to fill another questionnaire concerning their experience with the system provided covering all four runs, again slightly different when haptics is or is not provided as shown in Section A.6 and Section A.7. Finally, to streamline the debriefing after the experiment, a final questionnaire is provided to the participants for which an example is shown in Section A.8.

Three sets of debriefing questionnaires are used throughout this thesis: i) the set used in the first experiment, Table A.1, ii) one set for the groups who received haptic feedback in the later experiments, Table A.2, and iii) one for the group who did not receive haptic feedback in the last experiment where the questions are altered to reflect the visual interface, Table A.3. Note that the questions of the latter are numbered one to 14 when presented to the pilots, the numbering is changed here to match the question numbering of the other groups.

A.1. EXPERIMENT BRIEFING



Dear pilot

First of all, thank you for participating in our research experiment! Without your valuable contribution, our analysis cannot achieve a practical implication. Throughout this experiment, we will investigate the possibility of combining visual, aural and haptic feedback (also known as force feedback through the control device) in different sections of the flight envelope. To prepare you for the experiment, this document provides you with the most essential information. If you have any remaining questions, feel free to contact me!

Kind regards

Dirk Van Baelen

1 Logistical information

13:00 - approximately 16:00
Human-Machine-Interdace Simulator, Kluyverweg 1, 2629HS Delft
Route description can be found on https://iamap.tudelft.nl/poi/gebouw-62/
Service desk in the main entrance, you can ask at the desk for me (Dirk Van Baelen)
(0031) 15 278 9108
d.vanbaelen@tudelft.nl

2 Apparatus

The experiment is conducted in the Human Machine Interaction (HMI) simulator at the faculty of Aerospace Engineering, University of Technology Delft. It contains a glass cockpit setup resembling the displays of an Airbus. Furthermore, a projection system generates the outside world. You will be seated in the first officer position where you have access to the side stick, pedals, throttles and FCU which resemble an Airbus as close as possible in the simulator. An impression of the flight deck can be seen in Figure 1.

You will fly an Airbus A320-alike model. We are aware that you might not be accustomed to the size of such an aircraft, so we will provide you time to get accustomed to the model, its behavior and controls.



Figure 1: Human Machine Interaction Simulator

TDelft

A.2. INFORMED CONSENT

3 Experiment procedure

The experiment starts around 13:00 and ends at approximately 16:00. We have allocated time for a break, if you do need an extra break, please inform me.

Before we start flying, I will explain the visual, aural and haptic interfaces, as well as the emergency procedures for three scenarios. A small training flight is used to get you accustomed with the setup and model. If you have any questions during this part, or anywhere in the following, give a shout.

During the majority of the time, you are asked to fly through a series of climbing/descending gates which are shown on the outside visuals. Each run is ended with a small questionnaire concerning the visual/aural/haptic interfaces. Several runs are combined in a block, which is again ended with a small questionnaire. At the end of the experiment, one final questionnaire inquires you on the overall experience in the experiment. If any questions remain, I will make sure they are answered at that time!

Throughout the experiment, in each of the runs performed, we ask you to fly as you would in a real aircraft in the given circumstances. Note that due to limitations of the simulation, each flight is stopped when the limit of 50 ft AMSL is reached, regardless of any other event.

4 Your rights

Participation in the experiment is entirely voluntary. This also means that you can withdraw for any reason, even during or after the experiment. Collected data (log of the simulation states and paper forms) is saved anonymously. By participating you also agree that your (anonymous) data can be used in publishing this study. There will be no sound or image recordings. On the day of the experiment, I will ask you to sign a form indicating that you understand and agree with the above.

I am looking forward to welcoming you on the day of the experiment.

Page 2 / 4

A.3. BEFORE THE EXPERIMENT

pilot Personal and professional information To be able to put the results in relation to experience, you are asked to fill in this questionnaire. 1 PERSONAL 1.1 Age 1.2 Gender Male □ Female □ Other 2 FLIGHT EXPERIENCE 2.1 License type(s) 🗆 PPL 🗆 LAPL □ CPL □ ATPL 🗆 GPL 2.2 Flight hours Total: Airbus 2.3 Class-rating(s) Current Previous: 2.4 Type-rating(s) (if applicable) Current: Previous: 2.5 Current professional pilot function (if applicable) Captain □ First officer □ N/A 2.6 Do you have experience with glass cockpits? □ Yes, in flight 🗆 No □ Yes, simulator 2.7 Have you ever experienced windshear while airborne? □ Yes 🗆 No During which flight phase did the windshear occur? □ Take-off □ Approach □ Climb □ Landing Cruise □ Yes □ N/A, no ME 2.8 Have you ever experienced one engine out while airborne? During which flight phase did this occur? □ Take-off □ Approach Climb □ Landing □ Cruise 2.9 Have you ever experienced a sudden change in CG? □ Yes (for example a sudden load change, ...) □ Take-off During which flight phase did this occur? Approach \Box Climb □ Landing □ Cruise 2.10 Did you receive an upset prevention and recovery training (UPRT)? □ Yes 🗆 No 3 RESEARCH EXPERIENCE Have you ever participated in a research experiment? Yes No If yes, please elaborate on the type of experiment. (handling qualities research, motion cueing research, ...) Page 3 / 4

Informa	ation beyond this point is t	reated separately from	the above to ensure anonymity.	
	USEN I			
	DTICIDATION			
Recorded data will be sep provide be published that	parated from your identit allows you as an individu	y; at no time, neith al to be identified.	er now, nor in the future, will any info We certify to treat collected data accord	mation yo ing to goo
practice and follow sound	ethical rules.			
Note that we do record si	mulation states, and we	IO NUT record imag	ges or sounds throughout the experiment	
I hereby confirm that I and I declare that I volum relation to this work on a in the experiment at any	have read the experime tarily participate in this e symmetric vibrations. Fir moment before, during, c	nt briefing. I affir xperiment. Also, I a hally, I have been cle or after the experime	m that I understand the experiment m aware that the collected data can be early informed that I can opt-out of my p int.	nstruction published participatio
Please provide your sign above. Signing this form you as a participant.	nature below to indicate does not annul the respo	e that you agree to posibilities of the res	to participate in this experiment and searcher and Delft University of Technol	accept th ogy toward
Name	Age	Date	Signature	



A.4. AFTER EACH RUN WITH HAPTICS



A

A.6. AFTER EACH BLOCK WITH HAPTICS

1.1	How would you rate the	interface? Tie	ck one mark	on each line.			
			useful		useless		
			pleasant		unplease	ent	
			bad		good		
			nice		annoyin	g	
			effective		superflu	ous	
			irritating		likeable		
			assisting		worthles	s	
		L	Indesirable		desirable	e 4	
		raisin	g alertness		sleep-ind	ucing	
2	DISPLAY						
2.1	Indicate how helpful the	visual indicat	ions are to s	tay within the	limits.		
		Not at a	all Slightly	Moderately	Very	Extremely	
	Yellow bar						
	Red bar						
	Bar thickening						
2.2	Do you want to remove	an indication	from the dis	splay?			
	None	□ Red	bar				
	Yellow bar	🗆 Bar	thickening				
	If you want to remove a	in item, please	elaborate w	/hy:			
2.2		taltaat i i					
2.3	UU you want to add an	indication to t	ne aisplay?				
		what and where					
	ii yes, piease eiaborate (what and why:					

A

pilot HAPTICS 3.1 Indicate how helpful the haptic indications are to stay within the limits. Not at all Slightly Moderately Very Extremely Initial tick Repeated tick Stick shaker 3.2 Do you want to remove a cue from the haptics? □ None □ Repeated tick Initial tick Stick shaker If you want to remove a cue, please elaborate why: 3.3 Do you want to add an indication to the haptics? 🗆 No □ Yes If yes, please elaborate what and why:

Page 2 / 3

228



A.7. AFTER EACH BLOCK WITHOUT HAPTICS

	How would you rate the	he interface? Tick (one marl	k on each line.						
			useful		useless					
		p	leasant		unpleas	ent				
			bad		good					
			nice		annoyin	g				
		e	ffective		superflu	ous				
		ir	ritating		likeable					
		а	ssisting		worthles	55				
		und	esirable		desirabl	e 				
		raising a	lertness		sieep-in	aucing				
	Red bar									
	Yellow bar									
	Red bar									
	Bar thickening									
2.2	Do you want to remove an indication from the display?									
2.2	Do you want to remov				None Red bar					
2.2	Do you want to remov	□ Red bar	r							
2.2	Do you want to remov	Red bar Bar thic	r ckening							
2.2	Do you want to remov None Yellow bar If you want to remove	Red barBar thioan item, please ela	r ckening aborate v	vhy:						
2.2	Do you want to remov None Yellow bar If you want to remove	 Red bai Bar thic an item, please ela 	r ckening iborate v	vhy:						
2.2	Do you want to remov None Yellow bar If you want to remove Do you want to add a	Red bai Bar thic an item, please ela	r ckening aborate v display?	vhy:						
2.2	Do you want to remov None Yellow bar If you want to remove Do you want to add a No	Red ba Bar thic an item, please ela	r ckening aborate v display?	vhy:						
2.2	Do you want to remov None Yellow bar If you want to remove Do you want to add a No Yes	 Red bai Bar thio an item, please ela n indication to the 	r ckening uborate v display?	vhy:						
2.2	Do you want to remov None Yellow bar If you want to remove Do you want to add a No Yes If yes, please elaborate	Red bai Bar this an item, please ela	r ckening uborate v display?	vhy:						
2.2	Do you want to remov None Yellow bar If you want to remove Do you want to add a No Yes If yes, please elaborate	Red bai Bar this an item, please ela	r ckening uborate v display?	vhy:						



A.8. AFTER THE EXPERIMENT

Questionnaire after the experiment

Just one final set of questions left. Please complete the following questionnaire, it consists of 19 questions.

1.1	Which interface has your preference?					
	Only visual					
	Visual and haptics					
		Disagree	Slightly disagree	Disagree nor agree	Slightly Agree	Agree
1.2	The visuals and haptics gave conflicting signals.					
		Disagree	Slightly disagree	Disagree nor agree	Slightly Agree	Agree
1.3	I was fighting the haptic interface.					
		Decreased	Marginal decrease	Did not change	Marginal increase	Increase
1.4	Using the haptic system, my knowledge on the edges of the aircraft performance changed.					
		Disagree	Slightly disagree	Disagree nor agree	Slightly Agree	Agree
1.5	During the experiment (after the training), my understand- ing of the haptic interface increased after each flight.					
		Disagree	Slightly disagree	Disagree nor agree	Slightly Agree	Agree
1.6	The haptic interface requires a lot of training.					
		Disagree	Slightly disagree	Disagree nor agree	Slightly Agree	Agree
1.7	The visual interface requires a lot of training.					
		Disagree	Slightly disagree	Disagree nor agree	Slightly Agree	Agree
1.8	The haptic interface is distracting.					
		Disagree	Slightly disagree	Disagree nor agree	Slightly Agree	Agree
1.9	The visual interface is distracting.					
		Decreased	Marginal decrease	Did not change	Marginal increase	Increase
1 10	The hantic interface offected my workload					

pilot

Page 1 / 2

		Disagree	Slightly disagree	Disagree nor agree	Slightly Agree	Agree
1.11	The haptic interface helps in preventing critical situations.					
		Disagree	Slightly disagree	Disagree nor agree	Slightly Agree	Agree
1.12	If a critical situation occurs, the haptics helps in resolving it.					
		Disagree	Slightly disagree	Disagree nor agree	Slightly Agree	Agree
1.13	The haptics influenced my behavior.					
1.14	If you changed your behavior, can you indicate the change?					
		Much unsafer	Unsafer	Safer nor unsafer	Safer	Much safer
1.15	When implementing this system on an aircraft, what would be the effect on safety?					
	□ No □ Yes If you expect any negative outcome, please elaborate which:				_	
1.17	Do you have further comments concerning the above statemen	ts, or do you	want to ela	aborate an an	 swer? 	
2	SIMULATION Do you have any comments regarding the level of reality of the	simulation?				
					_	
					_	
3.1	Is there anything else you think we should know? Events the comments you would like to make?	at you think	we should	know about	or	
					_	
Question	Left label	Center label	Right label			
--------------------------------------------------------------------	-------------	--------------	-------------------			
What is your general feeling on the haptic feedback as an infor-	Negative	Neutral	Positive			
mation cue about the flight envelope?						
The haptic interface affected my workload.	Less	Neutral	More			
Using the haptic system, my knowledge on the edges of the air-	Less	Neutral	More			
craft performance changed.						
If a critical event occurs, the haptic interface helps to mitigate	Disagree	Neutral	Agree			
the consequences.						
The haptic interface changes the likelihood of human error.	Decreased	Neutral	Higher possibilit			
The haptic interface changed my behavior.	Disagree	Neutral	Agree			
The haptic interface distracts me.	Disagree	Neutral	Agree			
I was fighting the haptic interface.	Never	Sometimes	Always			
Do you expect any adverse impact on outcomes when using this	No		Yes			
technology? (on a two point scale)						
How would you grade the A320-alike dynamics?	Unrealistic	Acceptable	Perfect			
How would you grade the controls? (side stick with nominal	Unrealistic	Acceptable	Perfect			
feeling)						
How would you grade the displays?	Unrealistic	Acceptable	Perfect			
How would you grade the weather?	Unrealistic	Acceptable	Perfect			

3	estion	Left label		Center lauer		Might laber
-	Which interface has your preference? (on a two-point scale)		Only visual		Visual and haptics	
2	The visuals and haptics gave conflicting signals.	Disagree	Slightly disagree	Disagree nor agree	Slightly agree	Agree
ĉ	I was fighting the haptic interface.	Disagree	Slightly disagree	Disagree nor agree	Slightly agree	Agree
4	Using the haptic system, my knowledge on the edges of the air- craft performance changed.	Decreased	Marginal decrease	Did not change	Marginal increase	Increased
5	During the experiment (after the training), my understanding of the haptic interface increased after each flight.	Disagree	Slightly disagree	Disagree nor agree	Slightly agree	Agree
9	The haptic interface requires a lot of training.	Disagree	Slightly disagree	Disagree nor agree	Slightly agree	Agree
2	The visual interface requires a lot of training.	Disagree	Slightly disagree	Disagree nor agree	Slightly agree	Agree
8	The haptic interface is distracting.	Disagree	Slightly disagree	Disagree nor agree	Slightly agree	Agree
6	The visual interface is distracting.	Disagree	Slightly disagree	Disagree nor agree	Slightly agree	Agree
10	The haptic interface affected my workload.	Decreased	Marginal decrease	Did not change	Marginal increase	Increased
Π	The haptic interface helps in preventing critical situations.	Disagree	Slightly disagree	Disagree nor agree	Slightly agree	Agree
12	If a critical situation occurs, the haptics helps in resolving it.	Disagree	Slightly disagree	Disagree nor agree	Slightly agree	Agree
13	The haptics influenced my behavior.	Disagree	Slightly disagree	Disagree nor agree	Slightly agree	Agree
14	If you changed your behavior, can you indicate the change?			open question		
15	When implementing this system on an aircraft, what would be	Much unsafer	Unsafer	Safer nor unsafer	Safer	Much safer
	the effect on safety?					
16	Doyou expect any negative consequences of the haptic interface? If yes, please elaborate. (on a two-point scale)	No	Yes	space to indicate cha	mge	
17	Doyou have further comments concerning the above statements, or do vou want to elaborate an answer?	open question				
18	Do you have any comments regarding the level of reality of the simulation?	open question				
19	Is there anything else you think we should know? Events that you think we should know about or comments you would like to make?	open question				

Table A.2: Questions presented to the pilots in CHG and GHG after the experiment

Table A.3:
Q
uestions
g
resented
Ħ
Ă.
he
e pilots
E.
5
ΗG
aftei
the
experimen
+

Ques	tion	Left label		Center label	
4	Using the support system, my knowledge on the edges of the	Decreased	Marginal decrease	Did not change	Marginal
	aircraft performance changed.				
σı	During the experiment (after the training), my understanding	Disagree	Slightly disagree	Disagree nor agree	Slightly
	of the visual interface increased after each flight.				
7	The visual interface requires a lot of training.	Disagree	Slightly disagree	Disagree nor agree	Slightly
9	The visual interface is distracting.	Disagree	Slightly disagree	Disagree nor agree	Slightly
10	The visual interface affected my workload.	Decreased	Marginal decrease	Did not change	Marginal i
Ξ	The visual interface helps in preventing critical situations.	Disagree	Slightly disagree	Disagree nor agree	Slightly
12	If a critical situation occurs, the visuals help in resolving it.	Disagree	Slightly disagree	Disagree nor agree	Slightly
13	The visual influenced my behavior.	Disagree	Slightly disagree	Disagree nor agree	Slightly:
14	If you changed your behavior, can you indicate the change?			open question	
15	When implementing this system on an aircraft, what would be	Much unsafer	Unsafer	Safer nor unsafer	Safe
	the effect on sulery:				
16	Do you expect any negative consequences of the visual interface?	No	Yes	space to indicate cha	nge
	If yes, please elaborate. (on a two-point scale)				
17	Do you have further comments concerning the above statements,	open question			
	or do you want to elaborate an answer?				
18	Do you have any comments regarding the level of reality of the	open question			
	simulation?				
19	Is there anything else you think we should know? Events that	open question			
	you think we should know about or comments you would like				
	to make?				

Approach Charts and Checklists

This appendix shows the documents required to provide context to the scenarios for the pilots. In the first experiment, Chapter 3, professional pilots were invited to fly different approaches for which the required information is shown on an approach chart. The icing scenario involved an instrument landing in both the training and measurement phase, respectively to Rotterdam (EHRD) and Amsterdam/Schiphol (EHAM). Corresponding approach charts are shown in Section B.1 and B.2. Training and measurement of the windshear scenarios was performed during an approach to respectively Nice (LFMN, Section B.3) and Montpellier (LFMT, Section B.4). The checklist for the windshear recovery procedure as presented to the pilots is provided in Section B.5, which is retrieved fro mthe FCOM.

The later experiments did not use an approach chart as the flight path was shown on the outside visual. Pilots were provided with checklists for three emergency scenarios: i) a windshear recovery procedure, Section B.6, ii) a single engine stall, Section B.7, and iii) a sudden center of gravity shift, Section B.8. This second windshear recovery procedure is a modified version of the one used in the first experiment. The alterations are added to make pilots fly manually, and by stressing that loss of altitude had to be prevented, force pilots more towards the edge of the flight envelope. The single engine stall procedure is also heavily modified from the FCOM, and the checklist for the sudden center of gravity shift does not exist.



B.1. EHRD APPROACH CHART



B.2. EHAM APPROACH CHART

B.3. LFMN APPROACH CHARTS





242

B.4. LFMT APPROACH CHART



B.5. WINDSHEAR PROCEDURE (1)

A319 A320	ABNORMAL AND EMERGENCY	3.02.80	P 19
A321 CREW OPERATING MANUAL	MISCELLANEOUS	SEQ 100	REV 35
A red flag "WINDSHEA WINDSHEAR" repeat f windshear is de apply the follow A takeoff If before V The takeof variations of there is su If after V1 – THR LEVE – REACHII – SRS OR Airborne, ini – THR LEVEF – AP (if enga – SRS ORDE This includes to Note : 1. If eng 2. If the nece: – DO NOT C UNTIL OUT	WINDSHEAR AR" is displayed on each PFD associated with ed three times. etected either by the system or by ing recovery technique: 1 ff should be rejected only if sig occur below indicated V1 and the fficient runway remaining to stop /ERS NG VR DERS SAT TOGA S ged) he use of full back stick, if demanded. haged, the autopilot disengages when α is ge FD is not available, use an initial pitch assary to minimize the loss of height, increase HANGE CONFIGURATION (SLAT OF SHEAR. AONITOR FLIGHT PATH AND SP	an aural synti pilot obse pilot obse pilot decid the airplan 	hetic voice ervation, hirspeed des that ne. . TOGA ROTATE OLLOW DNFIRM . KEEP OLLOW prot. o 17.5°. If titude. GEAR)

B.6. WINDSHEAR PROCEDURE (2)

TUDelf	ft	ABNORMAL	AND	80.11A
A32X QUICK REFERENCE H/	ANDBOOK	EMERGENCY PROC	CEDURES	AUG 19
	<u> </u>	WINDSHEAI	R	
A red flag "WIN voice "WINDSH If windshear is apply the follow	DSHEAR" is IEAR" repea detected ei wing recove	displayed on each PFD asso ted three times. ither by the system or by pil ery technique:	ociated with an aur ot observation,	al synthetic
At takeoff	:			
If before	e V1:			
The taked indicated	off should b V1 and pilo	e rejected only if significant ot decides that there is suffi	t airspeed variation: cient runway remai	s occur below ning to stop.
If after \	V 1:			
THR LEV	/ERS			TOGA
REACHI	NG VR			ROTATE
SRS ORI	JERS			FOLLOW
If necessa	ary the fligh	nt crew may pull the sidestic	k full back.	
Note: I	If FD bars a Then, if nec	re not displayed, move towa cessary, to prevent loss in al	ard an initial pitch a titude, increase pite	ttitude of 17.5 ° ch attitude.
Airborne,	initial clin	nb or landing:		
AP and	A/THR			. DISENGAGE
THR LEV	/ERS AT TO	OGA	SET	OR CONFIRM
CONFIG	URATION	I	DO N	IOT CHANGE
PITCH				17.5
FLIGHT	PATH ANI	D SPEED		MONITOR
PREVEN	IT LOSS O	F ALTITUDE	INCREASE P	ITCH AS REQ
 Out of she 	ear:			
AP and/	or A/THR		ENC	GAGE AS REQ
FLIGHT	PATH			RESUME FPL

B.7. SINGLE ENGINE STALL

T UDelft	ABNORMAL AND	70.07
A32X QUICK REFERENCE HANDBO	EMERGENCY PROCEDURES	AUG 19
	ENG 1(2) STALL	
 N2 betweer 	50% and IDLE	
THR LEVER (ENG MASTE	AFFECTED ENGINE) R (AFFECTED ENGINE)	IDLE OFF
<u>ENG 1(2)</u>	6HUT DOWN	
 N2 above ID 	LE (title and procedure not displayed on I	ECAM):
 On grou 	nd :	
THR LEV ENG MA	ER (AFFECTED ENGINE) STER (AFFECTED ENGINE)	IDLE OFF
 In flight 		
THR LEV	R (AFFECTED ENGINE)	IDLE
∎ If abr	ormal ·	
ENG	MASTER (AFFECTED ENGINE)	OFF
ENG	1(2) SHUT DOWN	
■ If nor	mal :	
THR	EVER (AFFECTED ENGINE) SLOW	/LY ADVANCE
∎ If	a stall recurs :	
Т	IR LEVER (AFFECTED ENGINE)	REDUCE
■ If	a stall does not recur :	
C	ontinue engine operation.	
2T1 A32X	FOR RESEARCH PURPOSE ONLY	PRO-AE

B.8. SUDDEN CENTER OF GRAVITY SHIFT

		25.03A
A32X QUICK REFERENCE HANDBOOK	EMERGENCY PROCEDURES	AUG 19
	SUDDEN CG SHIFT	
A sudden CG shift can It is observed by an im without other immedia This event should be p	be caused by a shift in payload during a steep clim mediate change in pitch rate with AP disconnect, ate causes, typically accompanied by a loud bang revented by evenly distributing the payload and u	b or descent. or other noises. sing cargo net:
In case this does happe	en, the AP and AUTO TRIM will disable.	
The following actions a	are required:	
On ground:		
Return to gate		
AP		
PITCH		STABILIZE
PITCH TRIM	MANUAL	LY ADJUST
SPEED		. MONITOR
FLIGHT PATH		. MONITOR

C

EXPERIMENT DATA

T^{HIS} appendix contains more details on the results of the the final two experiments presented in Chapter 6 and 7. Section C.1 shows the flight paths of all trajectories for the vibrations design, as well as the flight paths of an experienced pilot. The latter is used to verify the results of the participants. Flight paths for the evaluation of the guidance design are shown Section C.2. Finally, responses to the open questions of the debriefing questionnaire of those experiments are shown in Section C.3.

Data for the flights are grouped per realization of the flight path. Plots show the entire flight of each pilot and indicate the flight phase where the windshear recovery had to be performed. The metrics used in the corresponding chapters are calculated on this specific part of the flights.

C.1. VIBRATIONS DESIGN FLIGHT PATHS

The plots in this section show the flights of the experiment presented in Chapter 6. Each plot includes a thicker black line which represents the flights of one extra participant (male) who was not included in the results. This participant was an outlier in terms of experience: although the participant had a valid current PPL, he was a retired professional pilot with 18,000 flight hours logged in jet fighters and large commercial aircraft (Boeing 747/767/777). Nevertheless, this participants was still invited to verify the results of the (in terms of large aircraft) inexperienced PPL/LAPL pilots.

For now, the verification of the results is limited to comparing the flight paths of the experienced pilot with the other pilots. This shows that in nominal conditions, the behavior of all pilots is comparable: they all follow the flight path presented on the outside visual. As discussed in Chapter 6, the flight path itself is rather challenging which is confirmed by the comments of the experienced pilot.

His first four flights (first block) were flown without haptic feedback, the final four flights (second block) are flown with haptic feedback. Flight paths are presented to the experienced pilot in the following order: five (Fig. C.1e), four (Fig. C.1d), eight (Fig. C.1h), six (Fig. C.1f), two (Fig. C.1b), one (Fig. C.1a), seven (Fig. C.1g), and three (Fig. C.1c). The figures show that the windshear performance (maximum altitude loss during recovery) is improving in that order. Furthermore, the initial performance on Fig. C.1e is similar to all other pilots. Starting from his fourth flight (flight path realization 6, Fig. C.1f), the experienced pilot is outperforming the other participants, even without haptic feedback.

As there is similar behavior during nominal and emergency situations and although this is based on one comparison, we believe that the results obtained with the 24 pilots can be extrapolated to results obtained with professional pilots.



(c) Flight path realization three

Figure C.1: All flown flight paths by all pilots, vertical lines indicate the gates on the outside visual, vertical thick blue line indicates windshear trigger, thick blue line represents flight path of experienced participant, dashed box indicates data used in the evaluation



(f) Flight path realization six

Figure C.1: (continued)



(h) Flight path realization eight

Figure C.1: (continued)

C.2. GUIDANCE DESIGN FLIGHT PATHS

The plots in this section show the flights of the experiment presented in Chapter 7. Light blue lines represent the group without haptic feedback, dark blue lines represent the group with cueing haptic feedback, and red lines represent the group with guidance haptic feedback.



(b) Flight path realization two

Figure C.2: All flown flight paths by all pilots, vertical lines indicate the gates on the outside visual, vertical thick blue line indicates windshear trigger, dashed box indicates data used in the evaluation







(d) Flight path realization four



⁽e) Flight path realization five

Figure C.2: (continued)



(h) Flight path realization eight

Figure C.2: (continued)

C.3. EXPERIMENT DEBRIEFING STATEMENTS

The questions at the end of the debriefing questionnaire provided the participants to elaborate on any of the above questions, comment on the reality of the simulator, and possible other comments. Additionally, expected negative effects and reported changes in behavior are reported below. Not every individual comments is shown, we tried to make sure that all comments made within one group are captured by one of the comments below.

Comments from the manual group include:

- Better anticipation of required aircraft control input based on visuals.
- More general awareness.
- More relaxed and more time to see ahead.
- A negative consequence could be that one starts flying by following only the visual signals.
- Display can distract from the actual flying.
- Resolving a critical situation could be further assisted with audible commands like "climb", "descend", etc.
- Besides this info, throttle awareness is the next most important item needing assistance.
- Beginning was quite a start-up, but I got used to the flying of the airplane.

Participants in the cueing group provided the following comments:

- I was not looking at instruments anymore, I reacted naturally to the haptics, and looked outside to monitor the behavior of the airplane.
- I made slightly stronger inputs.
- The haptic feedback could lead to too much stimulus together with sound/visual.
- I used the haptics to double check whether my input was correct, if so I made a stronger input.
- The haptic feedback indicates a problem, the visuals gives the possible space left (criticality).
- I followed the indicated direction for the suggested correction.
- Increased awareness of critical situation in angle of attack and speed.
- I made more subtle movements and anticipated more with the haptic feedback.
- When using all warning systems, extra training is recommended to process everything correctly.
- I think my performance improved solely due to experience with the sim.
- Audio warnings are best for VFR, with minimal visual information, and only shaker for stall.
- Sound can be added to support the protections.

Finally, the comments made in the guidance group included the following:

- You need to work harder to fly intentionally on the limit.
- Steers you easily towards the correct actions.
- Haptics makes you do a faster response, and you than check visuals for clues.
- I was struggling with conflicting visual and haptic inputs.
- More attention to critical situation, where you can follow your feeling.
- The haptic feedback might give a false sense of safety.
- I was distracted to what the reason was of the force? What is going on?
- Angle of attack should appear more prominently on the visuals.
- Training effect of each session has effect on my relation with haptic feedback.
- If possible, make the haptic settings adjustable to personal preferences.

REFERENCES

- [1] EASA, *EASA Annual Safety Review*, Tech. Rep. (European Aviation Environmental Report, 2019) Available at https://www.easa.europa.eu/eaer/.
- [2] EASA, EASA Annual Safety Review, Tech. Rep. (European Union Aviation Safety Agency, 2019) Available at https://www.skybrary.aero/index.php/Annual_Safety_Review_-_EASA.
- [3] IATA, IATA Safety Report 2017, Tech. Rep. (International Air Transport Association, 2018) Available at https://aviation-safety.net/airlinesafety/industry/reports/IATAsafety-report-2017.pdf.
- [4] R. H. Cook, An Automatic Stall Prevention Control for Supersonic Fighter Aircraft, Journal of Aircraft 2, pp. 171 – 175 (1965).
- [5] E. Field and D. Harris, A comparative survey of the utility of cross-cockpit linkages and autoflight systems' backfeed to the control inceptors of commercial aircraft, Ergonomics 41, pp. 1462 – 1477 (1998).
- [6] F. Wolfert, M. A. Bromfield, A. Stedmon, and S. Scott, *Passive Sidesticks and Hard Landings Is there a Link?* in *Proceedings of the AIAA Aviation Forum, Dallas (TX)* (2019) p. 10, AIAA-2019-3611.
- [7] Bureau d'Enquêtes et d'Analyses pour la sécurité de l'aviation Civile, Final Report On the accident on 1st June 2009 to the Airbus A330-203 registered F-GZCP operated by Air France flight AF 447 Rio de Janeiro - Paris (English courtesy translation), Tech. Rep. June 2009 (Bureau d'Enquêtes et d'Analyses pour la sécurité de l'aviation Civile, 2012).
- [8] Komite Nasional Keselamatan Transportasi, Aircraft Accident Investigation Report, PT. Indonesia Air Asia, Airbus A320-216, PK-AXC, Tech. Rep. December 2015 (Komite Nasional Keselamatan Transportasi, 2015).
- [9] L. Pollini, M. Razzanelli, M. Olivari, A. Brandimarti, M. Maimeri, P. Pazzaglia, G. Pittiglio, R. Nuti, M. Innocenti, and H. Bulthoff, *Design, Realization and Experimental Evaluation of a Haptic Stick for Shared Control Studies*, International Federation of Automatic Control 49, pp. 78 – 83 (2016).
- [10] D. Chatrenet, Les qualités de Vol des Avions de Transport Civil à Commandes de Vol Électriques, in Active Control Technology: Applications and Lesson Learned (AGARD, Neuilly-Sur-Seine, France, 1995) p. 5.

- [11] C. Favre, Fly-by-wire for Commercial Aircraft: the Airbus Experience, International Journal of Control 59, pp. 139–157 (1994).
- [12] D. Niedermeier and A. A. Lambregts, *Fly-By-Wire Augmented Manual Control -Basic Design Considerations*, 28th International Congress of the Aeronautical Sciences, Brisbane **100**, pp. 1 – 14 (2012).
- [13] Airbus, A319/A320/A321 Flight Crew Operating Manual, Tech. Rep. 36 (Airbus, 2003).
- [14] S. Hameed, T. Ferris, S. Jayaraman, and N. Sarter, Supporting Interruption Management Through Informative Tactile and Peripheral Visual Cues, Proceedings of the Human Factors and Ergonomics Society Annual Meeting 50, pp. 376–380 (2006).
- [15] P. J. Hopp, C. Smith, B. A. Clegg, and E. D. Heggestad, *Interruption Management: The Use of Attention-Directing Tactile Cues*, Human Factors: The Journal of the Human Factors and Ergonomics Society 47, pp. 1–11 (2005).
- [16] A. E. Sklar and N. B. Sarter, Good vibrations: Tactile feedback in support of attention allocation and human-automation coordination in event-driven domains, Human Factors 41, pp. 543–552 (1999).
- [17] M. K. Ngo, R. S. Pierce, and C. Spence, Using Multisensory Cues to Facilitate Air Traffic Management, Human Factors 54, pp. 1093–1103 (2012).
- [18] J. Prinet, Y. Wan, and N. Sarter, *Tactile spatial guidance for collision avoidance in NextGen flight operations*, Proceedings of the 60th Annual Meeting of the Human Factors and Ergonomics Society (HFES) pp. 303–307 (2016).
- [19] Y. Salzer, T. Oron-Gilad, A. Ronen, and Y. Parmet, Vibrotactile "on-thigh" alerting system in the cockpit. Human Factors: The Journal of the Human Factors and Ergonomics Society 53, pp. 118–131 (2011).
- [20] Y. Salzer and T. Oron-Gilad, Evaluation of an "on-thigh" vibrotactile collision avoidance alerting component in a simulated flight mission, IEEE Transactions on Human-Machine Systems 45, pp. 251–255 (2015).
- [21] W. Karlen, S. Cardin, D. Thalmann, and D. Floreano, *Enhancing pilot performance with a SymBodic system*, 2010 Annual International Conference of the IEEE Engineering in Medicine and Biology Society, EMBC'10 pp. 6599–6602 (2010).
- [22] A. H. Rupert, An instrumentation solution for reducing spatial disorientation mishaps, IEEE Engineering in Medicine and Biology Magazine 19, pp. 71–80 (2000).
- [23] J. B. F. van Erp, E. L. Groen, J. E. Bos, and H. a. H. C. van Veen, A tactile cockpit instrument supports the control of self-motion during spatial disorientation. Human factors 48, pp. 219–228 (2006).

- [24] J. B. Van Erp, L. Eriksson, B. Levin, O. Carlander, J. A. Veltman, and W. K. Vos, *Tactile cueing effects on performance in simulated aerial combat with high acceleration*, Aviation Space and Environmental Medicine 78, pp. 1128–1134 (2007).
- [25] F. Bouak, J. Kline, and B. Cheung, *Tactile cueing in detecting and controlling pitch and roll motion*, Aviation Space and Environmental Medicine 82, pp. 951–958 (2011).
- [26] A. Kelley, R. L. Newman, B. D. Lawson, and A. H. Rupert, A Materiel Solution to Aircraft Upset, Proceedings of the AIAA Guidance, Navigation, and Control Conference, National Harbor (MD) (2014), 10.2514/6.2014-0445, AIAA-2014-0445.
- [27] A. M. Kelley, C. M. Grandizio, A. Estrada, and J. S. Crowley, *Tactile Cues in Continuous Operations: A Preliminary Study*, Aviation, Space, and Environmental Medicine 85, pp. 172–176 (2014).
- [28] A. H. Rupert, B. D. Lawson, and J. E. Basso, *Tactile Situation Awareness System: Recent Developments for Aviation*, Proceedings of the Human Factors and Ergonomics Society Annual Meeting 60, pp. 722–726 (2016).
- [29] K. Fellah and M. Huiatni, *Tactile Display Design for Flight Envelope Protection and Situational Awareness*, IEEE Transactions on Haptics 12, pp. 87 98 (2019).
- [30] H. Jung, G. Park, and M. Todd, *Structural impact detection with vibro-haptic interfaces*, Smart Materials and Structures **25** (2016).
- [31] J. M. McGuirl and N. B. Sarter, Presenting in-flight icing information: A comparison of visual and tactile cues, AIAA/IEEE Digital Avionics Systems Conference -Proceedings 1, pp. 2A21–2A28 (2001).
- [32] T. M. Lam, H. W. Boschloo, M. Mulder, and M. M. Van Paassen, Artificial Force Field for Haptic Feedback in UAV Teleoperation, IEEE Transactions on Systems, Man, and Cybernetics Part A:Systems and Humans 39, pp. 1316–1330 (2009).
- [33] R. J. Kuiper, D. J. F. Heck, I. A. Kuling, and D. A. Abbink, *Evaluation of Haptic and Visual Cues for Repulsive or Attractive Guidance in Nonholonomic Steering Tasks*, IEEE Transactions on Human-Machine Systems 46, pp. 672–683 (2016).
- [34] M. M. van Paassen, R. P. Boink, D. A. Abbink, M. Mulder, and M. Mulder, Four design choices for haptic shared control, in Advances in Aviation Psychology, Volume 2, Advances in Aviation Psychology, Vol. 2, edited by M. A. Vidulich, P. S. Tsang, and J. M. Flach (Routhledge, New York, NY, 2017) Chap. 12, pp. 237–254.
- [35] M. Mulder, J. J. Pauwelussen, M. M. van Paassen, M. Mulder, and D. A. Abbink, *Active Deceleration Support in Car Following*, IEEE Transactions on Systems, Man, and Cybernetics Part A:Systems and Humans 40, pp. 1271–1284 (2010).

- [36] J. Navarro, F. Mars, J. F. Forzy, M. El-Jaafari, and J. M. Hoc, Objective and Subjective Evaluation of Motor Priming and Warning Systems Applied to Lateral Control Assistance, Accident Analysis and Prevention 42, pp. 904–912 (2010).
- [37] D. A. Abbink and M. Mulder, *Exploring the Dimensions of Haptic Feedback Support in Manual Control*, Journal of Computing and Information Science in Engineering 9, pp. 11006 1–9 (2009).
- [38] S. M. Petermeijer, D. A. Abbink, M. Mulder, and J. C. F. De Winter, *The Effect of Haptic Support Systems on Driver Performance: A Literature Survey*, IEEE Transactions on Haptics 8, pp. 467–479 (2015).
- [39] K. Goodrich, P. C. Schutte, and R. A. Williams, Haptic-multimodal flight control system update, in Proceedings of the AIAA Aviation Technology, Integration, and Operations Conference, Virginia Beach (VA) (2011) AIAA 2011-6984.
- [40] S. de Stiger, M. Mulder, and M. M. van Paassen, *Design and evaluation of a haptic flight director*, Journal of Guidance, Control and Dynamics **30** (2007).
- [41] D. G. Beeftink, C. Borst, D. Van Baelen, M. M. van Paassen, and M. Mulder, *Haptic Support for Aircraft Approaches with a Perspective Flight-path Display*, in *IEEE International Conference on Systems, Man, and Cybernetics, Miyazaki* (2018) pp. 3016 3021.
- [42] M. Olivari, F. Nieuwenhuizen, H. Bülthoff, and L. Pollini, *Pilot Adaptation to Different Classes of Haptic Aids in Tracking Tasks*, Journal of Guidance, Control, and Dynamics 37, pp. 1741–1753 (2014).
- [43] N. Gandhi, N. D. Richards, and A. J. Bateman, Simulator Evaluation of an In-Cockpit Cueing System for Upset Recovery, AIAA Guidance, Navigation, and Control Conference pp. 1–22 (2014).
- [44] M. S. Whalley, W. Hindson, G. Thiers, and M. Rutkowski, A Comparison of Active Sidestick and Conventional Inceptors for Helicopter Flight Envelope Tactile Cueing, Tech. Rep. (American Helicopter Society, Inc., Virginia Beach (VA), 2013) available at https://ntrs.nasa.gov/search.jsp?R=20000092066.
- [45] V. Sahasrabudhe, J. F. Horn, N. Sahani, A. Faynberg, and R. Spaulding, *Simulation investigation of a comprehensive collective-axis tactile cueing system*, Journal of the American Helicopter Society 51, pp. 215 224 (2006).
- [46] G. J. Jeram and J. V. R. Prasad, *Tactile Avoidance Cueing for Pilot Induced Oscillation*, in *Proceedings of the AIAA Atmospheric FLight Mechanics Conference*, *Austin (TX)* (2003) pp. 1–9.
- [47] D. H. Klyde and D. T. McRuer, Smart-Cue and Smart-Gain Concepts Development to Alleviate Loss of Control, Journal of Guidance, Control, and Dynamics 32, pp. 1409–1417 (2009).

- [48] D. H. Klyde, C. Liang, N. Richards, and B. Cogan, Use of active inceptor cueing to mitigate pilot-vehicle system loss of control, in Proceedings of the AIAA Guidance, Navigation and Control Conference, Minneapolis (MN) (2012) AIAA 2012-4765.
- [49] J. M. Wilson, M. E. Peters, and G. W. Pettit, Flight tests of a general aviation envelope protection and stability augmentation system, in Proceedings of the AIAA Guidance, Navigation and Control Conference, National Harbor (MD) (2014) AIAA 2014-0602.
- [50] F. J. J. Schmidt-Skipiol and P. Hecker, *Tactile Feedback and Situation Awareness*, in Proceedings of the 15th AIAA Aviation Technology, Integration, and Operations Conference, Dallas (TX) (2015) p. 10.
- [51] K. A. Ackerman, D. A. Talleur, R. S. Carbonari, E. Xargay, B. D. Seefeldt, A. Kirlik, N. Hovakimyan, and A. C. Trujillo, *Automation Situation Awareness Display for a Flight Envelope Protection System*, Journal of Guidance, Control, and Dynamics 40, pp. 964–980 (2017).
- [52] V. Stepanyan, K. Krishnakumar, G. Dorais, S. Reardon, J. Barlow, A. Lampton, and G. Hardy, *Loss-of-Control Mitigation via Predictive Cuing*, Journal of Guidance, Control, and Dynamics 40, pp. 831–846 (2017).
- [53] J. Ellerbroek, M. J. M. Rodriguez Martin, T. J. J. Lombaerts, M. M. van Paassen, and M. Mulder, Design and Evaluation of a Flight Envelope Protection Haptic Feedback System, in 13th IFAC Symposium on Analysis, Design, and Evaluation of Human-Machine Systems HMS 2016, Kyoto, Vol. 49 (2016) pp. 171 – 176.
- [54] T. J. J. Lombaerts, G. Looye, J. Ellerbroek, and M. Rodriguez Martin, *Design and Piloted Simulator Evaluation of Adaptive Safe Flight Envelope Protection Algorithm*, Journal of Guidance, Control and Dynamics 40, pp. 1902–1924 (2017).
- [55] Boeing, 777 Flight Crew Training Manual, Tech. Rep. (Boeing, 2008).
- [56] W. Fu, M. M. Van Paassen, and M. Mulder, *Developing Active Manipulators in Aircraft Flight Control*, Journal of Guidance, Control, and Dynamics 42, pp. 1755 1767 (2019).
- [57] International Organization for Standardization, Ergonomics of human-system interaction - Part 910: Framework for Tactile and Haptic Interaction (ISO 9241-910:2011), Tech. Rep. (International Organization for Standardization, Brussels, Belgium, 2011).
- [58] P. A. Hancock, J. E. Mercado, J. Merlo, and J. B. F. van Erp, *Improving target detection in visual search through the augmenting multi-sensory cues*, Ergonomics 56, pp. 729–738 (2013).
- [59] D. Harris, *Human factors for civil flight deck design*, (Ashgate Publishing Limited, Cornwall, 2004) Chap. 4,6, pp. 69–102,141–155.

- [60] J. B. van Erp, K. Kyung, S. Kassner, J. Carter, S. Brewster, G. Weber, and I. Andrew, Setting the standards for haptic and tactile interactions: Iso's work, in Proceedings of EuroHaptics: International Conference on Human Haptic Sensing and Touch Enabled Computer Applications, Amsterdam (Netherlands) (2010) pp. 353 – 358.
- [61] S. G. Corps, *Airbus a320 side stick and fly by wire an update*, in *Proceedings of the Aerospace Technology Conference and Exposition, Long Beach (CA)* (1986) p. 13.
- [62] J. W. Hegg, M. P. Smith, and L. Yount, Sidestick controllers for advanced aircraft cockpits, in Proceedings of the IEEE/AIAA 11th Digital Avionics Systems Conference, Seattle (WA) (1992) pp. 491 – 499.
- [63] G. Warwick, *Active Sidestick Controls Make Commercial Debut*, Aviation Week & Space Technology (2015).
- [64] J. C. Gibson and R. A. Hess, *Stick and Feel System Design*, (AGARD, Neuilly-sur-seine, 1997) Chap. 1.4, pp. 34 36.
- [65] V. Sahasrabudhe, J. F. Horn, N. Sahani, A. Faynberg, and R. Spaulding, *Simulation investigation of a comprehensive collective-axis tactile cueing system*, Journal of the American Helicopter Society 51, pp. 215 224 (2006).
- [66] W. von Grünhagen, M. Müllhäuser, M. Abildgaard, and R. Lantzsch, Active inceptors in fhs for pilot assistance systems, in Proceedings of the 36th European Rotorcraft Forum, Paris (France) (2010) p. 11.
- [67] S. de Stigter, M. Mulder, and M. M. van Paassen, *Design and Evaluation of a Haptic Flight Director*, Journal of Guidance, Control, and Dynamics **30**, pp. 35–46 (2007).
- [68] J. Smisek, E. Sunil, M. M. van Paassen, D. A. Abbink, and M. Mulder, *Neuromuscular-System-Based Tuning of a Haptic Shared Control Interface for UAV Teleoperation*, IEEE Transactions on Human-Machine Systems 47, pp. 449–461 (2017).
- [69] G. J. J. Ruijgrok, *Elements of Airplane Performance*, (VSSD, Delft, 2009) Chap. 1,3, pp. 1–23,51–62, 2nd ed.
- [70] Airbus, A318/A319/A320/A321 Flight Crew Training Manual, Tech. Rep. (Airbus, 2015).
- [71] Z. Huang, Y. Wu, and J. Liu, Research on Effects of Pattern, Amplitude and Frequency of Pulse Steering Torque Warnings for Lane Departure, Transportation Research Part F: Traffic Psychology and Behaviour 31, pp. 67 – 76 (2015).
- [72] T. J. J. Lombaerts and G. H. N. Looye, Design and flight testing of nonlinear autoflight control laws, in Proceedings AIAA Guidance, Navigation, and Control Conference, Minneapolis (MN) (2012) p. 24.
- [73] T. H. G. Megson, Aircraft Structures for Engineering Students, 4th ed. (Elsevier, 2007).

- [74] M. S. Whalley and M. Achache, Joint u.s./france investigation of helicopter flight envelope limit cueing, in Proceedings of the 52th American Helicopter Society Forum, Washington D.C. (DC) (1996) p. 30.
- [75] D. H. Klyde, N. Richards, and B. Cogan, Mitigating unfavorable pilot interactions with adaptive controllers in the presence of failures/damage, in Proceedings of the AIAA Atmospheric Flight Mechanics Conference, Portland (OR) (2011) p. 17.
- [76] FAA, Windshear Training Aid Volume 1, Tech. Rep. (Federal Aviation Administration, 1990).
- [77] J. K. Evans, An Examination of Aviation Accidents Associated with Turbulence, Wind Shear and Thunderstorm, Tech. Rep. (NASA Langley Research Center, Hampton (VA), 2013) available at https://ntrs.nasa.gov/search.jsp?R=20130013459.
- [78] C. D. Wickens and A. L. Alexander, Attentional Tunneling and Task Management in Synthetic Vision Displays, The International Journal of Aviation Psychology 19, pp. 182–199 (2009).
- [79] FAA, Instrument Flying Handbook (FAA-H-8083-15B), (United States Department of Transportation, Federal Aviation Administration, Airman Testing Standards Branch, Oklahoma City, 2012) Chap. 4, pp. 13–17.
- [80] R. Parasuraman and V. Riley, *Humans and automation: Use, misuse, disuse, abuse,* Human Factors **39**, pp. 230 – 253 (1997).
- [81] M. Mulder, M. Mulder, M. M. van Paassen, and D. A. Abbink, *Haptic gas pedal feedback*, Ergonomics 51, pp. 1710–1720 (2008).
- [82] Pooley, *Air Pilot's Manual: Flying Training*, 7th ed., edited by P. Godwin (Air Pilot Publishing, Cranfield, 2005) p. 389.
- [83] A. W. Black and P. A. Taylor, *The Festival Speech Synthesis System: System Documentation*, Tech. Rep. HCRC/TR-83 (Human Communciation Research Centre, University of Edinburgh, Scotland, UK, 1997) available at http://www.cstr.ed.ac.uk/projects/festival.html.
- [84] Netherlands AIP, AD 2.EHM-ADC Aerodrome Chart, Online (2018).
- [85] F. R. H. Zijlstra, Efficiency in work behaviour: A design approach for modern tools, Ph.D. thesis, Delft University of Technology (1993).
- [86] G. J. S. Wilde, Risk homeostasis theory and traffic accidents : propositions, deductions and discussion of dissension in recent reactions, Ergonomics 31, pp. 441–468 (1988).
- [87] T. Melman, J. C. F. de Winter, and D. A. Abbink, *Does haptic steering guidance instigate speeding? A driving simulator study into causes and remedies*, Accident Analysis and Prevention **98**, pp. 372–387 (2017).

- [88] T. J. J. Lombaerts, G. H. N. Looye, A. Seefried, M. Neves, and T. Bellmann, Proof of concept simulator demonstration of a physics based self-preserving flight envelope protection algorithm, Engineering Applications of Artificial Intelligence 67, pp. 368–380 (2018).
- [89] Boeing, Statistical Summary of Commercial Jet Airplane Accidents, Worldwide Operations | 1959 – 2016, Tech. Rep. (Boeing Commercial Airplanes, 2019).
- [90] CAST/ICAO Common Taxonomy Team, Aviation occurence categories definitions and usage notes, (2017).
- [91] KNKT, *Aircraft Accident Investigation Report PK-AXC*, Tech. Rep. (Komite Nasional Keselamatan Transportas, 2015).
- [92] V. Ho, C. Borst, M. M. van Paassen, and M. Mulder, *Increasing acceptance of haptic feedback in uav teleoperation by visualizing force fields*, IEEE International Conference on Systems, Man, and Cybernetics (2018).
- [93] N. Meuleau, C. Neukom, C. Plaunt, D. E. Smith, and T. Smith, *The Emergency Landing Planner Experiment* (NASA, 2011).
- [94] A. D. T. Rijndorp, C. Borst, C. C. de Visser, O. Stroosma, M. Mulder, and M. M. van Paassen, Aviate, navigate: Functional visualizations of asymmetric flight envelope limits, AIAA Information Systems-AIAA Infotech at Aerospace (2017).
- [95] T. Lombaerts, S. Schuet, D. Acosta, J. Kaneshige, K. Shish, and L. Martin, *Piloted simulator evaluation of safe flight envelope display indicators for loss of control avoidance*, Journal of Guidance, Control, and Dynamics 40, pp. 948–963 (2017).
- [96] K. A. Ackerman, B. D. Seefeldt, E. Xargay, D. A. Talleur, R. S. Carbonari, A. Kirlik, N. Hovakimyan, A. C. Trujillo, and I. M. Gregory, *Flight envelope information-augmented display for enhanced pilot situation awareness*, AIAA Infotech at Aerospace (2015).
- [97] D. Van Baelen, J. Ellerbroek, M. M. van Paassen, and M. Mulder, Design of a Haptic Feedback System for Flight Envelope Protection, in Proceedings of the AIAA Modeling and Simulation Technologies Conference, Kissimmee (FL) (2018) AIAA-2018-0117.
- [98] J. W. Rustenburg, D. A. Skinn, and D. O. Tipps, *Statistical Loads Data for the Airbus A-320 Aircraft in Commercial Operations*, Tech. Rep. (FAA, 2002).
- [99] C. D. Wickens, J. Lee, Y. Liu, and S. G. Becker, *An Introduction to Human Factors Engineering*, 2nd ed. (Pearson Prentice Hall, 2004).
- [100] FAA, *Electronic Flight Displays, Advisory Circular 25-11B*, Tech. Rep. (Federal Aviation Administration, 2014).
- [101] J. Langer, *Operational use of Angle of Attack on Modern Commercial Jet Airplanes*, in *Aero*, 12 (Boeing Commercial Airplanes Group, 2000).

- [102] M. M. van Paassen, O. Stroosma, and J. Delatour, DUECA Data-Driven Activation in Distributed Real-Time Computation, (Modeling and Simulation Technologies Conference, 2000) pp. 1–7.
- [103] OpenStreetMap contributors, OpenStreetMap maps, https://www.openstreetmap.org (2019).
- [104] J. D. Van Der Laan, A. Heino, and D. De Waard, A simple procedure for the assessment of acceptance of advanced transport telematics, Transportation Research Part C: Emerging Technologies 5, pp. 1–10 (1997).
- [105] M. L. Cummings, K. Myers, and S. D. Scott, Modified cooper harper evaluation tool for unmanned vehicle displays, in Proceedings from UVS Canada: Conference on Unmanned Vehicle Systems Canada '06 (2006).
- [106] Airplane Upset Prevention & Recovery Training Aid for Transport Category Airplanes, ICAO (2017), rev. 3.
- [107] L. Le Vie, Survey of Quantitative Research Metrics to Assess Pilot Performance in Upset Recovery, Tech. Rep. (NASA, 2016).
- [108] D. A. Abbink, M. Mulder, and E. R. Boer, *Haptic shared control: Smoothly shifting control authority*, Cognition, Technology and Work 14, pp. 19–28 (2012).
- [109] H. W. Tappeiner, R. L. Klatzky, B. Unger, and R. Hollis, Good Vibrations: Asymmetric Vibrations for Directional Haptic Cues, in 3rd Joint Eurohaptics Conference and Symposium on Haptic Interfaces for Virtual Environment and Teleoperator Systems, Salt Lake City (UT) (2009) pp. 285 – 289.
- [110] D. Van Baelen, J. Ellerbroek, M. M. van Paassen, and M. Mulder, Evaluation of a Haptic Feedback System for Flight Envelope Protection, in Proceedings of the AIAA Guidance, Navigation and Control Conference, San Diego (CA) (2019) p. 25, AIAA-2019-0367.
- [111] P. Rabbitt, *Inside psychology: A science over 50 years*, (Oxford University Press, Oxford, 2008) Chap. 13, pp. 177 – 189, 1st ed.
- [112] M. M. Van Paassen, *Biophysics in Aircraft Control. A model of the neuromuscular system of the pilot's arm*, Ph.D. dissertation, TU Delft (1994).
- [113] W. Fu, A. Landman, M. M. Van Paassen, and M. Mulder, *Modeling Human Difference Threshold in Perceiving Mechanical Properties From Force*, IEEE Transactions on Human-Machine Systems 48, pp. 359 368 (2018).
- [114] W. Fu, M. M. Van Paassen, D. A. Abbink, and M. Mulder, *Framework for Human Haptic Perception with Delayed Force Feedback*, IEEE Transactions on Human-Machine Systems 49, pp. 171 182 (2019).
- [115] F. A. A. Kingdom and N. Prins, *Psychophysics: A Practical Introduction*, (Elsevier, Oxford, 2010) Chap. 5, pp. 121 – 151, 1st ed.
- [116] L. Sherry and R. Mauro, Design of Cockpit displays to explicitly support flight crew intervention tasks, in IEEE/AIAA 33rd Digital Avionics Systems Conference, Colorado Springs, (CO) (2014) p. 13.
- [117] L. J. Kramer, T. Etherington, E. Evans, T. Daniels, S. D. Young, J. Barnes, S. Schuet, T. J. J. Lombaerts, and L. Sherry, *Evaluation of Technology Concepts for Energy, Automation, and System State Awareness in Commercial Airline Flight Decks,* in *Proceedings of the AIAA Modeling and Simulation Technologies Conference, San Diego (CA)* (2019) p. 20, AIAA-2019-0982.
- [118] S. Schuet, T. J. J. Lombaerts, V. Stepanyan, J. Kaneshige, G. Hardy, K. H. Shish, P. Robinson, T. Etherington, L. Kramer, E. Evans, T. Daniels, S. D. Young, and D. Rodzon, *Piloted Simulation Study Findings On Stall Recovery Guidance*, in *Proceedings of the AIAA Modeling and Simulation Technologies Conference, San Diego (CA)* (2019) p. 23, AIAA-2019-0981.
- [119] G. de Rooij, D. Van Baelen, C. Borst, M. M. van Paassen, and M. Mulder, Supplementing Haptic Feedback through the Visual Display of Flight Envelope Boundaries, in Proceedings of the AIAA Guidance, Navigation, and Control Conference, Orlando (FL) (2020) p. 25, AIAA-2020-0373.
- [120] T. J. J. Lombaerts, G. Looye, A. Seefried, M. Neves, and T. Bellmann, *Development and Concept Demonstration of a Physics Based Adaptive Flight Envelope Protection Algorithm*, International Federation of Automatic Control 49, pp. 248 253 (2016).
- [121] A. H. Wertheim, J. E. Bos, and W. Bles, *Contributions of roll and pitch to sea sickness*, Brain Research Bulletin 47, pp. 517–524 (1998).
- [122] A. H. Roscoe and G. A. Ellis, A subjective rating scale for assessing pilot workload in flight: a decade of practical use, Tech. Rep. 900019 (Royal Aerospace Establishment, 1990).
- [123] R Core Team, *R: A Language and Environment for Statistical Computing*, R Foundation for Statistical Computing, Vienna, Austria (2017).
- [124] D. M. Pool, G. A. Harder, and M. M. van Paassen, *Effects of Simulator Motion Feedback on Training of Skill-Based Control Behavior*, Journal of Guidance, Control, and Dynamics **39**, pp. 889–902 (2016).
- [125] P. Deldycke, D. Van Baelen, D. M. Pool, M. M. van Paassen, and M. Mulder, *Design and Evaluation of a Haptic Aid for Training of the Manual Flare Manoeuvre*, in *Proceedings of the AIAA Modeling and Simulation Technologies Conference*, *Kissimmee (FL)* (2018) p. 21, AIAA-2018-0113.
- [126] D. Van Baelen, J. Ellerbroek, M. M. van Paassen, D. A. Abbink, and M. Mulder, Using Asymmetric Vibrations for Flight Envelope Protection, in Proceedings of the AIAA Modelling and Simulation Technology conference, Orlando (FL) (2020) p. 27, AIAA-2020-0409.

- [127] R. A. Schmidt, D. E. Young, S. Swinnen, and D. C. Shapiro, Summary Knowledge of Results for Skill Acquisition: Support for the Guidance Hypothesis, Journal of Experimental Psychology Learning Memory and Cognition 15, pp. 352 – 359 (1989).
- [128] C. J. Winstein and R. A. Schmidt, *Reduced Frequency of Knowledge of Results Enhances Motor Skill Learning*, Journal of Experimental Psychology Learning Memory and Cognition 16, pp. 677 691 (1990).
- [129] S. M. Petermeijer, D. A. Abbink, and J. C. F. De Winter, Should Drivers Be Operating Within an Automation-Free Bandwidth? Evaluating Haptic Steering Support Systems With Different Levels of Authority, Human Factors 57, pp. 5–20 (2015).
- [130] Z. Wang, T. Kaizuka, and K. Nakano, Effect of haptic guidance steering on lane following performance by taking account of driver reliance on the assistance system, in 2018 IEEE International Conference on Systems, Man, and Cybernetics (SMC), Miyazaki (JP) (2018) pp. 2717–2723.
- [131] D. Van Baelen, J. Ellerbroek, M. M. van Paassen, and M. Mulder, *Design of a Haptic Feedback System for Flight Envelope Protection*, Journal of Guidance, Control, and Dynamics 43, pp. 700–714 (2020).
- [132] D. Van Baelen, J. Ellerbroek, M. M. van Paassen, D. A. Abbink, and M. Mulder, Just Feeling the Force: Just Noticeable Difference for Asymmetric Vibrations, in Accepted to Proceedings of the 2020 IEEE International Conference on Human-Machine Systems, Rome (Italy) (2020) p. 6.
- [133] M. A. Vidulich, P. S. Tsang, and J. Flach, eds., Advances in Aviation Psychology, Volume 2, (Taylor & Francis, London, 2017) Chap. 12, pp. 237 – 254, 2nd ed.
- [134] R. A. Schmidt and R. A. Bjork, New Conceptualizations of Practice: Common Principles in Three Paradigms Suggest New Concepts for Training, Psychological Science 3, pp. 207 – 217 (1992).
- [135] Y. Benjamini and Y. Hochberg, *Controlling the false discovery rate: A practical and powerful approach to multiple testing*, Journal of the Royal Statistical Society. Series B (Methodological) 57, pp. 289–300 (1995).
- [136] K. Labrujere, The Effect of Complementing Haptic Shared Control with Visual Feedback on Driving Behaviour during Overtake Manoeuvres, Master's thesis, Delft University of Technology (2020).
- [137] A. A. Lambgregts, Flight Envelope Protection for Automatic and Augmented Manual Control, in EuroGNC 2nd CEAS Specialist Conference on Guidance, Navigation & Control, Delft (NL) (2013) pp. 1364 – 1383.
- [138] H. Koolstra, H. J. Damveld, and J. A. Mulder, Envelope Determination of Damaged Aircraft, in Proceedings of the AIAA Guidance, Navigation, and Control Conference, Minneapolis (MN) (2012) p. 22, AIAA-2012-4699.

- [139] A. Landman, E. L. Groen, M. M. van Paassen, A. W. Bronkhorst, and M. Mulder, Dealing With Unexpected Events on the Flight Deck: A Conceptual Model of Startle and Surprise, Human Factors 59, pp. 1161 – 1172 (2017).
- [140] J. A. Kochan, E. G. Greiter, and F. Jentsch, Surprise and Unexpectedness in Flying: Database Reviews and Analyses, Proceedings of the Human Factors and Ergonomics Society Annual Meeting 48, pp. 335 – 339 (2004).
- [141] C. M. Belcastro and J. V. Foster, Design and Evaluation of a Haptic Aid for Training of the Manual Flare Manoeuvre, in Proceedings of the AIAA Guidance, Navigation, and Control Conference, Toronto (CA) (2010) p. 39, AIAA-2010-8004.
- [142] L. Reder and R. L. Klatzky, *The Effect of Context on Training: Is Learning Situated?*, Tech. Rep. CMU-CS-94-187 (School of Computer Science, 1994).
- [143] D. Comerford, S. L. Brandt, J. Lachter, S.-C. Wu, R. Mogford, V. Battiste, and W. W. Johnson, NASA's Single-Pilot Operations Technical Interchange Meeting: Proceedings and Findings, Tech. Rep. (NASA, 2013).
- [144] A. J. Stimpson, J. Ryan, and M. L. Cummings, Assessing Pilot Workload in Single-Pilot Operations with Advanced Autonomy, Proceedings of the Human Factors and Ergonomics Society Annual Meeting 60, pp. 675 – 679 (2016).
- [145] W. Khan, D. Ansell, K. Kuru, and M. Bilal, *Flight Guardian: Autonomous Flight Safety Improvement by Monitoring Aircraft Cockpit Instruments*, Journal of Aerospace Informatoin Systems 15, pp. 203 214 (2018).
- [146] Disruptive cockpit for large passenger aircraft, https://www.cleansky.eu/european-aviation-in-the-driving-seatclean-skys-disruptive-cockpit-for-large-passenger-aircraft (2014), june 19 2020.

ACKNOWLEDGEMENTS

As for most clichés, there is truth to this: "The name on the cover is just one, but this thesis has been completed thanks to multiple people." It clearly holds for my thesis and it is therefore paramount to spend considerable space of this booklet to acknowledge those people who have contributed and made the results possible. But before that, I would like to apologize to everyone at the department who saw me in my bicycling gear, it probably was not the best view, but it rides so much better.

First and foremost to my family; the person who was/is most important, is the person most dear and close to me: Isabelle. Although sometimes you laughingly wondered "When I would get a real job?", I do appreciate you making the big step of moving to The Hague which allowed me to pursue this path. Throughout my thesis you have calmed me when I was stressing for an experiment, cheered with me when I got a success, and comforted me when I was down. Without your unconditional support and encouragement, my thesis just would not have been finished.

To my parents, Guido and Lydia, thank you for supporting me through the years, believing in me, and allowing me to get my flying license, which really sparked my interest for aviation (even more). Next, you enabled me to study abroad in the far North, which in the end brought me to this point. Both my sisters, Maria and An (and 'aanhangsels' Lieven and Bjorn), thank you for surviving with me and making the wedding-streak a success (three in 1.3 years!). To my in-law family, Wim, Karin, Dieter and Wouter, thank you for allowing me to enter your 'Boelen'-world and supporting both Isabelle and me.

Next, I want to thank my 'academic family': they raised me from my BSc to the person I am now. I think my 'academic family' has four fathers: the first father to thank is René who always knew the answer, made DUECA a pleasure and challenge to work with, and provided me with the possibility to do things I enjoy during both my MSc and PhD. Without your knowledge and input, the content and programming would not have been what it is now, and would have taken so much longer.

Thank you to my second 'academic father', Max, for talking to me when I was wondering which MSc track I was going to do, I still remember sitting in your office for the first time where you took time to explain the entire process. I even asked you what the possibilities were of a PhD, I still wonder whether it influenced me starting this PhD. During my PhD, Max was there to ask me to be better, do greater things, critically look at all the papers, and maybe most importantly, point me to my deadlines.

The third father, Joost, you introduced me to the project at the beginning, and helped me out with the initial programming. Later, when I got the project more in my own hands, I appreciated the external view which pointed me to things I sometimes skipped.

The involvement from my fourth 'academic father-in-law' (different faculty), David, steadily increased from the start to the end of my project: you correctly pointed me to 'use more of my supervisors'. I enjoyed your high-level point of view and the discussions on the possibilities of haptic feedback. In my humble opinion, the combination of René,

Max, Joost and David was a perfect supervision *team* for me: a mix of high- and low-level feedback at the correct times and places, a push to go in detail and to finish things, and support on both the professional and personal side.

My starting point was the MSc thesis of Mitchell Rodriguez, supervised by Max, Joost and Olaf, and supported by Gertjan Looye. I would like to thank them all for their work. Another thank you to Gertjan for supporting me with the aircraft model, and to Thomas Lombaerts for the adaptive control laws (and wine in California, and alligator in Florida).

The experiments I conducted would never have been possible, or have any meaning, without the participation of numerous pilots who spent four to six hours with me (and/or Gijs). Therefore, thank you to all of you (in a random order): Rob, Dimitri, Manos, Jose, Edgar, Peter, Thibaut, Tony, Carsten, Gerhard, Frank, Annemarie, Giel, Christian, Peter, Henk, Ad, Jurriën, Casper, Dennis, Maarten, Rolf, George, Kees, Martien, Jan, Rik, Arno, Willemjan, Fons, Sander, Ferdinand, Peter, Jacco, Olaf, Richard, Tanguy, Jan, Tom, Luuk, Dirk, Robin, Willem, Erik, Hans, Sander, Jasper, Arno, Menno, Lorenzo, Christophe, Fabian, Peter, Joris, John, Erik, Patrick, Gerben, Peter, Joel, Arthur, Jan, Jeroen, Rogier, Twan, Gerben, Dinesh, Peter, Marc, Manuel, Storm, Arjan, Chretien.

If my promotors/supervisors are my 'academic fathers', my most important 'academic uncle' would be Daan: it was a crucial moment when I moved to your office. The number of possibilities I have had thanks to you are close to endless: you helped me navigate administration, you invited me to Pieter-Jan's committee, you guided me through my first conference, you told me to speak up if I wanted to teach, you got me to grade turbulence reports, you supported me in giving the neuro-muscular system lecture, you told me to not do stuff (and focus on my own work), you said I should ''Just write, it will be fine" (which it did), you pushed me to do better, you had a contagious laugh in the office, you did not hate me for interrupting your meetings (only occasionally), ... My thesis wouldn't have been the same (and probably took much longer) if it wasn't for you: thanks Daan!

There are so much more 'academic family members' to thank. To start, all of the C&S staff with more 'uncles' who guided me in my first conference: Erik-Jan, Coen, Ping; 'uncle' Clark who made lunches easy-peasy, lemon-squeezy; both 'uncles' Hans and Xander who make the flight practical the most awesome and enjoyable practical ever. 'Auntie' Bertine who helped in the administrative maze, fixed requests in a flash, and shares stories about Pim. 'Uncles' like 'Software' Olaf, Alwin, Harold, Ferdinand, Andries, Menno, Fred, 'Hardware' Olaf, who make sure the simulators and PH-LAB were/are in top notch condition and who fixed everything I damaged. Especially to 'Software' Olaf, thank you for helping me out and sorry for breaking stuff. At some point you must have been thinking: "If he is touching the simulator, what is he going to break today?"

Next, thank you to my 'academic family members' who I help during their MSc: to Pieter-Jan who taught students what flying is like and endured me next to him in the simulator control room; to Derek who guided pilots in the tunnel and I could support you with haptics; to Annemarie who fooled pilots in banking and I could help with setting the simulation up; and all the other students I helped with their simulation. Especially to Gijs who made a visual display supplementing the haptic feedback, hence making it more transparent; Chapter 4 is the article Gijs wrote for his MSc thesis, and the haptics would not have been received as well as it did without his work: thanks Gijs.

And to my supporting 'academic siblings': thanks to all of you fellow C&S PhD'ers for joining me in the wild ride, for being present at my wedding, and for making the PhD-life at C&S an experience I will never forget/regret. Special thanks to Sarah for joining the journey from before DSE to here; Wei for helping me out in my MSc/PhD work and staying in San Diego; Isabel for exchanging chocolate for a bike; Matei and Sihao for the C&S rides; Jelmer for joining the trip to China and allowing me to work on another simulation; Kasper for being the PhD-next-door who no-one could beat in terms of journal papers and making dinner when I was having a bad day; Julia for sharing the good and bad times about being a PhD student; Daniel for having a perfect sense of humour; and the SciTech2020 Roomies (Daniel, Gijs, Rowenna) for an awesome conference and final year.

To my 'in-law siblings' from the haptics lab at 3mE, especially Sarvesh and Timo, thank you for sharing your views on haptics. The final family members are all my 'academic nieces and nephews', PhD students in other sections and faculties. Especially those who joined me in the PhD Board/Council: thank you to the 'old board', Fabricio, Fardin, Roberto, and, Svenja for introducing me to the work, and thank you to the 'new board', Bram, Bruce, Ines, Tiago, Vis, and Yuxin, for trusting me as chairman and making all events and meetings a success. Here a special thanks to Emmanuel for inviting me to join the PhD Board when he needed a successor, and thanks to Mario for succeeding me. The council would not have worked without the support from the GS Director, both Dick and Stephan, and the immense support of the secretaries Geeta, Colette, and Nancy. Thanks to all of you for sharing the experiences.

Next to my (academic) family, more people were not directly involved in my thesis, but made me who I am today. Thank you to Wouter for signing my papers (I was happy signing yours), I value the discussions on energy policies and how to analyse them to avoid working; and to Joyce for being the ray of sunlight in the room and decorating it perfectly. Thank you to Nathalie, for signing Isabelle's papers and being the positive spirit in the group; to Thomas for conducting events (partly including our wedding) and analysing politics. To Alexander, and, Yanou and Peter, thank you for the revival dinners.

To all the people in the Red Cross who introduced me to first aid, getting me started into teaching, and supporting me in my first organizational roles: thank you for introducing me to this valuable life experience. In particular, I want to thank Bart, who go me in, and provided me with wisdom both in and outside the Red Cross.

To our Dutch friends Sander and Femke, thank you for surviving dance classes with us and making the paso doble enjoyable. For Julia, Michiel and Senna, thank you for sharing house-buying-issues. For Daniel, Sary and Aaron, thank you for having us for dinner and zoo, it was never boring with you around. And to our favourite neighbours, Tom and Myrthe, although we got to meet rather late as neighbours in the building when you decided to move, I still consider us neighbours (country-wise).

Finally, a word of thank you to 'The PHD Movie', although I value my supervising team and enjoyed the topic, the particularities of the 'PhD-life' showcased in the movies (and comics) helped me through my deepest lows in my time as a PhD student. For everyone who is still pursuing their degree and has a feeling of "What am I doing here?", I advise you to watch the movies, and know: "You're not alone."

275

CURRICULUM VITÆ

Dirk VAN BAELEN

22-05-1992 Born in Herentals, Belgium.

EDUCATION

2016 - 2020	PhD Aerospace Engineering	
	Delft University of Technology	
	Thesis:	Haptic Feedback for Flight Envelope Protection
	Promotors:	Dr. ir. M. M. van Paassen
		Prof. dr. ir. D. A. Abbink

- 2013 2016 MSc Aerospace Engineering Delft University of Technology
- 2010 2013 BSc Aerospace Engineering Delft University of Technology

ACADEMIC ACTIVITIES

Reviewer for IEEE Int. Conf. on Systems, Man, and Cybernetics (SMC) Reviewer for IFAC Conf. on Cyber-Physical & Human-Systems (CPHS)

AWARDS

2020	Winner of the pitch competition
	organized by the Aerospace Faculty PhD Council/Graduate School
2019	AIAA Guidance Navigation and Control Student Paper Finalist
2018	Excellent Student Prize of the International Graduate Summer School
	in Aeronatics and Astronautics, Beihang University
2017	Winner of the 'Extreme environments'-session at the PhD symposium
	organized by the Aerospace Faculty PhD Council/Graduate School
2013	Bachelor acquired Cum Laude

LIST OF PUBLICATIONS

- D. Van Baelen, M. M. van Paassen, J. Ellerbroek, D. A. Abbink, and M. Mulder, *Evaluating Stick Stiffness and Position Guidance for Feedback on Flight Envelope Protection*, Accepted for publication in the proceedings of the 2021 AIAA Modeling and Simulation Technologies Conference, Nashville (TN).
- A. van den Hoed, H. M. Landman, D. Van Baelen, O. Stroosma, M. M. van Paassen, E. L. Groen, M. Mulder, A Scenario for Inducing the Leans Illusion in a Hexapod Motion Simulator, Accepted for publication in the proceedings of the 2021 AIAA Modeling and Simulation Technologies Conference, Nashville (TN)
- 8. **D. Van Baelen**, J. Ellerbroek, M. M. van Paassen, D. A. Abbink, M. Mulder, *Just Feeling the Force: Just Noticeable Difference for Asymmetric Vibrations*, Proceedings of the 2020 IEEE International Conference on Human-Machine Systems, Rome (Italy).
- D. Van Baelen, J. Ellerbroek, M. M. van Paassen, M. Mulder, *Design of a Haptic Feedback* System for Flight Envelope Protection, Journal of Guidance, Control, and Dynamics 43, pp. 700 – 714 (2020)
- 6. **D. Van Baelen**, J. Ellerbroek, M. M. van Paassen, D. A. Abbink, M. Mulder, *Using Asymmetric Vibrations for Feedback on Flight Envelope Protection*, Proceedings of the 2020 AIAA Modeling and Simulation Technologies Conference, Orlando (FL).
- G. de Rooij, D. Van Baelen, C. Borst, M. M. van Paassen, M. Mulder, Supplementing Haptic Feedback Through the Visual Display of Flight Envelope Boundaries, Proceedings of the 2020 AIAA Guidance, Navigation and Control Conference, Orlando (FL).
- 4. **D. Van Baelen**, J. Ellerbroek, M. M. van Paassen, M. Mulder, *Evaluation of a Haptic Feedback System for Flight Envelope Protection*, Proceedings of the 2019 AIAA Guidance, Navigation and Control Conference, San Diego (CA).
- 3. D. G. Beeftinkg, C. Borst, **D. Van Baelen**, M. M. van Paassen, M. Mulder, *Haptic Support for Aircraft Approaches with a Perspective Flight-Path Display*, Proceedings of the 2018 IEEE International Conference on Systems, Man, and Cybernetics, Miyazaki (Japan).
- P.-J. Deldycke, D. Van Baelen, D. M. Pool, M. M. van Paassen, M. Mulder, *Design and Evaluation of a Haptic Aid for Training of the Manual Flare Manoeuvre*, Proceedings of the 2018 AIAA Modeling and Simulation Technologies Conference, Kissimmee (FL).
- 1. **D. Van Baelen**, J. Ellerbroek, M. M. van Paassen, M. Mulder, *Design of a Haptic Feedback System for Flight Envelope Protection*, Proceedings of the 2018 AIAA Modeling and Simulation Technologies Conference, Kissimmee (FL).

How do you do it, Mike? How come you're never stressed?

It's easy in research to get tunnel vision. You feel isolated, like you're the only person with your problem. But the truth is, you're not alone. And whenever I get stressed, I just ask myself the question: Do I really care? Would I rather be doing anything else?

- The PHD Movie

